

Three Studies in Healthcare Costing and Quality. New evidence from England and Wales on the Impact of the Reference Cost Index and Payment by Result Systems

By Abdullah Awadh D Alotaibi

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

Norwich Business School University of East Anglia September 2021

© Abdullah Alotaibi 2021

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with the author and that use of any information derived therefrom must be in accordance with current UK Copyright Law. In addition, any quotation or extract must include a full attribution.

Declaration

I confirm that this is my own work and the use of all material from sources has been properly and fully acknowledged.

. 2 Signed:

Date: September 2021

My parents: father (Awadh) and mother (Modhi); my siblings: Joseph, Mariam, Mohammed, Yasser and Maha;

my close friend, soulmate, and wife (Haya);

my princesses (Haneen and Leen).

Acknowledgements

Firstly, I wish to express my deepest gratitude to Allah for giving me this opportunity and for helping me to succeed in my academic studies. I recognise that this is thanks to His assistance and that without it I could not have achieved as much as I have.

I also recognise the importance of my University in Saudi Arabia, Imam Abdualrahman Bin Faisal University. They have invested so much in me by allowing me to be in the UK for eight years and for funding me throughout this period, while I completed the required languages courses and my Masters, as well as my PhD. It has been a long journey and they have supported me throughout, for which I express my heartfelt thanks.

I am also deeply grateful for the help and encouragement of my supervisors in the Department of Accounting in the Business School at UEA. Dr Pinar Guven-Uslu and Dr Ricardo Santana have provided invaluable support and have always encouraged me to produce work of the highest possible standard, for which I thank them most sincerely. I am grateful for their forbearance during those times I struggled to produce much work and for being so accommodating about the difficulties I faced as an overseas student during the Covid-19 pandemic.

None of this would have been possible without the love, support and encouragement of my parents. They have always believed in me even when I had no confidence and it is through their determination as much as my own that I have come so far. I will never be able to repay them but I wish to express my deep gratitude and love for them. I would also like to thank all my siblings who have always been there for me, with their support and encouragement.

I am also deeply grateful to my wife, Haya, for all she has done for me. It is said in both Arabic and English that behind every successful man there is a woman, and for me, Haya is that woman. I really appreciate the many, many sacrifices she has made so that I could complete my PhD. She has shouldered most of the burden of looking after the house and caring for our daughter, Haneen, so that I could spend my time focusing on my work. Icannot thank her enough for all she has done.

I also really appreciate my colleagues in the department at UEA, Aiman Farran, Saif Almutairi, Abdulaziz Alomran, Ahmed Albalushi and Pavlo Ulianiuk. It has meant so much to me to have the support of other people who were also studying at this level and I wouldlike to thank them for all they have done to help me.

Access Condition and Agreement

Each deposit in UEA Digital Repository is protected by copyright and other intellectual property rights, and duplication or sale of all or part of any of the Data Collections is not permitted, except that material may be duplicated by you for your research use or for educational purposes in electronic or print form. You must obtain permission from the copyright holder, usually the author, for any other use. Exceptions only apply where a deposit may be explicitly provided under a stated licence, such as a Creative Commons licence or Open Government licence.

Electronic or print copies may not be offered, whether for sale or otherwise to anyone, unless explicitly stated under a Creative Commons or Open Government license. Unauthorised reproduction, editing or reformatting for resale purposes is explicitly prohibited (except where approved by the copyright holder themselves) and UEA reserves the right to take immediate 'take down' action on behalf of the copyright and/or rights holder if this Access condition of the UEA Digital Repository is breached. Any material in this database has been supplied on the understanding that it is copyright material and that no quotation from the material may be published without proper acknowledgement.

Table of Contents

Table 1: List	of Abbreviations	7
List of Figure	es	8
List of Tables	s	.8
Abstract	1	1
Chapter 1: In	troduction1	4
1.	Objective1	6
2.	The Choice of UK Healthcare	16
3.	Basic Concept: National Reference Cost Index and the Market Forces Factor	18
4.	The Structure of the Thesis	24
Chapter 2: Tr	rend Analysis of the Reference Cost Index in National Health Service Trusts in	
England and	Wales from 1997 to 2016	
1. Intr	roduction	
2. Prie	or Literature	
	2.1. Payment by Results in Healthcare Systems	
3. Res	search Questions	
4. Dat	ta and Methodology	
	4.1 Data	
	4.2 Methodology	
5. Res	sults	
5.	1 Descriptive Statistics	
5.	2 Descriptive Statistics for Trust Region	
5.	3. Descriptive Statistics for Types of Trust40	
5	4 Descriptive Statistics for Foundation Status	
5.	5 Comparison between the National Average Value, which is 100, and the Mean	
of	Reference Cost Index of Trusts in each Region, Type and Foundation Status .42	
5.	6 Comparison between the Means of Reference Cost Indices of Trusts in each	
Cl	haracteristic (i.e., Region, Type and Foundation Status)	
6. Co	nclusion	
Chapter 3: Th	ne Relationship of National Health Service Trust Characteristics to the Reference Cost	

Index	
1. Introduction	50
2. Theoretical Framework	52
3. Literature Review and Hypotheses	53
3.1 Efficiency Studies using Data Envelopment Analysis	58
3.2 Trusts Characteristics	59
4. Data and Methodology	66
4.1. Data	66
4.2. Dependent Variable	67
4.3. Independent Variables	67
4.4. Control Variables	68
4.5. Methodology	69
5. Results	71
6. Discussion	76
6.1. Theoretical Contribution	
7. Conclusion	79
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Serv	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Serv	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Serv 1. Introduction	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Serv 1. Introduction	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction 2. Background 3. Quality Definition and Measures 	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction 2. Background 3. Quality Definition and Measures 4. Literature Review 	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction 2. Background 3. Quality Definition and Measures 4. Literature Review 5. Data and Methodology 	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
 Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare
Chapter 4: Reference Cost Index Reduction and Care Quality of National Health Server 1. Introduction	vice Healthcare

Table 1: List of Abbreviations

No.	Acronym	Full Name
1	ABC	Activity-Based Costing
2	ANOVA	One-way Analysis of Variance
3	BPT	Best Practice Tariffs
4	CCG	Clinical Commissioning Groups
5	CMS	Centers for Medicare and Medicaid Services
6	CQC	Care Quality Commission
7	CQUIN	Commissioning for Quality and Innovation
8	DEA	Data Envelopment Analysis
9	DRG	Diagnostic Related Groups
10	FP	For-Profit
11	FT	Foundation Trusts
12	GDP	Gross Domestic Product
13	GMM	General Method of Moments
14	GP	General Practitioner
15	HRG	Healthcare Resource Groups
16	ISTC	Independent Sector Treatment Center
17	LOS	Length of Stay
18	MAX	Maximum
19	MFF	Market Forces Factor
20	MIN	Minimum
21	NFP	Not-For-Profit
22	NHS	National Health Service
23	NICE	National Institute for Health and Clinical Excellence
24	Non-FT	Non-Foundation Trusts
25	NRCI	National Reference Cost Index
26	OECD	Organisation for Economic Co-operation and Development
27	OLS	Ordinary Least Square
28	PbR	Payment by Results
29	РСТ	Primary Care Trusts
30	PFI	Private Finance Initiative
31	RCI	Reference Cost Index
32	SDD	Same-day Discharge
33	SSN	Servizio Sanitario Nazionale
34	STD	Standard Deviation
35	VBC	Volume-Based Costing

List of Figures

Chart 1A. The Maximum and the Minimum Reference Cost Index for all Trusts from 1997 to 2016	39
Chart 1B. Percentage of Trust with Reference Cost Index within and below the National Average	
from 1997 to 2016	39
Chart 1C. Number of Trusts from 1997 to 2016	39

List of Tables

Chapter 1: Introduction

Table 1: An Example of Reference Cost Index calcu	ation
---	-------

Chapter 2. Trend Analysis of the Reference Cost Index in National Health Service Trusts in England and Wales from 1997 to 2016

Table 1. Descriptive Statistics of RCIs of all Trusts from 1997 to 2016 (five-year intervals)	38
Table 2. Results of One Sample t Test of the Mean RCI of each Region on the National Average	. 43
Table 3. Results of One Sample t Test of the Mean RCI of each Type on the National Average	43
Table 4. Results of One Sample t Test of the Mean RCI of each Foundation Status on the National Average .	43
Table 5. Results of Comparing the Means of Reference Cost Indexes	. 45
Table 6. Results of Games Howell Post Hoc Test to Compare the Means of RCIs of each Region with	
every Other Individual Region from 1997 to 2016	. 46
Table 7. Results of Games Howell Post Hoc Test to Compare the Means of RCIs of each Type with every	
Other Individual Type from 2002 to 2016	46
Table 8. Results of Independent One Sample t Test to Compare between the Means of RCIs of FTs and	
Non-FTs from 2005 to 2016	46

Chapter 3. The Relationship of National Health Service Trust Characteristics to the Reference Cost Index

Table 1. Descriptive Statistics of RCIs, Size and the Control Variables from 2009 to 2016	73
Table 2. Results of OLS Regression Test of RCI Relations with Regions from 2009 to 2016	75
Table 3. Results of OLS Regression Test of RCI Relations with Trust Type from 2009 to 2016	77
Table 4. Results of OLS Regression Test of RCI Relations with Foundation Status and Size	
from 2009 to 2016	

Chapter 4. Reference Cost Index Reduction and Care Quality of National Health Service Healthcare

from 2010 to 2016
Table 1. Panel B. Number of National Health Service Acute Trusts Included in this Study from 2010 to
2016 by Foundation Status and Cost_dummy105
Table 2. Descriptive Statistics of the Continuous Variables used in this Study from 2010 to 2016 105
Table 3. Results of Correlations between the Variables used in the Regressions from 2010 to 2016 109
Table 4. Results of Panel Fixed Effect Regression of Cost Relations with Healthcare Quality from
2010 to 2016
Table 5. Results of Panel Fixed Effect Regression of Cost_dummy Relations with Healthcare Quality
from 2010 to 2016

Appendix 1

Table A. The Development of Reference Cost Index from	n 1998 to 2016 1	63
---	------------------	----

Appendix 2

Table A. Descriptive Statistics of RCIs of all Trusts from 1997 to 2016	165
Table B. Descriptive Statistics of RCIs of all Trusts in West Midlands from 1997 to 2016	166
Table C. Descriptive Statistics of RCIs of all Trusts in Wales from 1997 to 2016	167
Table D. Descriptive Statistics of RCIs of all Trusts in South West from 1997 to 2016	168
Table E. Descriptive Statistics of RCIs of all Trusts in South East from 1997 to 2016	169
Table F. Descriptive Statistics of RCIs of all Trusts in North West from 1997 to 2016	. 170
Table G. Descriptive Statistics of RCIs of all Trusts in North East from 1997 to 2016	. 171
Table H. Descriptive Statistics of RCIs of all Trusts in Greater London from 1997 to 2016	. 172
Table I. Descriptive Statistics of RCIs of all Trusts in East of England from 1997 to 2016	173
Table J. Descriptive Statistics of RCIs of all Trusts in East Midlands from 1997 to 2016	. 174
Table K. Descriptive Statistics of RCIs of all Trusts in Yorkshire and the Humber from 1997 to 2016	175
Table L. Descriptive Statistics of RCIs of all Acute Trusts from 1997 to 2016	176
Table M. Descriptive Statistics of RCIs of all Ambulance Trusts from 2002 to 2016	177
Table N. Descriptive Statistics of RCIs of all Community Trusts from 1997 to 2016	178
Table O. Descriptive Statistics of RCIs of all Mental Health Trusts from 1997 to 2016	179
Table P. Descriptive Statistics of RCIs of all Specialist Health Trusts from 1997 to 2016	180
Table Q. Descriptive Statistics of RCIs of all Non-Foundation Trusts from 2005 to 2016	181
Table R. Descriptive Statistics of RCIs of all Foundation Trusts from 2005 to 2016	181
Table S: Results of One Sample t Test of the Mean RCI of each Region on the National Average	182
Table T: Results of One Sample t Test of the Mean RCI of each Type on the National Average	. 182
Table U. Descriptive Statistics for ANOVA Test	183
Table V: ANOVA test to compare between the means of RCIs of regions from 2002 to 2016	. 184
Table W: Welch and Brown-Forsythe tests for regions from 2002 to 2016	184
Table X: ANOVA test to compare between the means of RCIs of regions from 2005 to 2016	. 184
Table Y: Welch and Brown-Forsythe tests for regions from 2005 to 2016	184
Table Z: ANOVA test to compare between the means of RCIs of types from 1997 to 2016	. 185

Table AA: Welch and Brown-Forsythe tests for types from 1997 to 2016	. 185
Table AB: ANOVA test to compare between the means of RCIs of types from 2005 to 2016	. 185
Table AC: Welch and Brown-Forsythe tests for types from 2005 to 2016	. 185
Table AD: ANOVA test to compare between the means of RCIs of regions Excluding Wales	
from 1997 to 2016	186
Table AE: Welch and Brown-Forsythe tests for regions Excluding Wales from 1997 to 2016	. 186

Appendix 3

Table A. Results of OLS Regression Test of RCI Relations with Regions from 2012 to 2016	187
Table B. Results of OLS Regression Test of RCI Relations with Trust Type from 2012 to 2016	189
Table C. Results of OLS Regression Test of RCI Relations with Foundation Status and Size	
from 2012 to 2016	190

Appendix 4

Table A. Results of Dynamic Panel Data Estimation	. 191
Table B. Results of Panel Fixed Effect Regression of Cost_dummy Relations with Healthcare Quality	
from 2010 to 2016	. 193

Abstract

Healthcare provision accounts for a high proportion of public finances worldwide. Therefore, efficiency, quality and value for money are paramount. This thesis focuses on healthcare resource usage in the National Health Service (NHS), in England and Wales where reforms have long sought to ensure value for money and accountability. However, evaluating healthcare performance is complex. Since 1997, the National Reference Cost Index (NRCI) costing system has been applied to generate reliable data for all NHS clinical treatments and to drive improvements to the system. This thesis encompasses three studies.

Study 1 Contributes to knowledge in the field of healthcare accounting by presenting an updated mapping of RCIs and an examination of trends within the NHS trusts. The study investigated trends and changes to RCIs, trust¹ numbers, trust types and variations between the mean RCI of each group (i.e., region, type and foundation status) from the national average. Descriptive statistics were used, from 1997 (the introduction of the Reference Cost system) to 2016: for 1997 to 2009 data were collected from the National Archives of the Department of Health website; data for 2010 to 2016 were collected from the NHS research attention, extending our understanding of healthcare costs, trends, patterns figures and differences between foundation and non-foundation trusts over the various regions and trust types, for the first 20 years of NRCI application.

¹ NHS Trust is a legal entity that provides goods and services for the purposes of the health service (NHS, 2019b).

Study 2 examines cost variations between 2009 and 2016 by analysing the relationship between specific organisational characteristics (i.e., location, the type of service, foundation status and size), which were taken as individual uncontrollable variables, and RCIs. The National Archives of the Department of Health website provided the data for 2009, while NHS Reference Costs Collection website were used for data from 2010 to 2016. Size data was from England NHS website. Patient classification and gender, collected from NHS Digital website, were taken as control variables. Ordinary Least Square (OLS) regression technique was used and cost differences across trusts were identified. The study is distinct from previous work in this field as it focuses on trust characteristics and, to the author's knowledge, is the first to analyse the associations between combined trust characteristics and RCIs over an extended time period. Furthermore, the study uses data for all HRGs, rather than selecting specific categories, as previous research has done. Study 2 takes contingency theory as its theoretical framework, since it suggests that applying a single cost system for organisations which differ as to their geography, type, size, ownership and technology may not be appropriate and that variations should be taken into account.

Study 3 examines the relationship between RCI reduction and the quality of trust services provided, in the context of the introduction of Payment by Results (PbR) in the NHS to improve the cost efficiency of trust management. The study applied seven quality measures: mortality rates, Same-day Discharge (SDD), infection rates, mean Length of Stay (LOS), emergency and in-patient and out-patient waiting times. Data were collected from three NHS websites for the period 2010 to 2016. Data for RCIs and foundation status were from the NHS Reference Costs Collection website. Data for the dependent and other control variables were extracted from the NHS Digital website and the National Archives of the Department of health. Study 3 uses panel fixed effect regression. This study contributes to the fields of costing and

healthcare quality by examining the association between cost and the quality of trust services provided. Study 3 is the first, to the researcher's knowledge, to use infection rates as a measure of quality, or to evaluate quality while making a distinction between trusts at the upper and lower ends of the RCI scale.

Chapter 1: Introduction

Introduction

Healthcare provision worldwide requires considerable financial input, accounting for a large percentage of public finances globally. Healthcare providers have to manage increasing service costs and growing requirements for up-to-date medical technology, which are expected to raise the quality of the service provided (Gok and Altindağ, 2015). Additionally, the challenge of treating aging populations (Atella et al., 2019) increases overall expenditure (The World Bank, 2015; Labro, 2015; Gebreiter and Ferry, 2016; Malmmose, 2018). As a result, the performance of healthcare systems has been a matter of international concern (e.g., Smith, 2002; Ferrari, 2006; The World Bank, 2015; Labro, 2015; Labro, 2015; Labro, 2015) and health services in many countries have undergone reorganisation with a view to combining efficiency, service quality and value for money. Examples of countries in which healthcare reform has been a priority include those in Eastern Europe and the West (Mossialos et al., 2002; Kounetas and Papathanassopoulos, 2013; Cantor et al., 2018; Atella et al., 2019).

England and Wales's reforms of the National Health Service (NHS) have long striven to give value for money to taxpayers (Bojke et al., 2017; Gaughan et al., 2018) and to tackle the waste of resources from inefficient health services which burdens the system and hinders the ability of the service to offer quality care to patients who require it (Appleby and Thomas, 2000). Various attempts have been made over previous decades to introduce both management and financial reforms (e.g., Bourn and Ezzamel, 1986a; Broadbent et al., 1991; Broadbent, 1992; Preston et al., 1992; Harrison et al., 1994; Hood, 1995; Jones and Dewing, 1997; Llewellyn, 1998; Jones, 1999). However, clinicians have often rejected cost-saving initiatives, which as a result, have been poorly implemented and have failed to provide appropriate support for managerial decision-making (Pollitt et al., 1988; Preston et al., 1992; Jacobs, 1995;

Abernethy, 1996; Doolin, 1999). Thus, maximising efficiency has been high on the government agenda to ensure that budgets are used effectively and efficiently (Bourn and Ezzamel, 1986a; Broadbent et al., 1991; Broadbent, 1992; Preston et al., 1992; Harrison et al., 1994; Hood, 1995; Jones and Dewing, 1997; Llewellyn, 1998; Jones, 1999; Lapsley, 2001). Reform of public services in general and of healthcare, in particular, has been an important issue in the field of accounting, where it has given rise to several innovations (Kaplan and Norton, 1992; Hood, 1995; Jackson and Lapsley, 2003; Lapsley and Wright, 2004), as a means of enhancing efficiency and accountability including in the public health sector (Chow et al., 1998; Aidemark, 2001; Zelman et al., 2003; Modell, 2004), and encompassing the many complex factors involved in evaluating healthcare performance.

1. Objective

The objective of this thesis is to describe, through a trend analysis, the patterns, differences and variations in NHS trusts costs over a twenty-year period. The thesis provides a comparison of Reference Cost Index (RCIs) between 1997 and 2016 in terms of region, type, and foundation status. The thesis also examines variations in cost between 2009 and 2016 by testing the association between organisational characteristics (i.e., location, the type of service provided, foundation status and size) and RCIs, using contingency theory as a lens through which to examine the issues involved. Furthermore, the thesis aims to determine whether RCI reduction is negatively associated with the quality of trust services provided, by examining seven quality measures (i.e., mortality rates, Same-day Discharge (SDD), infection rates, Length of Stay (LOS), emergency and in-patient and out-patient waiting times).

2. The choice of UK healthcare

This thesis focuses on the efficient use of healthcare resources in England and Wales because of the size of the budgets and the risks that any inefficiency in the service poses to patients who require treatment within the service. Efficiency can be defined as maximising output without increasing input, or minimising input without compromising output. The Reference Cost system follows this approach as trusts that provide services included in the RCI at a lower cost than the average NHS provider have an index below 100, NHS trusts that reduce their comparative costs for the same output will see a fall in their RCI.

Furthermore, UK healthcare trusts are held accountable for their use of resources since these are supplied through general taxation. England and Wales are particularly interesting as a sample when analysing resource use because, since 1997, the costing system known as the NRCI has provided a mechanism to produce reliable data for all NHS clinical treatments and to facilitate cost comparisons by means of "reference costs" which highlight cost variations and are used to inform managerial decision-making (Department of Health, 1997).

According to the Department of Health (2014, p.17), "The RCI provides a comparison of costs at the aggregate level for each trust." It is therefore meaningful to examine changes across this composite system because the whole system has been designed to enhance comparability, through the use of Healthcare Resource Groups (HRGs) as the basis of RCI calculation. This is one of the reasons the Reference Cost system has been developed. There have been some changes over the twenty-year period which this study acknowledges and addresses by considering the statistics for different time periods within the overall twenty-year.

The Reference Cost system is the largest ever cost information resource to be used in healthcare. The data collection system covers all hospitals that use HRGs, unlike those used elsewhere in the world which only collect representative samples (Schreyögg et al., 2006). The UK provides a useful focus for this study because its public benchmarking system is recognised worldwide as leading the field in healthcare governance (McKee, 2002). When cost variations across trusts are known and published, the resulting pressure on institutions who are not performing to the required standards can lead to improvements (Northcott and Llewellyn, 2004; Tillema, 2010), which may benefit patients and the general public (Dawson and Street, 1998).

3. Basic Concepts: National Reference Cost Index and the Market Forces Factor

The calculation of RCI is made by dividing actual costs by expected costs, as explained by the Department of Health (2014). To calculate expected costs, the national average unit cost for the organisation is multiplied by the activity of that organisation. Comparisons are made between similar organisations (i.e., trusts are compared against trusts and Primary Care Trusts (PCTs) against PCTs). All data, except those related to services which are commissioned or contracted out, is taken into account when calculating the RCI. Activity and unit costs from different services and organisations are collected. Weighting of unit costs is according to the activity. Calculation of a national weighted average is then made. Merely dividing total expenditure by patient numbers does not produce meaningful unit costs. Casemix adjustments are therefore made when possible, whereby each case of patient care receives a classification according to its complexity, or mix, which is expressed as HRGs. HRGs are groupings of treatments which are equivalent in terms of resource use, as determined by clinicians, and are representative of UK clinical practice. The version of HRG currently in use is HRG4+, which was introduced in 2012.

The system of using RCIs has developed over the years following its introduction. In 1999, the use of RCIs was extended to cover other types of treatment (Llewellyn and Northcott, 2005). Accident and Emergency services, certain community-based and outpatient services were included in the system from 2000. In 2002, the Ambulance Service came under the auspices of the scheme. In the same year, financial incentives were put in place and a Payment by Results (PbR) system was instigated, with funding for treatment based on average costs (Llewellyn and Northcott, 2005).

The year 2005 heralded the first appearance of Foundation Trusts (FT). FTs are accountable to the public, staff, patients and partner organisations, and are allowed to retain surpluses, borrow money and raise investment from both public and private sectors, but are still performance rated annually. Non-Foundation Trusts (Non-FTs) cannot choose what services they provide or boost their finances by carrying out private work, so trusts aspire to be awarded foundation status (Department of Health, 2005). For more details on the history of the development of the Reference Cost system (see Appendix 1, Table A).

While the costs to trusts of such necessities as consumables and equipment are the same nationwide, other expenses are dependent on location. The Market Forces Factor (MFF) is a method used to adjust all costs reported by trusts to take into account differences in the factor costs they incur. It is an index covering geographical variations in land prices and the costs of buildings, and labour (Department of Health, 2013), since these so-called "unavoidable costs" or "uncontrollable costs" vary in different areas of the country (Department of Health, 2013).

The purpose of MFF is to estimate differences in uncontrollable costs. Since healthcare providers in areas such as London and the South East face higher staffing costs and land and

building expenditure as a result of external market forces, the MFF is used to compensate for this as the PbR tariff provides organisations with extra funding using the MFF, which offers a level basis for ensuring equivalent amounts of healthcare per pound are provided in all areas of the country. The MFF is also used to adjust the RCI (Department of Health, 2013).

As explained by The Department of Health (2013), the fixed national price for each healthcare activity is paid by commissioners according to the national tariff, based on the national average costs which every NHS trust reports. Initially, MFF was used for resource allocation for commissioners of healthcare providers. Those receiving additional funding to compensate for the high costs in their area could afford to provide the same level of services as those in less costly areas, although trusts can select locations within a limited area, they must remain within their region and cannot control any extra costs incurred. MFF is therefore applied as a proportion of the tariff price paid over and above the tariff for each unit of activity. If a trust's uncontrollable costs are 20% higher than providers in other regions, they should be given 20% of extra funding, thus reimbursing them for uncontrollable cost differences.

An overall MFF value for an organisation is created by multiplying the index value for each MFF element by its proportion of total running costs. In this method a weight is applied to each element of the MFF, which is equivalent to its weight within total costs. A total figure for the organisation is reached by adding together the weighted index values. (A single overall MFF value has an impact on the income of an organisation which is equivalent to the effect of using the index value for each element separately. Thus, the proportion of expenditure represented by a single element is reflected in total income).

The elements of provider costs used to account for MFF variations are:

(1.) non-medical staff,

(2.) medical and dental staff,

(3.) land, and

(4.) buildings.

Cost variations for each element are separately calculated initially, then combined to provide an overall index for each organisation. The MFF value for each NHS organisation thus reflects its uncontrollable costs and an index is used to determine the relativities between organisations. All organisations receive a ranking within the index. To make the adjustment, each organisation's index is divided by its MFF to ensure that comparisons between organisations throughout the country are "fair" (the term fair used in the NHS document to indicate that unavoidable costs incurred by some trusts as a result of external market forces are taken into account when calculating their RCIs and setting tariffs).

Two versions of the MFF index are used: (1) The underlying index is used to calculate RCI. (2) The payment index is used in PbR. The basis for both the underlying and the payment indices is the same data set, which is converted into different indices for different purposes. An index comprises a set of values which all relate to a single base value. An index can be used in different ways if the base value is changed. The first version of the index to be calculated is the underlying index. It is this underlying index which is the basis for allocating Clinical Commissioning Groups (CCGs), and the average value forms the base to which all other index values relate. The average value of the underlying index is 1.0. Organisations incurring higher uncontrollable costs than average organisations receive a value above 1.0, while those with lower than average costs are allocated a value below 1. The RCI is calculated using the underlying index. One example of how the MFF affects the underlying index value of a trust in a low-cost region is the case of the Royal Cornwall Hospitals NHS Trust. In 2007, after MFF adjustment, their RCI increased from 92 to 102 (MFF value of 0.9). Conversely, in an example of a trust from a high-cost region, for the same year, the RCI for University College London Hospitals NHS Trust, with an MFF value of 1.27, the index decreased from 129 to 101. For a further example of how RCIs are calculated, taking into account the MFF, please see Table 1.

		А	В	С	D= C/A	Е	F = B * D	G = B * E	H = F/G*100		
Trust	HRG	MFF	Activity	Unit cost (£)	Unit cost adjusted for MFF (£)	National average unit cost adjusted for MFF (£)	Actual cost adjusted for MFF (£)	Expected cost adjusted for MFF (£)	RCI adjusted for MFF		
Trust A	HRG1	1.1	10	12	10.9	11.2	109.1	112			
Trust A	HRG2	1.1	20	22	20	23.6	400	472			
Total							509.1	584	87		
Trust B	HRG1	0.9	15	10	11.1	11.2	166.7	168			
Trust B	HRG2	0.9	15	25	27.8	23.6	416.77	354			
Total							583	522	112		
Note: This table shows an example of how PCIs are calculated taking into account the Market Forces Factor, PCI- reference cost											

 Table 1: An Example of Reference Cost Index calculation²

Note: This table shows an example of how RCIs are calculated taking into account the Market Forces Factor. RCI= reference cost index; HRG= healthcare resource group; MFF= market forces factor.

To calculate the national tariff, the payment index is used. In this version of the index, the base value is the minimum and all other values relate to this (as opposed to the average which is used in the underlying index). The minimum value for the payment index is 1.0. The organisation incurring the lowest uncontrollable cost receives 1.0 as its value. Those incurring higher costs than the minimum are allocated a value above 1.0. Thus, if an organisation incurs uncontrollable costs which are assessed as being 5% above those of the lowest cost provider, they are allocated an index value of 1.05. When the MFF is applied to reimburse providers, each organisation's index value is multiplied by the tariff price set for each unit of activity. The tariff price includes the lowest possible level of uncontrollable costs faced by an NHS organisation. The provider incurring the lowest cost receives no extra funding for their activity,

² This table was extracted from the NHS website:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/380322/01_Final_2013-14_Reference_Costs_publication_v2.pdf

while the funding received by all other providers includes an extra proportion of the tariff price to compensate them for their uncontrollable costs. Each provider's PbR income/total income is calculated according to the following formula:

Total income = (Activity * Tariff price) * MFF value

For example: 1000 units of activity for HRG A are undertaken by trust Y. The tariff price of HRG A is £300, the trust has an MFF value of 1.25

Total income = $1000 * \pounds 300 * 1.25 = \pounds 375,000$

Thus, trust Y receives £75,000 more than the lowest cost provider to compensate for higher uncontrollable costs.

Cornwall Partnership NHS Foundation Trust is the provider incurring the lowest level of uncontrollable costs, so they receive the lowest payment index value. The setting of the lowest value of the payment index as 1.0 ensures that no MFF payments to providers are below zero. This base minimum of 1.0 allows for a single national price to be used, in accordance with PbR funding policy, by giving the same price to all providers then allocating separate compensation for their uncontrollable costs. If the MFF underlying version of the index were to be applied in calculating PbR, it would be the middle range of providers' uncontrollable costs which were included in the tariff price, so those with costs below this median, would receive less than the tariff price once the MFF had been applied. Thus, the payment index differentiates between controllable costs and uncontrollable costs. All providers receive the tariff according to the costs incurred by all providers. The MFF then allows those incurring uncontrollable costs above the minimum to be compensated appropriately.

To produce the payment index, the underlying index is converted by dividing each of its values by the minimum value, thus generating an index with 1.0 as its base value. Thus, the

values allotted to an individual provider in the underlying and payment indices will differ, although the between-provider relatives remain constant. Thus, if one trust has 20% higher costs than another, its MFF will be 20% higher than the other trust, regardless of whether the underlying index or the payment index is used.

4. The Structure of the Thesis

This PhD thesis contains three studies. The first study provides descriptive statistics to determine trends, patterns and differences in cost variations over the period from 1997 (the introduction of the Reference Cost system) to 2016. Data for years between 2010 and 2016 were extracted from the NHS Reference Costs Collection website, while data for the years between 1997 and 2009 were gathered from the National Archives of the Department of Health website. The study examines trends and how figures have changed in terms of RCIs, trust numbers, types of trusts and differences between the mean of RCIs of each group (i.e., region, type and foundation status) from the national average. Study 1 contributes to the knowledge in the field of healthcare accounting by presenting an updated mapping of RCIs and examining trends within the NHS. The focus of the study is on trust characteristics, in contrast to many previous studies which have examined patient characteristics or productivity (e.g., Oh et al., 2016; Lovecchio et al., 2016; Huang et al., 2017; Bettin et al., 2018). The study extends our understanding of the trends and patterns of the figures across regions and types, for the first twenty-year following the introduction of the Reference Cost sytem. It also examines the differences between FTs and Non-FTs.

The second study aims to examine variations in costs between 2009 and 2016 by testing the relationship of certain trust characteristics (i.e., location, the type of service, foundation status and size) on their RCIs. RCI data for 2009 were collected from the National Archives of the Department of Health website. For the years between 2010 and 2016 NHS Reference Costs Collection website data were used. Data concerning size were extracted from England NHS website. The control variables were patient classification (i.e., emergency, in- and outpatients) and gender, and this information was taken from NHS Digital website. Many studies in the field of accounting and economics have considered the impact of trust characteristics, such as ownership, on decision making (e.g., Krishnan and Yetman, 2011; Holzhacker et al., 2015). These studies assume that trust characteristics are important and impact decisionmaking and efficiency. Study 2 considers the relationship of the individual uncontrollable variables of location, trust type, foundation status and trust size on RCIs between 2009 to 2016, and uses Ordinary Least Square (OLS) regression technique. This study is distinct from earlier work in the field which has focused on patient characteristics (e.g., Hollingsworth, 2008; Castelli et al., 2015b; Cantor and Poh, 2018), because it examines trust characteristics, and I know of no previous research which has analysed the associations between the combined trust characteristics and RCI over an extended period. The impact of location has previously been considered, but in these cases the focus has been on specific HRGs, rather than all treatments provided by the NHS (e.g., Daidone and Street, 2013). The theoretical framework for Study 2 is contingency theory, which suggests that organisational structures are not all the same and that it may be difficult and inappropriate to apply a "one size fits all" approach.

In 2002, the NHS in England introduced PbR to improve the cost efficiency of trust. Study 3, therefore, examines the relationship between RCI reduction and the quality of services provided by trusts. Seven quality measures were used in the study: mortality rates, SDD, infection rates, LOS, emergency and in-patient and out-patient waiting times. Data were taken from two NHS websites for the period between 2010 and 2016. Data for RCIs and foundation status were extracted from the NHS Reference Costs Collection website. The NHS Digital

website provided the data for the dependent and other control variables. The design of Study 3 used panel fixed effect regression. The contribution of this study to the field of accounting, costing and healthcare quality is that it examines the relationship between cost and the quality of services provided. To the best of the researcher's knowledge infection rates have not previously been used as a measure of quality. This study is also the first to undertake a quality evaluation while distinguishing between trusts at the upper end of the RCI scale and those at the lower end.

The overall structure of the rest of this PhD thesis is as follows. Chapter 2 presents the descriptive statistical Study 1; Chapter 3 presents the second study which examines the associations between organisational characteristics and cost, while Chapter 4 presents the third study which examines the relationship between RCI and the quality of services provided. In the conclusion, the results are summarised and the limitations of the research are acknowledged.

Chapter 2: Trend Analysis of the Reference Cost Index in National Health Service Trusts in England and Wales from 1997 to 2016

Trend Analysis of the Reference Cost Index in National Health Service Trusts in England and Wales from 1997 to 2016

Context: The NHS in England and Wales introduced the Reference Cost system in 1997, in order to provide reliable and comparable cost data for maximising efficiency and reducing variation in treatment costs.

Objective: To describe, through a trend analysis, patterns and differences in variations in costs over a twenty-year period. The study also provides a comparison of RCIs between 1997 and 2016 in terms of region, type and foundation status.

Data Sources: Data for years between 2010 and 2016 were extracted from NHS Reference Costs Collection website, while data for years between 1997 and 2009 were gathered from the National Archives of the Department of Health website.

Study Design: Descriptive statistical tests.

Main Findings: Cost variations between trusts between 1997 and 2016 were reduced. The variation in 1997 was higher than it was in 2016 by 5%. In comparison with the national average, RCIs for six of the regions and three trust types are significantly higher. Non-FTs have significantly higher RCIs than the national average. There are also significant differences in RCI among regions, types and foundation status.

Conclusion: Overall, it is possible to observe that there is a slow movement towards the average, with the service providers with the lower RCIs moving costs upwards towards the average and service providers with the higher RCIs moving costs downwards. Further workis needed to examine in more detail the remaining variations between trust characteristics and their RCIs.

Key Words: NHS, RCI, Cost efficiency, Reference Cost system

1. Introduction

The performance of healthcare systems is a worldwide concern, mainly due to the higher costs associated with the provision of this service (e.g., Smith, 2002; Ferrari, 2006; The World Bank, 2015; Labro, 2015). A high proportion of public finances is devoted to healthcare throughout the developed countries, with costs set to rise due to the expense of purchasing up-to-date medical technology and the challenges of treating aging populations (Atella et al., 2019). Reorganisation of health services to improve efficiency, effectiveness, quality and value for money has therefore been high on the political agendas in Eastern European countries and in the West (Mossialos et al., 2002; Kounetas and Papathanassopoulos, 2013), including England and Wales, where governments have sought to ensure that the NHS gives value for taxpayers' money (Bojke et al., 2017; Gaughan et al., 2018). Inefficient health services waste resources and reduce the ability of the system to provide adequate care (Appleby and Thomas, 2000).

To address the problem of waste, there were various earlier attempts to introduce management and financial reforms in the NHS in England and Wales during the 1980s, as discussed by numerous authors (e.g., Bourn and Ezzamel, 1986a; Broadbent et al., 1991; Broadbent, 1992; Preston et al., 1992; Harrison et al., 1994; Hood, 1995; Jones and Dewing, 1997; Llewellyn, 1998; Jones, 1999). However, such initiatives have often lacked acceptance from clinicians and have not been implemented in such a way as to genuinely support the decision-making of the management of the NHS providers (Pollitt et al., 1988; Preston et al., 1992; Jacobs, 1995; Abernethy, 1996; Doolin, 1999).

The Reference Cost system was therefore introduced in 1997 as a mechanism to reduce variations in the cost of healthcare provision. It aimed to enable trusts' managers to "tackle

unacceptable variations in performance and raise overall standards across the NHS by sharing information and comparing performance" (Department of Health, 1997, p.2). This has made the UK a leader in using public benchmarking in healthcare governance (McKee, 2002). This costing system, covering all treatments and all NHS trusts, is the largest ever cost information resource, providing reliable data for all NHS clinical treatments and allowing cost comparisons by means of "reference costs" to highlight and reduce variations in costs and to facilitate managerial decision-making (Department of Health, 1997). The cost-collection system in England is unique because it is the only one that collects all data from all trusts, rather than just using representative samples (Schreyögg et al., 2006). However, it has been suggested that the information has not been used effectively to guide setting treatment tariffs³ (Street and Maynard, 2007). Nevertheless, improved efficiency results from putting pressure on poorly performing institutions (Northcott and Llewellyn, 2004; Tillema, 2010), thus providing quality healthcare at a reasonable cost, which was a priority for patients and the general public as a whole (Dawson and Street, 1998).

This study contributes to the literature by presenting an updated mapping of RCIs and its trends within the NHS. The research focuses on trust characteristics rather than patient characteristics or productivity, as has been the case with many previous studies across the world (e.g., Oh et al., 2016; Lovecchio et al., 2016; Huang et al., 2017; Bettin et al., 2018). This will enhance our understanding of the trends and patterns, across regions, types and the differences between FTs and Non-FTs, covering data for the first twenty-year following the introduction of Reference Cost system. The breadth of this descriptive study and length of time it covers makes the data more robust. It is meaningful to look at changes across such a

³ The amount paid for a specific treatment.

range of providers because the Reference Cost system is standardised with the purpose of facilitating comparability between trusts. Changes during the years do not affect the results because trusts are compared with each other for any given year. However, the efficiency comparisons can only relate to services included in the RCIs, many of an individual provider's services in non acute trusts, particularly in the early years of the study may be excluded.

The following sections of this chapter are organised as follows: section two reviews the relevant literature; section three presents the research questions; section four presents the data and methodology; section five details the results and section six concludes with suggestions for future work.

2. Prior Literature

The Department of Health (2014, p.4) has defined the NRCI as "the average unit cost to the NHS of providing defined services to NHS patients in England in a given financial year". RCI data have been collected on an annual basis since 1997. However, in order to produce such an index, it is first necessary for the organisations within the service to identify their unit costs and understand them. When organisations are large and decentralised, as is the NHS, consistency of cost calculation is vital. However, since the work of the NHS differs so vastly from manufacturing, where total expenditure can be divided by numbers of goods produced, other ways of calculating unit costs must be developed. Total expenditure cannot simply be divided by the number of patients treated because the cost per patient will vary according to the needs of that patient and the treatment they receive. Costs will be influenced by the type of patient treated and the nature of the treatment given. HRGs provide a national standard framework to adjust for variations in casemix by producing groups of treatments which are clinically comparable and require similar resources. It is clinicians not accountants who group

the treatments, because the intention behind HRGs is that they should be clinically coherent and allow for a "common language" comprehensible to clinicians and managers alike.

This is important because if inefficient trusts are to recognise that they have a problem, they need to be convinced either that the patients they treat are the same as those treated elsewhere, or that suitable adjustments have been made in calculating their performance in comparison with other health institutions. Calculation of HRGs is undertaken in retrospect and is based on the actual costs a trust has incurred, and uses the national average cost for the previous 3 years for all patients falling into that particular HRG (Castelli et al., 2015a). An individual costed HRG is made up of a weighted average calculated from the cost of these groups of treatment. The overall cost of an HRG is then divided by its total activity.

The Reference Cost system allows comparisons to be made between these average unit costs for each NHS trust and to produce a single figure for each NHS trust, which makes it possible to compare the actual cost of its treatments with the same treatments calculated in terms of average costs nationally (Department of Health, 1997). The RCI is a simple number reached by the following calculation. The actual total costs incurred by a trust, are taken to be the costs for its HRG x the number of such resource groups. The expected costs of the trust are the national average costs of HRGs x the number of such resource groups. The RCI is reached by dividing actual costs by expected costs. Adjustment to the RCI is made to take into consideration discrepancies in cost resulting from differences in location, and those MFF which trusts cannot control. This is a much more detailed weighting system applying across locations throughout England than the previous London weighting, and the adjustment alters RCIs considerably. Benchmarking clubs could be used as a way of encouraging hospitals to lower their costs. Therefore the NHS ranks trusts in accordance with their RCI values and the results published on the Department of Health website. An average performance in terms of cost is given an index score of 100. A score exceeding 100 suggests an above-average cost of performance, while a score below 100 indicates a below-average cost. An RCI score of 95 suggests a trust is performing more efficiently than the average, while a score of 105 indicates costs 5% above the average and hence a lower level of efficiency.

Patient-level information and cost systems (PLICS) have also been used by the NHS, since the mid-2000s, to calculate the costs of individual patient care episodes. These IT systems combine activity, financial and operative data and aim to reduce waste and enhance efficiency. PLICS utilize a 'bottom up' rather than a 'top down approach' and are based on all patientassociated cost drivers (Ellwood et al., 2016).

2.1.Payment by Results in Healthcare Systems

PbR systems of various kinds have been used worldwide to encourage cost efficiency (Grašič et al., 2015; Longo et al., 2017). Any system of PbR to healthcare providers requires differentiation of patients according to the type of treatment required (Bojke et al., 2017). The healthcare sector is diverse and complex. This makes it difficult to ensure that payment for treatment is fair and meets the needs of patients (Grašič et al., 2015). Payment systems should thus take into account the various aspects of trusts' complex needs and motivate providers to provide an adequate level of service and ensure good patient outcomes. Efficiency needs to be increased and transparency maximised (Busse, 2012). The oldest method of tackling issues of fairness when calculating payments is the Diagnostic Related Groups (DRG), used by the US Medicare system since 1983 for the billing of patients and insurers (Jackson, 2001). HRGs, which are the British version of DRGs, classify patients depending on diagnosis, treatments and certain patient characteristics (Atella et al., 2019). Tariffs then reflect the national average

costs for the HRG across all trusts (Bojke et al., 2017; Longo et al., 2017). Thus, HRGs need to account for all the possible variables if the RCIs are to be correct and the PbR system is to work fairly. However, Geissler et al. (2011) acknowledge that no payment system can work perfectly in every respect.

Northcott and Llewellyn (2002) stress the importance of addressing such issues as variations in how standard costs for procedures are produced, differences in clinical practice, differences in trust running costs and variations in the LOS if comparative costs data is to be used effectively. Healthcare payment systems similar to DRGs in the US and HRGs in the UK are being increasingly adopted around the world (e.g., Netherlands, Australia, Italy, Germany and Spain) (Jackson, 2001). However, there is still limited knowledge about which design features work best (Busse, 2012). Hence, the current study aims to contribute to the knowledge in the field of healthcare accounting by determining the association between RCIs and variations in costs over a twenty-year period, and by testing trends related to location, type of trust (whether it is specialist, general, mental health, ambulance or community) and foundation status.

Some empirical evidence suggests that the Reference Cost system is associated with greater standardisation of trusts (Llewellyn and Northcott, 2005). Between 1997 and 2000 only 60% of trusts managed to keep their RCIs within 10% of the national average, but by the fifth year after the introduction of the Reference Cost system, this figure had risen to 72%. According to Guven-Uslu (2005), there is management support for performance measurement. However, some clinicians distrust the system because it is difficult to ensure the comparability of cost data (Bourn and Ezzamel, 1986b; Jones, 1999; Llewellyn and Northcott, 2005; Scarparo, 2006; Guven-Uslu and Conrad, 2008; Kurunmäki and Miller, 2008; Chapman et al., 2014). However,

this may not have been exclusively due to the influence of the Reference Cost system. Ten other possible influences on their RCIs and cost reporting for this period have been identified and grouped into four categories by Llewellyn and Northcott (2005):

- 1. Variations in approaches to costing. This category includes differences in the way costs are allocated, as well as variations in the way "care profiles" are generated and costed.
- 2. There were also variations in the basic clinical activities undertaken to meet the legitimate needs of patients, but not subject to the HRG3 adjustment.
- 3. Information quality. Clinical coding varied, as did the way in which activities were counted. Trust information systems also varied as to their capacity for data collection.
- 4. Efficiency differences in cost performance.

3. Research Questions

This study aims to examine the current trends in RCIs in England and Wales. Given the exploratory nature of the research, and based on the literature presented above, our further analysis is guided by three questions:

- 1. What are the trends for trusts in each region, different trust types and foundation status over the first twenty-year since the introduction of the Reference Cost system?
- 2. Which regions, types and foundation status differ significantly from the national average for RCIs?
- 3. How do the mean RCIs of each region, type and foundation status compare with one another?

4. Data and Methodology

4.1. Data

A large amount of secondary quantitative data was used for this study. Data for 1997-2009
was gathered from the National Archives of the Department of Health website ⁴. Data for 2010-2016 was obtained from the NHS Reference Costs Collection website ⁵. These data contained the name of the trust and the MFF adjusted RCI for each trust for each year and provided a table for every year of the Reference Cost system. Organisation codes were only introduced in 1999, so matching names and codes for 1997 and 1998 was problematic. This led to the exclusion of 70 observations for those years. The data tables were then collated to provide an overview of the entire period of the scheme. This produced a total of 4,946 observations, once those which could not be matched had been excluded.

4.2. Methodology

Descriptive statistics were calculated, using the 99 percentile for the maximum of RCI, and 1 percentile for the minimum, to exclude any outliers in order to examine variations and movements towards the average. The number of trusts for each year was noted and this made it possible to see which trusts had RCIs below the national average (i.e., 101%)⁶. Calculations were made for all trusts and separately for each region, type and foundation status.

In order to provide a more comprehensive picture, the above results were tabulated and the table was then extended to include trust characteristics. The first of these characteristics is region. Following government information, ten regions in England and Wales were identified: West Midlands, Wales, South West, South East, North West, North East, Greater London, East of England, East Midlands and Yorkshire and The Humber⁷. The region in which each

⁴ This data is available at:

https://webarchive.nationalarchives.gov.uk/+/http://www.dh.gov.uk/en/Managingyourorganisation/Financeandp lanning/NHScostingmanual/index.htm

⁵ This data is available at: <u>https://www.gov.uk/government/collections/nhs-reference-costs</u>

⁶ One percent was allowed for small deviation from the average which is 100%.

⁷ Details of these ten regions are available at: <u>https://www.ons.gov.uk/ons/guide-method/geography/beginner-s-guide/maps/regions--former-government-office-regions--gors---effective-at-31st-december--2011.pdf</u>

trust is operating was identified by means of the postcode of that trust.

The second characteristic is the type of organisation. There are five different organisational types within the NHS: acute, specialist, ambulance, community and mental health⁸. The type of each trust was identified using NHS websites and the websites of the individual institutions. Ambulance was not included in the data until 2002. The third characteristic is whether or not a trust is a FT. A trust can only be a FT if it has demonstrated best practice and sustainable financial viability (Department of Health, 2005). From the names provided for each identification code across the twenty-year, it was possible to highlight when a trust received foundation status.

Two statistical tests have been conducted. The first test, the one-sample t-test, compares the mean of RCI for each characteristic classification (i.e., regions, types and FTs and Non-FTs) with the national average and secondly, One-way Analysis of Variance (ANOVA) is used to compare the means of RCI between each classification. To compare the variation between the mean of RCIs of various regions and types, Games Howell Post Hoc was used to overcome the limitation of unequal sample size.

Three periods were examined, the entire twenty-year period (from 1997 to 2016), since the introduction of the Reference Cost system. From 2002 to 2016, following the introduction of PbR, and the inclusion of ambulance trusts, and from 2005 to 2016 covering the period following the introduction of foundation status.

⁸ Some trust types did not have HRGs from 1997. However, they appeared in the Department of Health publication of RCIs for 1997 because they offer acute services. For example, there were no mental health HRGs in 1997. However, the document for that year includes 24 mental health trusts.

5. Results:

5.1.Descriptive Statistics

Table 1 and Chart 1 (A, B, and C) provide a general overview of RCIs over the two decades prior to 2016. The results set out in Table 1 and Chart 1 (A, B and C) show that the number of trusts increased between 1997-2016 by 38 new trusts. However, when PbR were introduced in 2002, trust numbers were 273. Thus, 39 trusts merged, possibly to reduce costs. The highest RCI scores for all trusts decreased toward the national average from 148 in 1997 to 128 in 2016. The results for the minimum of RCI between 1997-2016 increased by 10 points over the period. The variation between trusts decreased over the period as the standard deviation was 14% in 1997 and 9% in 2016.

There was a 4% increase in trusts falling within the acceptable national average range (i.e., between 99 and 101), between 1997-2016 and, surprisingly, a 4% decrease in the number of trusts falling below 101. Thus, in 2016 more trusts were exceeding the national average than in 1997 (see Appendix 2, Table A for further details).

Table 1. Descriptive Statistics of RCI	s of all Trusts	1997 irom 1997	to 2016 (IIV	e-year inter	vals)
All Trusts	1997	2002	2007	2012	2016
Number of Trusts	196	273	238	244	234
MAX	148.28	151.38	142.56	136.08	128.19
Median	99.00	99.72	99.11	99.86	100.41
MIN	73.76	71.93	81.90	78.17	83.63
STD	13.71	14.62	10.58	19.03	9.09
Average	100.29	101.21	100.41	101.58	100.54
Number of Trusts within 99-101	10	27	16	26	20
% of Trusts within 99-101	5.08	9.89	6.72	10.66	8.55
Number of Trusts below 101	113	153	135	138	124
% of Trusts below 101	57.36	56.04	56.72	56.56	52.99

Table 1. Descriptive Statistics of RCIs of all Trusts from 1997 to 2016 (five-year intervals)

Note: This table reports the summary of the statistics for all NHS trusts between the period 1997 to 2016. For the sake of brevity, only the statistical results for every five years are reported. See Appendix 2, Table A for the full table. RCI= reference cost index; Number of trusts= number of trust that their RCI was published; MAX= the 99 percentile of RCIs in a given year; MIN= the 1 percentile of RCIs in a given year; STD= standard deviation; Number of trusts within 99-101= number of trust that their RCI was below 101= number of trust that their RCI was below 101%.



Chart 1A. The Maximum and the Minimum Reference Cost Index for all Trusts from 1997 to 2016

Chart 1B. Percentage of Trusts with Reference Cost Index within and below the National Average from 1997 to 2016







5.2.Descriptive Statistics for Trust Region

Previous literature, such as Llewellyn and Northcott (2005), has highlighted managerial concerns about trusts in Greater London and the South East being disadvantaged due to their location. Therefore, this study examines the significance of regional variations. There were no clear trends that could be applied to all regions. From 1997 to 2016 the number of trusts in all regions increased, except for Yorkshire and The Humber, which decreased by two trusts. However, the number of trusts from 2002, following the PbR system, decreased in seven regions. The maximum RCI increased in East Midlands and East of England between 1997-2016 and remained approximately the same in Greater London, Yorkshire and The Humber, the South West. The minimum was below the average and increased towards the national average in Greater London, Yorkshire and The Humber, the South West, East of England and the East Midlands, while the minimum RCI in the North West and Wales were below the average but decreased further over the period. The variation did not increase in any region. It either decreased or remained constant.

There are only three regions that had a higher percentage of trusts with an RCI below 101: the North East, and also South East and Greater London. The latter are two of the regions whose managers complained that they were disadvantaged (Northcott and Llewellyn, 2003). The differences in patterns between the regions and the lack of any definite trends across the entire country, suggest that region is an important characteristic when discussing cost. Therefore, it would make sense for region to be taken into account in other calculations at least as a control variable (see Appendix 2, from Table B to Table K).

5.3.Descriptive Statistics for Types of Trust

When examining trends for the types of trust, data was taken from 2002 to 2016 because

2002 was the year when the ambulance trusts started reporting their RCIs. All types of trust decreased in number, except for community trusts. Overall, the variation in RCIs between trusts in each type decreased, which suggests that the Reference Cost system is having the desired effect on costs and standardising the work of the trusts. Community and mental health trusts demonstrated similar trends. In both cases, all trusts had a slight increase in their maximum RCI over the period and the minimum remained constant. A percentage was calculated by taking the number of trusts in a given type with an RCI below the national average, in a given year, divided by the total number of trusts of the same type, in the same year. This percentage for both community and mental health trusts was higher in 2016 than it had been in 2002. However, between 2002 and 2009 only two community trusts were reporting their data. The percentage for specialist trusts was also higher but the maximum and minimum trends were not the same as those for community and mental health trusts but were the same as those for acute and ambulance trusts, where their maximum RCIs decreased and their minimum increased (see Appendix 2, from Table L to Table P).

5.4.Descriptive Statistics for Foundation Status

A further important characteristic is whether a trust is FT or Non-FT. It would be expected that a FT would perform better than a non-FT because of the standards required to gain foundation status. There were no data from before 2005 because this was the year in which FTs began. In 2005 only 12% of trusts achieved foundation status but this number increased gradually until, in 2016, 65% of trusts had become FTs, leaving only 35% as Non-FT. FTs and Non-FTs usually went in different directions. FTs showed a slight increase in the maximum while the RCIs for Non-FTs decreased. FTs already had minimum RCIs below the national average but decreased still further. Non-FTs showed an increase towards the average. Non-FTs showed a decrease in variation in RCI, while FTs remained the same. Both FTs and Non-FTs and Non-FTs was and the same.

FTs showed a slight decrease in the percentage of trusts falling below the national average. It is noteworthy that during the years 2011 and 2012 Non-FTs had better average RCIs than FTs. The year 2012 was an exception, with an unusually high maximum of 353%, but the data for 2011 are difficult to explain (See Appendix 2, Table Q and Table R).

5.5. Comparison between the National Average Value, which is 100, and the Mean of Reference Cost Index of Trusts in each Region, Type and Foundation Status

This section compares and assesses the differences in mean RCI and the national average. Trusts were grouped by regions, types and foundation status using one-sample t-test.

Taking the whole period from 1997 to 2016 (Table 2), the average RCI of six regions (i.e., West Midlands, Wales, North West, North East, Greater London and East Midlands) was significantly higher than the national average⁹. East of England, Yorkshireand The Humber had averages that were significantly lower than the national average. For the other two periods 2002-2016 and 2005-2016, please see Appendix 2, Table S. In terms of type (data from 2002 to 2016), acute had RCIs below the national average, while specialist, community and mental health trusts were significantly higher than the national average (Table 3). Two further tables covering the time periods from 1997 to 2016 and 2005 to 2016 can be found in Appendix 2, Table T. For the FTs and Non-FTs between 2005 and 2016, Non-FTs had RCIs which were significantly above the national average (Table 4).

⁹ The calculation of the national average is explained in Chapter 1 and in the literature section.

Pagions			1997-2016	5
Regions	Ν	Mean	t	Mean Difference
West Midlands	558	101.85	3.633	1.847**
Wales	48	105.4	3.148	5.396**
South West	555	99.59	-0.807	-0.410
South East	566	100	-0.001	-0.000
North West	898	101.08	2.538	1.083*
North East	451	101.43	2.815	1.433**
Greater London	904	101.63	3.164	1.625**
East of England	441	98.272	-3.310	-1.728**
East Midlands	439	102.64	3.559	2.640**
Yorkshire and The Humber	86	97.718	-2.096	-2.282*
Note: RCI= reference cost index. * P-value is	$\leq 0.05, **I$	P-value is ≤ 0	0.01	

 Table 2. Results of One Sample t Test of the Mean RCI of each Region on the National Average

Table 3. Results of One Sample t Test of the Mean RCI of each Type on the National Average

Tunas		20	002-2016				
Types	Ν	Mean	t	Mean Difference			
Acute	2194	98.929	-7.299	-1.071**			
Ambulance	237	101.651	1.695	1.651			
Community	137	102.480	3.003	2.480**			
Mental health	857	101.187	2.268	1.187*			
Specialist	290	111.719	13.210	11.719**			
Note: RCI= reference cost index. * P-value is ≤ 0.05 , ** P-value is ≤ 0.01							

Table 4. Results of One-Sample t-Test of the Mean RCI of each Foundation Status on the National Average

Equindation Status			2005-2016				
Foundation Status	Ν	Mean	t	Mean Difference			
Non-foundation	1443	101.248	4.796	1.248**			
Foundation Trusts	1464	100.352	1.139	0.352			
Note: RCI= reference cost index. * P-value is ≤ 0.05 , ** P-value is ≤ 0.01							

5.6. Comparison between the Means of Reference Cost Indices of Trusts in

each Characteristic (i.e., Region, Type and Foundation Status)

In the previous section, the analysis showed that the RCI of trusts from some regions, types and Non-FTs were significantly different from the national average (100%). In this section, the differences in RCI between the regions, types and FTs and non-FTs are tested using ANOVA test.

Significant differences were found for all characteristics tested. Tables 5 (Panel A and Panel B) show the significance (p-value $\leq 5\%$) of the mean differences in RCI within regions and types respectively. The Welch and Brown-Forsythe tests were conducted for these results, to confirm the robustness of the ANOVA tests (see Table 5 Panel C and Panel D). Further ANOVA tests were conducted to compare the means of RCI for both regions and types covering the time periods: 2002 to 2016 and 2005 to 2016, indicating the existence of a significant difference in each case (see Appendix 2, From Table V to Table AC). Furthermore, given that Wales has a low number of NHS trusts, the ANOVA tests for three time periods were re-run, excluding Wales, and the results remained significant and unchanged. However, the only time period which was tabulated was for 1997-2016 (see Appendix 2, Table AD and Table AE)¹⁰. The one-way ANOVA test only reveals the existence of a significant difference. It does not show between which region or type this occurs.

In order to compare the mean of RCIs of trusts in each region with every other individual region, the Games Howell Post Hoc test was conducted (see Table 6). East of England had a lower mean of RCI than six other regions (i.e., West Midlands, Wales, the North West, the North East, Greater London and the East Midlands) which makes it the region with the most other regions significantly above it. The South East is the only region that has no significant difference from any other region.

¹⁰ In Wales most health services are provided by health boards. Wales has no FTs and only three health trusts (an ambulance trust, a specialist cancer trust and a public health trust). Many NHS trusts and FTs in England provide acute health services.

These results suggest that the MFF, as explained in the introduction, might be working for Greater London and the South East, contrary to the claims in Northcott and Llewellyn (2003). Table 7 shows the results of the Games Howell Post Hoc Test for trust type. Acute has a lower mean of RCI than any other type, while the mean RCI for specialist is higher than that of all the others. In terms of foundation status, the independent sample test was used. The mean of RCIs of FTs was found to be significantly lower than that of Non-FTs for the years 2005-2016 (Table 8). There might also be other factors which have influenced RCIs and cost reporting for this period, as discussed by Llewellyn and Northcott (2005).

 Table 5. Results of Comparing the Means of Reference Cost Indexes

 Panel A. Results of ANOVA Test between Regions from 1997 to 2016

Sum of Squares	df	Mean Square	e F	Sig
8610.376	9	956.708	5.754	0.000
820696.449	4,936	166.268		
829306.826	4,945			
VA Test between Trus	st Types fre	om 2002 to 20)16	
Sum of Squares	df	Mean Squar	e F	Sig.
42934.134	4	10733.534	4 91.228	0.000
436502.877	3710	117.656		
479437.011	3714			
h and Brown-Forsyth	e Tests for	Regions fron	n 1997 to 2016	
	Statistic	df 1	df 2	Sig.
	6.710	9	765.433	0.000
	6.299	9	2229.070	0.000
ch and Brown-Forsyth	e Tests for	Trust Types	from 2002 to 2	016
Statistic	c df	1	df 2	Sig.
58.046	4	52	1.030	0.000
	Sum of Squares 8610.376 820696.449 829306.826 VA Test between Trus Sum of Squares 42934.134 436502.877 479437.011 h and Brown-Forsyth Statisti 58.046	Sum of Squares df 8610.376 9 820696.449 4,936 829306.826 4,945 VA Test between Trust Types from Sum of Squares df 42934.134 4 436502.877 3710 479437.011 3714 h and Brown-Forsythe Tests for Statistic 6.710 6.299 eth and Brown-Forsythe Tests for Statistic 58.046 4	Sum of Squares df Mean Squares 8610.376 9 956.708 820696.449 4,936 166.268 829306.826 4,945 VA Test between Trust Types from 2002 to 20 Sum of Squares df Mean Square 42934.134 4 10733.534 436502.877 3710 117.656 479437.011 3714 h and Brown-Forsythe Tests for Regions from Statistic df 1 6.299 9 Statistic df 1 Statistic df 1	Sum of Squares df Mean Square F 8610.376 9 956.708 5.754 820696.449 4,936 166.268 829306.826 4,945 VA Test between Trust Types from 2002 to 2016 Sum of Squares df Mean Square F 42934.134 4 10733.534 91.228 436502.877 3710 117.656 479437.011 3714 h and Brown-Forsythe Tests for Regions from 1997 to 2016 Statistic df 1 df 2 6.299 9 2229.070 ch and Brown-Forsythe Tests for Trust Types from 2002 to 2 6.299 9 2229.070 Ch and Brown-Forsythe Tests for Trust Types from 2002 to 2 2 Statistic df 1 df 2 58.046 4 521.030 5

Regions	Wales	South West	South East	North West	North East	Greater London	East of England	East Midlands	Yorkshire and The Humber
West Midlands	-3.549	2.257	1.847	0.764	0.413	0.330	3.604**	-0.793	3.976*
Wales		5.806	5.396	4.313	3.963	3.879	7.153**	2.756	7.525*
South West			-0.410	-1.493	-1.843	-1.927	1.347	-3.050*	1.719
South East				-1.083	-1.434	-1.517	1.757	-2.640	2.129
North West					-0.350	-0.434	2.840**	-1.557	3.212
North East						-0.083	3.191**	-1.207	3.563
Greater London							3.274**	-1.123	3.646
East of England								-4.397**	0.372
East Midlands									4.769*
Note: RCI= refere	Note: RCI= reference cost index. * P-value is ≤ 0.05 . ** P-value is ≤ 0.01								

Table 6. Results of Games Howell Post Hoc Test to Compare the Means of RCIs of each Region with every Other Individual Region from 1997 to 2016

Table 7. Results of Games Howell Post Hoc Test to Compare the Means of RCIs of each Type with every Other Individual Type from 2002 to 2016

	Ambulance	Community	Mental health	Specialist		
Acute	-2.723*	-3.551**	-2.259**	-12.790**		
Ambulance		-0.829	0.464	-10.067**		
Community			1.293	-9.239**		
Mental health				-10.531**		
Note: RCI= reference cost index. * P-value is ≤ 0.05 , ** P-value is ≤ 0.01						

Table 8. Results of Independent One-Sample t-Test to Compare between the Means of RCIs of FTs and non-FTs from 2005 to 2016

	Levene's Test		t-test for I	t-test for Equality of Means		
	F	t	df	Mean Difference		
Equal variances assumed	3.546	-2.217	2905	-0.897*		
Equal variances not assumed		-2.220	2830.528	-0.897		
Note: RCI= reference cost index; FTs= foundation trusts; Non-FTs= non-foundation trusts. * P-value is ≤ 0.05 , ** P-value is ≤ 0.01						

6. Conclusion

Previous research has recognised the importance of working towards greater efficiency in healthcare. In England and Wales considerable attention has been given to the NHS budget. The NHS is funded by general taxation and trusts must be accountable for efficient use of resources. The Reference Cost system and RCIs have been used in the NHS in England and Wales since 1997 to provide cost data to prompt efficiency improvements and reduce treatment cost variations. This research aimed to describe trends and patterns in cost variations over a twenty-year period. It also sought to examine the variations in RCIs between regions, types and foundation status, to provide descriptive statistics and to identify trends. Data were drawn from NHS Reference Cost Collections and the National Archives of the Department of Health website and analysed using descriptive tests, ANOVA and one-sample t-test.

The results demonstrate that overall variations between trusts decreased over the period, suggesting that, since the introduction of the Reference Cost system and RCIs, costs between trusts have been standardized. While there were significant differences in costs between regions, there were no clear trends across all regions. The variation in cost did not increase in any region: it either decreased or remained constant. When examining RCIs by hospital type (acute, specialist, community, mental health or ambulance): acute hospitals had lower RCIs while specialist hospitals had the highest; again the RCIs showed reducing variation, but there were differing trends in RCIs by hospital type when looking at maximum and minimum percentiles. When examining foundation trust status or not, the analysis of RCIs confirmed FTs as more efficient. However, there were differences between the characteristics tested (i.e., region, type and foundation status). One limitation of this study is that it does not examine the association between organisational characteristics and costs. Such analysis requires the collection of additional data and further statistical analysis. However, this will be addressed in the next chapter which will consider the association between organisational characteristics and RCIs. This is important under the current public healthcare payment system, where costs must be controlled while standards are maintained. Understanding these characteristics will provide the data to underpin potential action and the policy decisions required to improve efficiency and standardisation across the entire healthcare sector.

Chapter 3: The Relationship of NHS Trust

Characteristics to the Reference Cost Index

The Relationship of National Health Service Trust Characteristics to the Reference Cost Index

Context: The Reference Cost Index has been used in the NHS in England since 1997, to generate comparable cost data to improve efficiency and reduce variation in treatment costs.

Objective: This study aims to examine variations in costs between 2009 and 2016 by testing the relationship of certain organisational characteristics (i.e., location, the type of service provided, foundation status and size) on RCIs. Contingency theory is used as a basis for this study, as a lens through which to examine the issues involved.

Data Sources: Data for 2009 were gathered from the National Archives of the Department of Health website, while data for the years between 2010 and 2016 were extracted from NHS Reference Costs Collection website. Size data were gathered from England NHS website and the control variables (i.e., patient classification and gender) were obtained from NHS Digital website. **Study Design:** OLS regression.

Main Findings: This study found that the East of England, South East and South West were the most efficient regions, while East Midlands and the North West and Yorkshire and The Humber were the least efficient. In terms of type, mental health and acute were most efficient and specialist and community least efficient. FTs were found to be more efficient than Non-FTs. Smaller trusts were found to be more efficient than larger trusts.

Conclusion: Contingency theory suggests that there is no one ideal costing system to fit all types of institutions. Thus, the factors examined in this study are likely to be associated with the RCIs of trusts which differ as to the nature of their work and other pressures from their specific circumstances. The results indicate that MFF adjustment for regional differences and the last version of the HRG, HRG4+, has gone some way towards improving the Reference Cost system, but greater refinement both in the cost system and adjustments to RCI may be needed to take into account other variations in trusts such as the factors examined in this study.

Key Words: NHS, RCI, Cost efficiency, Cost System, Benchmarking.

1. Introduction

A large proportion of public finance worldwide is devoted to healthcare provision, mainly due to high costs and increasing expenditure, associated with the need to purchase up-to-date medical technology, and the challenges of treating aging populations (The World Bank, 2015; Labro, 2015; Gebreiter and Ferry, 2016; Malmmose, 2018). Consequently, obtaining value from healthcare budgets is a concern of governments. In the United Kingdom, governments have long striven to maximise the efficiency of the NHS, to address issues of waste, to improve efficiency and obtain better value from its budget (e.g., Bourn and Ezzamel, 1986a; Broadbent et al., 1991; Broadbent, 1992; Preston et al., 1992; Harrison et al., 1994; Hood, 1995; Jones and Dewing, 1997; Llewellyn, 1998; Jones, 1999; Lapsley, 2001).

To apply efficiency in a health care context in England and Wales, the Reference Cost system was introduced into the NHS in 1997. It was designed to reduce cost variations between NHS trusts, to tackle variations in performance and to improve standards through information sharing and performance comparison (Department of Health, 1997). The Reference Cost system has developed over the twenty-year it has been in place (please see Table A Appendix 1). Chapter 2 in this PhD thesis provided descriptive statistics of RCIs and concluded that there had been a move towards the national average over the period. Nevertheless, there are still considerable variations in RCIs across trusts. Many accounting and economic studies have examined the effect of trust characteristics, such as ownership, on decision making (e.g., Krishnan and Yetman, 2011; Holzhacker et al., 2015). An underlying assumption across these studies was that trust characteristics matter and variations on decision making and consequently on efficiency could be influenced by those characteristics.

Previous studies have recognised that there are numerous potential variables that may affect health service cost (Llewellyn and Northcott, 2005). However, despite some distrust of the cost systems among clinicians (Bourn and Ezzamel, 1986b; Jones, 1999; Llewellyn and Northcott, 2005; Scarparo, 2006; Guven-Uslu and Conrad, 2008; Kurunmäki and Miller, 2008; Chapman et al., 2014), there is empirical evidence suggesting that Reference Costs have improved cost standardisation (Llewellyn and Northcott, 2005). The introduction and literature review presented in Chapters 1 and 2 of this thesis examined the previous work on policy and showed how Reference Costs were introduced and have developed throughout the years summarized in Table A. The literature review for the current study will consider more granular cost systems applied in the health service through Patient Level Information and Cost Systems (PLICS) using activity-based costing (ABC) techniques; and studies covering efficiency, patient classification and patient and trust characteristics.

This study examines and discusses the system which has been applied for more than twenty-year to the healthcare payment system in England and Wales, using a combination of data not previously applied to the study of health care efficiency. The study measures the relationship between the location, trust type, foundation status and trust size and RCI from 2009 to 2016. The focus on trust characteristics, rather than patient characteristics or productivity, distinguishes this study from previous studies (e.g., Hollingsworth, 2008; Castelli et al., 2015b; Cantor and Poh, 2018). This study analyses the association between individual uncontrollable variables (i.e., location, trust type, foundation status and size) and RCIs employing linear regression. Patient classification and some of the patient characteristics will be used in this study as control variables. To the best of my knowledge, no studies have examined the association between the combined trust characteristics and RCIs over an extended period. Where the influence of location has been examined, it has mostly focused on selected HRGs, rather than considering all treatments provided by the NHS (e.g., Daidone and Street, 2013). It reinforces the view presented by contingency theory, illustrating that organisational structures may differ and that a "one size fits all" approach is challenging to apply. Nevertheless, the study demonstrates that, since the introduction of the Reference Cost system, progress has been made in the assessment of cost efficiency of NHS trusts with varied structures and characteristics.

The following section provides the theoretical framework. Section three provides a review of previous research in this field and discusses recent studies on healthcare treatment costing and the various studies which have assessed trust efficiency. Section four presents the data and the methodology of the current study. Section five displays the results. Section six presents the discussion, and the final section provides conclusions and suggestions for future research.

2. Theoretical Framework

Contingency theory was initially developed in the early to mid-1960s but became a popular theory within management accounting research in the 1970s (Otley, 1980), in response to rapidly changing circumstances and environmental uncertainty. It has become significant within management accounting because academics have recognised that an accounting system's organisational context is important (Otley, 2016). Contingency theory dominates the design of control systems for explaining variations in the relationships between organisational variables, contingency variables and environmental variables (Dent, 1990).

This theory provides a contrasting approach to traditional management theories, such as scientific management theory, bureaucratic theory and administrative theory, which suggest that there is one ideal way to structure an organisation. In this respect, they overlook the significance of the relationship between contextual variables and organisational structure and suggest that a single way of organising affairs is applicable regardless of the situation (Emmanuel et al., 1990). This theoretical background provides a useful alternative to these universal approaches by acknowledging that organisational structure is influenced by other

variables. Since environmental factors (e.g., technology, size and structure) have a role to play, there can be no universally applicable accounting system for all organisations under every possible context (e.g., Otley, 1980; Haldma and Lääts, 2002; Gerdin and Greve, 2004). Thus, there must be an association between the context of an organisation and the control system used (Islam and Hu, 2012).

The application of Reference Costs, as a control system, across all NHS trusts throughout England and Wales could be regarded as problematic in the context of contingency theory, as the system is applied to different trusts with different organisational structures (FT or Non-FT) and different sizes, doing different types of work (from acute care to mental health, specialist and community services), in vastly differing regions. Efforts have been made to ensure that differences in treatments are taken into account, through the introduction of the most recent version of the HRG classification, HRG4+, and the MFF adjustments made for regional variations. Nevertheless, inconsistencies may remain.

The literature review below highlights various factors, both internal and external which may affect a trust's ability to keep its costs within the national average. Type, region, size and foundation status may all have a role to play in influencing how well a trust managesto keep its costs within the average. For this reason, contingency theory has been selected as providing the most useful theoretical framework to inform this study and help develop appropriate hypotheses to test through quantitative analysis.

3. Literature Review and Hypotheses

Payment for healthcare treatment in England and Wales is based on the figures produced by the RCIs (as discussed in chapters I and 2). PbR of various kinds have been used worldwide to encourage cost efficiency in health care (Lowe and Doolin, 1999; Roeder et al., 2002; Lungen and Lapsley, 2003; Grašič et al., 2015; Holzhacker et al., 2015; Longo et al., 2017). The working of these systems has been explained in chapters 1 and 2. Any system of PbR to healthcare providers requires differentiation of patients according to the type of treatment required (Fetter et al., 1980; Fetter and Freeman, 1986; Bojke et al., 2017). The healthcare sector is diverse and complex. This makes it difficult to ensure that payment for treatment is fair and meets the needs of patients (Grašič et al., 2015). Payment systems should thus take into account the various aspects of trusts' complex needs and motivate providers to provide an adequate level of service and ensure good patient outcomes. Efficiency needs to be increased and transparency maximised (Busse, 2012).

Many studies argue that such PbR systems affect healthcare services significantly and improve the efficiency of healthcare organisations more than the previous system (Shleifer, 1985; Sloan et al., 1988; Chua, 1995; Lowe and Doolin, 1999). However, healthcare managers face difficulties in making strategic decisions due to uncertain revenues associated with the PbR system (Eastaugh, 1999; Kuntz et al., 2008). Therefore, several accounting and economic studies examine the effect of PbR and the operating risk that it presents on trust costs, behaviour and their decisions.

Healthcare costs have important implications for the social and institutional environment of trusts, as it has an impact on the lives of human beings. Consequently, numerous researchers have studied healthcare costs and the changes in healthcare services (e.g., Kurunmaki et al., 2006; Jones and Mellett, 2007; Cardinaels and Soderstrom, 2013; Chapman et al., 2014); for example, examining the associations and the effect of aging populations on healthcare cost (e.g., Fries, 1989; Zweifel et al., 1999), demographic changes and technology (e.g., Hopwood, 1992; Lapsley, 2001) or physicians' behaviour, involvement and attitudes (e.g., Bourn and Ezzamel, 1986b).

The introduction of the PbR has led to a degree of manipulation of financial reporting by trust managers. For example, in the US, more overhead costs have been allocated to outpatient services because their payment system uses a fixed price for inpatients. Financial manipulation has also affected the public-private patient balance in the US (Eldenburg and Kallapur, 1997). Kessler (2007) and Bai (2016) found that Californian trusts can charge private patients more than public patients and, as a result, trusts will reduce the number of public patients and increase the number of private patients.

The introduction of PbR has also led to the development of new accounting techniques which more accurately associate clinical activities with their financial outcomes (Chen et al., 2015). One important example of such a technique is the Activity-Based Costing (ABC) method. ABC is a management accounting innovation (Wegmann and Stephen, 2009), based on activities, which was developed by Kaplan and Cooper in the late 1980s and promoted and implemented in the 1990s (Dragija and Lutilsky, 2017). The system maintained a high profile for several years, attracting the interest of academics and practitioners (Malmi, 1999; Liu and Pan, 2007). The ABC method is a suitable technique for evaluating the cost of each healthcare activity, but since healthcare institutions vary as to their specificity, ABC needs to be implemented in ways which are appropriate (Dragija and Lutilsky, 2017). The aim of using an ABC system is to allocate costs fairly and accurately and to evaluate product profitability, focusing on the best way of allocating indirect costs to cost objects, thus rejecting the traditional costing assumption that resources are directly consumed by products and services (Awashi, 1994). The ABC is a system which can help control costs because it identifies the activities and resources which are most costly (Cannavacciuolo et al., 2015).

In an ABC system, products are seen as consuming activities and it is the activities which consume resources (Horngren et al., 2003). Activities and cost objects are linked by cost and activity drivers (Aldogan et al., 2014), a cost driver being any activity or factor directly affecting resource consumption. An activity can be any action, event or work sequence involved in the production of a product or service which incurs an expense (Weygandt et al., 2009). Since direct costs can be traced directly to services, customers or products, the greatest focus in an ABC system is on indirect cost, in contrast to the attention paid to cost centres by traditional accounting systems (Lanen et al., 2013), such as Volume-Based Costing (VBC), where each cost object (such as a unit of a trust) is allocated indirect costs by means of a single cost driver. The results thus produced are often approximate or inaccurate. However, the ABC method has a higher level of accuracy (Cao et al., 2006).

The introduction of the ABC method coincided with the growth of information technology which fundamentally altered the structure of production costs, with a move to the prevalence of indirect costs and a decrease in such direct production costs as labor and materials, and the system seemed to meet the requirements of these changes by providing precise data on product or service costs (Dragija and Lutilsky, 2017). In contrast, a traditional cost accounting system would not produce information quickly and accurately enough to meet modern management needs (Cooper and Kaplan, 1991). ABC's use of more cost centers and various kinds of cost drivers, as well as the assignation of activity costs to cost objects based on the usage of cost drivers, means that it can assess more accurately the resources required for each cost object (Drury, 2013). Using various cost drivers highlights cause and effect connections between activities and cost objects (Dragija and Lutilsky, 2017). The accuracy of the ABC method assists planning and the setting of prices (Homburg, 2004), and the identification of activities which do not add value allows these to be decreased or eliminated (Walther and Skousen, 2009; Dragija and Lutilsky, 2017).

Another advantage of ABC is that it is applicable for costing of both services and products.

Furthermore, it is appropriate in both the public and private sectors, supporting performance management techniques, for example balanced scorecards and continuous improvement (Eldenburg and Wolcott, 2005). It helps promote cost-reduction initiatives (Brimson and Antos, 1994), allows accountants to be more aware of how a service functions and why costs are incurred (Kirton and Hazlehurst, 1991) and enhances accounting performance measures, from the perspective of processes (Kohlbacher, 2010). However, its benefits differ according to subject (Dragija and Lutilsky, 2017).

O'Reilly et al. (2012) suggested that the ABC methodology was applicable to healthcare systems with varied organisational structures, funding systems and public/private sector involvement. Its implementation in five healthcare systems in Europe has led to greater efficiency and improved sustainability of the healthcare systems (Dragija and Lutilsky, 2017). Nevertheless, despite its advantages the ABC's adoption rate has remained low, giving rise to the debate about the so-called "ABC paradox" (Gosselin, 1997). In 1997, Dowless noted that ABC was not being used widely in the healthcare sector, while Lawson (2005), noted a decline in ABC use with fewer healthcare organisations even contemplating adopting the method.

Although ABC is an attractive method of accounting with many advantages, including its scientific approach and its sophistication (Wegmann and Stephen, 2009), the healthcare sector has been reluctant to adopt it because it is complex and challenging to implement in practice. Traditional methods are less complicated in terms of time, methodology and resource consumption. It is only realistic to adopt it if its advantages outweigh the disadvantages (Dragija and Lutilsky, 2017). Furthermore, if direct costs, direct material and direct labour consume a higher proportion of costs than overheads, ABC is not a suitable accounting method to use, and it can be difficult to identify an activity or cost object's cost driver (Drury, 2013). Popesko (2013) pointed out that the complex structure of trusts in terms of outputs, activities

and customers create barriers to trust managers considering ABC adoption. Additional staff with experience in ABC implementation may need to be employed and existing information management systems may need to be modified (Dowless, 1997). A further obstacle to the use of ABC has been a lack of senior management commitment (Lawson, 2005) and concerns that it might be a temporary trend which may rapidly be replaced by another new system (Kaplan, 1990).

In the NHS, PLICS are used to help calculate and reduce costs. These are IT systems which are used to calculate the costs of individual patient care episodes, by combining activity, financial and operative data. They began to be introduced in the mid-2000s as a result of the Department of Health recommendation that the patient be used as a cost object when drivers of healthcare costs were analyzed (Ellwood et al., 2016), and are still in use at the time of writing. By 2013, PLICs were being used by 120, mainly acute hospital trusts, with a further 86 undergoing the planning or implementation process (DoH, 2013), with the aim of improving the NHS economy as a whole, rather than just that of individual trusts, although initially it was the latter which saw the greatest benefits from the system. PLICS encourage waste reduction by identifying variation and providing clinicians with the financial information on which to base decisions which could enhance efficiency. PLICS data can be applied to allow benchmarking of activity at the patient level either against local protocols or in the context of national treatment practices. PLICS can assist when services are redesigned, can highlight the need to move services to alternative care settings and can help reduce the numbers of superfluous tests, interventions and referrals. PLICS data is collected and stored on a national database (Ellwood, et al., 2015). The PLICSs costing methodology differed from such methods as HRG costing, in that they took a 'bottom up' rather than a 'top down approach' and endeavored to take all patient-associated cost drivers into account (Ellwood et al., 2016). According to Llewellyn et al. (2016) and Ellwood et al. (2016), PLICS were used by trusts to drive improvements to costs and to help meet Cost Improvement Programmes (CIPs). They note, however, that, at that time, a lack of PLICS data and an unwillingness of trusts to share data meant that the potential of the system had not been realized.

3.1. Efficiency Studies using Data Envelopment Analysis

Many studies have investigated the efficiency of trusts, often using Data Envelopment Analysis (DEA) and measuring numerous inputs and outputs (Hollingsworth, 2008; Castelli et al., 2015a; Cantor and Poh, 2018). This is a one-stage efficiency analysis, in which achieving a high score for efficiency is the main aim (Kaya Samut and Cafri, 2016) and productivity has been the focus of many DEA studies (Giancotti and Mauro, 2015). Dixon et al. (2018) argue that productivity, while not usually key to NHS reform, should be given greater importance. While clinicians associate it with working harder and patients with costcutting, it really means more efficient working and waste reduction. It should not involve reducing quality. Atella et al. (2019), for example, demonstrated how policy objectives in England and Italy led to differential growth rates, with the NHS in England prioritising the reduction of waiting times, increasing activity and improving quality, and the Servizio Sanitario Nazionale (SSN) in Italy prioritising cost restriction and rationalising provision.

However, while numerous DEA studies have been published over the years, they have had only a limited practical impact on trust management and policy makers. This is partly due to concerns that the technique may not be robust enough and that the insights it provides are limited (Castelli et al., 2015a). One limitation of the method is that efficiency is taken to be a single aggregate measure. This means that they can provide a measure of a trust's relative overall efficiency but cannot explain why some trusts perform better than others. Neither can DEA analysis investigate how a single factor (e.g., location) can have an impact on efficiency (Chen et al., 2005). The present study uses OLS regression technique to analyse the association between single factors and RCIs. It identifies several characteristics of trusts which are variables outside the control of trust management. This is important because if the evidence on which policies are based is to be sound, uncontrollable variables must be identified and taken into account. Many studies to date have only taken some uncontrollable causes into account or have not accounted for interactions between them (Crémieux and Ouellette, 2001). By examining the uncontrollable characteristics (e.g., region, type, size and foundation status) selected for in the current study, it should be possible to better understand efficiency problems and their association with high RCIs, which could reflect factors beyond their control. The current study focuses on trust characteristics and their association with trust costs.

3.2. Trust Characteristics

The region, the type of trust, the foundation status and size could affect trusts' RCIs (e.g., Dredge, 2003; Roht et al., 2013; Herrera et al., 2014; Rezaee and Karimdadi, 2015; Giancotti et al., 2017; Longo et al., 2017; Atella et al., 2019). Both location (whether rural or urban) and provision of specialised services also affect cost (Palmer, 2005) and all of these characteristics are tested in the current study, using OLS regression to produce results that make it possible to identify associations between specific characteristics and RCIs.

Statistically significant geographical variations in cost have been identified by various studies, although these have not separated them from other influences (Lave and Lave, 1970; Vitaliano, 1987; Zuckerman et al., 1994). Indeed, there have been very few studies that take geographical location into account in DEA when assessing trust efficiency. Yet each region has its special status and issues, such as variations in labour costs, which need to be taken into account (Rezaee and Karimdadi, 2015). Research has highlighted significant regional economic disparities in many countries (Beenstock and Felsenstein, 2007). Although there is

an assumption in the healthcare literature that regions are homogenous, wages will be higher in areas where the concentration of experienced workers is high and there are fewer low earners, such as women, young people and those from ethnic minorities (Beenstock and Felsenstein, 2008). Demographic shifts, for example, an aging population or urbanisation, also affect the demands on the healthcare system (Mitropoulos et al., 2013). Such external, demographic factors can be regarded as environmental variables affecting the management and organisational structure of healthcare services (Mitropoulos et al., 2016), although the effect of location may be moderated by the range of services being offered (Mitropoulos et al., 2013). In Italy, for example, there are regional differences in trusts' capacity to deliver the services required, with a distinct North-South divide. Different population sizes alsoaffect the quality of provision, with regions of under 2 million inhabitants scoring morehighly in surveys on patient satisfaction (Elba et al., 2017).

In England there are clearly documented regional economic discrepancies. In terms of housing, London and the South East are the most expensive England regions and have higher than average housing costs (Office for National Statistics, 2019). Rienzo (2017) has identified economic differences across the England, in terms of expenses, wages, housing costs and regional demographics, including human capital. Housing costs represent the largest part of overall household expenditures and have been shown to vary greatly across the England regions (Cinzia, 2010). According to the Office for National Statistics, between 2004 and 2018, average house prices in the North East were approximately a third of those in London and a half of those in the South East (Office for National Statistics, 2019). Variations also exist in levels of total expenditure, excluding housing. Regional prices have been shown to be important in regional analysis (Hayes, 2005). In 2016, London, the South East and the South West had higher relative regional consumer price levels than the national average (Office for National Statistics, 2018b). Furthermore, average weekly earnings show regional variations,

with Londoners having 25% higher gross weekly earnings than England average (Office for National Statistics, 2018a). Blanchflower and Oswald (2005) have highlighted the huge regional differences in wages in the private sector, while public spending per person also demonstrates regional differences (Brien, 2018).

The NHS recognises these regional economic differences and has therefore introduced the MFF, which is an adjustment to take account of non-controllable regional variations in costs. (For further details about how MFF works and how it is calculated, please see page 19). This means that regional variations should be taken into account. Similar approaches with payment adjustments have been followed in other countries, including the US (Street and Maynard, 2007). However, Daidone and Street (2013) noted that costs for treating patients were higher in London, despite regional adjustment, and that the MFF was not completely resolving the problem of regional variations. These results, combined with the ideas expressed through contingency theory, suggest that region remains a factor in healthcare costs despite the adjustments introduced so far.

H1: RCIs differ between regions

Specialist service is defined in England as services that "support people with a range of rare and complex conditions" (NHS Specialised services, 2019). It may be needed if a patient has a particularly severe condition, if there are serious underlying problems, or if complications arise. Such services are therefore provided by trusts with the appropriate infrastructure and medical teams (Daidone and Street, 2013). In the US, specialisation of For-Profit (FP), physician-owned trusts have been criticised due to the practice of "cream-

skimming" for financial purposes (Shactman, 2005; Schneider et al., 2008)¹¹. In England, using specialist trusts is supposed to give the advantages of a "focused factory"¹² (Skinner, 1974), such as increased efficiency and quality, but not necessarily a reduction in costs (Longo et al., 2017). Indeed, if the payment system does not reflect the increased cost of treating more expensive patients, specialist trusts will be at a disadvantage (Bojke et al., 2018). Daidone and Street (2013) recognised the risk of financial disadvantage for trusts with specialist care teams if patients needing specialist care are more costly to treat than others in the same HRG, as discussed in Chapter 2 of this thesis. Analysing costs for over 12 million patients in 163 trusts for the fiscal year 2008/9, they estimated the extra costs associated with specialised care. They discovered that, while nineteen types of specialised care are no more expensive than others with the same HRG allocation, patients with cancer, spinal, neurological, rheumatological, colorectal or orthopaedic issues are more expensive to treat, as are children and those with cystic fibrosis. Those requiring specialist care tended to have longer LOS, which increased costs (Longo et al., 2017). Some of these specialisms were later covered by the top-up system for a restricted number of trusts (Longo et al., 2017).

Bojke et al. (2017) demonstrated that certain care markers (e.g., LOS) make a significant difference to base costs. They note that failing to take these additional costs into account, penalises trusts providing complex care services and recommend either refining HRGs to take complex care into account or using a top-up payment. In a US example, using patient, trust and geographic confounders as independent variables, Russell et al. (2016) demonstrated that childhood cancers and chemotherapy increased the cost severity of treatment, although other factors, such as diagnosis, age and type of trusts were also important.

¹¹ Cream-skimming refers to the practice of selecting the most profitable cases to treat.

¹² A focused factory refers to trusts specialising in particular types of treatments at which they are then able to excel.

Longo et al. (2017) used RCI data to examine a sample of specialist orthopaedic trusts and trauma departments in general trusts. Given the refinements to the payment system to ensure fairness and the advantages of specialist trusts in terms of expertise and focusing on a limited number of services, specialist trusts were expected to be more profitable than general trusts. However, this was not the case. Their regression results demonstrated a 13% average lower profit margin for the specialist trusts. Furthermore, two out of three specialist trusts were making a loss.

Since different types of treatment and specialisms have been shown to be associated with RCIs, the second hypothesis for the study is as follows:

H2: RCIs differ between types of trust.

Ownership and its impact on trusts' decisions and budgets have been studied in accounting and economic literature (e.g., Krishnan and Yetman, 2011; Holzhacker et al., 2015). The main reason for managers to act differently is the trust objective. The objective of trusts run FP is to maximise the profit, while Not-For-Profit (NFP) and public trusts aim to maximise the number of patients they treat (Dranove, 1988; Hoerger, 1991). Therefore, the impact of ownership has been studied as a factor affecting efficiency (Chang et al., 2004) and the role of the private sector in healthcare provision is a matter for debate (Kruse et al., 2018). Those who favour private provision argue that the competitive market and the drive to maximize profit will also improve efficiency, transparency and accountability, consumer choice and responsiveness (Hsu, 2010). It is also argued that, under the right conditions, private trusts could perform better than those under public provision (Kruse et al., 2018).

Those who are proponents of public provision highlight the incompatibility of profit

maximisation and public health. Managers of NFP trusts, which rely heavily on donations, are more motivated to shift administrative expenditures to the programme services in order to be more efficient, because the more efficient the trust, the more donations they receive (Weisbrod and Dominguez, 1986; Harvey and McCrohan, 1988; Khumawala and Gordon, 1997; Krishnan and Yetman, 2011). Another reason for managers to shift administrative costs is that managers have a compensation based on programmatic efficiency (Baber et al., 2002). The way in which ownership of health care institutions impacts on the provision of health care is important internationally (Chalkley and Sussex, 2018) and ownership of trusts has been discussed by many commentators since the 1980s (e.g., Gruca and Nath, 2001; Xirasagar and Lin, 2006; Shen et al., 2007; Bernet et al., 2011). Much of the evidence has been collected from the US, especially that which predates 2000 (Chalkley and Sussex, 2018). Much research has tried to establish how the type of ownership of trusts affects efficiency (Czypionka et al., 2014). Despite this, empirical studies have not produced any conclusive results (Chalkley and Sussex, 2018). However, all the above literature emphasises that trust managers should change their strategy and accounting choices in order to adapt the payment system and to reduce the operating financial risk.

Studies suggesting that private is more efficient than public provision have been undertaken in several countries including the US (Cowing and Holtmann, 1983), Taiwan (Chang et al., 2004), Germany (Robra and Werblow, 2006), Portugal (Barros et al, 2013) and Zambia (Masiye, 2007). Researchers who have found public provision to be more efficient than private include Custer and Willke (1991), Menke (1997) and Helmig and Lapsley (2001). The lack of observed efficiency differences between public and private providers was reported by a number of researchers (e.g., Vita, 1990; Steinmann and Zweifel, 2003; Staat, 2006; Farsi and Filippini, 2008; Herrera et al., 2014). Further distinctions can be found between the private and public sectors. Privately owned trusts can be run for FP or NFP; public trusts can be state-owned or run by public bodies (Saltman, 2003). Studies suggesting that FP trusts are more efficient than either public or NFP trusts include: Wilson and Jadlow (1982), Staat and Hammerschmidt (2000) and Tiemann and Schreyögg (2012). However other studies suggest that managers will foster their own interests at the expense of operational efficiency (Chang et al., 2004), therefore suggesting that FP trusts are less efficient than NFP or public providers (Rosko and Chilingerian, 1999; Rosko, 2001; Daidone and D'Amico, 2009; Lee et al., 2009; Herwartz and Strumann, 2012). However, the evidence is still inconclusive as some researchers have found no efficiency differences between FP and NFP and public providers (Becker and Sloan, 1985; Berta et al., 2010; Herr et al., 2011).

In the current study, the association between foundation and non-foundation status and RCIs are tested, neither of which are FP, as previously mentioned in Chapter 2. However, FTs have considerable freedom as to how they manage their resources and which services they provide (Department of Health, 2002). Thus, in terms of organisational structure and governance, the FTs are similar to private NFP trusts while Non-FTs are more similar to public trusts. Even in this context, the prior results are mixed and inconclusive. Some studies from various countries have found NFP to be more efficient than public; others found that public healthcare providers are more efficient than NFP, while some have found no efficiency differences between the two (e.g., Mutter and Rosko, 2007; Barbetta et al., 2007; Herr, 2008; Tiemann and Schreyögg, 2009; Lindlbauer and Schreyögg, 2014; Sommersguter-Reichmann and Stepan, 2015). Efficiency is key factor in whether or not a trust can achieve their more independent status. Therefore, although they are not privately owned, their level of financial and decision-making freedom means that they are likely to be more efficient than Non-FTs. Therefore, it might be expected that:

H3: The RCIs of foundation trusts are lower than those of non-foundation trusts.

Size has been discussed in terms of scale efficiency, since one large trust may require lower levels of investment in staff and buildings than two smaller ones (Athanassopoulos et al., 1999; Weaver and Deolalikar, 2004). Azevedo and Mateus (2014) have argued in favour of economies of scale for trusts since fixed costs are spread while the volume of output increases. Administrative costs can be reduced; there are greater opportunities for specialisation and the market position of the organisation may be strengthened, allowing improved negotiations with suppliers. This has led to pressure on smaller trusts to merge. Although there is extensive literature on economies of scale, the evidence concerning the effect of size is inconsistent (Azevedo and Mateus, 2014). However, consistent evidence has been found supporting economies of scale for trusts with between two and three hundred beds (Giancotti et al., 2017). Various activities, such as use of labour, may not have many fixed costs which can be 'spread' with an increase in capacity. Thus, there may be a decrease in scale returns. There are some advantages to units remaining smaller because an increase in the size of the trust may lead to problems of co-ordination and congestion (Asmild et al., 2013) and a further disadvantage of mergers is that they may decrease competition (Gaynor et al., 2012). These findings concur with the earlier work of Hefty (1969), who found that the long-term average cost curve for trusts is U-shaped.

H4: RCIs differ between trusts of different sizes.

4. Data and Methodology

4.1. Data

Secondary quantitative data extracted from two NHS websites were used for this study.

MFF adjusted RCIs data for 2009 was gathered from the National Archives of the Department of Health website¹³. Data for 2010-2016 was obtained from the NHS Reference Costs Collection website¹⁴. These data contained the name of the trust and the RCI for each trust for each year and provided information for every year of the Reference Cost system. The RCI data was complemented with the trust characteristics, which are the independent variables to be used in this study. Trust size was assessed for each year of the sample and this data was extracted from the England NHS website¹⁵. The control variables used in this study are gender and patient classification, whether cases were in-patient, out-patient or emergency. Data on these control variables were collected from NHS Digital website from 2009 to 2016¹⁶.

4.2. Dependent Variable

The dependent variable is the trust average cost this was measured by the RCIs of each trust.

4.3. Independent Variables

The independent variables are region, type, foundation status and size. Following the UK Office for National Statistics, ten regions were identified for this study: West Midlands, Wales, South West, South East, North West, North East, Greater London, East of England, East Midlands and Yorkshire and The Humber¹⁷. The region in which each trust is operating was identified by means of the postcode. The second independent variable is the type of trust. There

¹³ This data is available at:

https://webarchive.nationalarchives.gov.uk/+/http://www.dh.gov.uk/en/Managingyourorganisation/Financeandp lanning/NHScostingmanual/index.htm

¹⁴ This data is available at: <u>https://www.gov.uk/government/collections/nhs-reference-costs</u>

¹⁵ This data is available at: <u>https://www.england.nhs.uk/statistics/statistical-work-areas/bed-availability-and-</u>

occupancy/bed-data-overnight/

¹⁶ This data is available at: <u>https://digital.nhs.uk/</u>

¹⁷ Details of these ten regions are available at: <u>https://www.ons.gov.uk/ons/guide-method/geography/beginner-s-guide/maps/regions--former-government-office-regions--gors---effective-at-31st-december--2011.pdf</u>

are five different organisational types within the NHS: acute, specialist, ambulance, community and mental health.

The type of each trust was identified using NHS websites and the websites of the individual institutions. The third independent variable was a foundation status. A trust can only be a FT if it demonstrated best practice. From the names provided for each identification code, it was possible to highlight when a trust received FT status. The fourth independent variable is size. This was measured by the number of available beds overnight. Some researchers mentioned in the literature review used both overnight bed capacity and the number of discharges as their measures. Asmild et al., (2013) used both approaches and found the results to be comparable.

4.4. Control Variables

Patient gender and classification were selected as control variables because according to the literature (e.g., Diwakar et al., 2016; Huang et al., 2017; Bettin et al., 2018), these are factors that have a significant impact on the treatment cost. Gender has been selected as a control variable due to the different medical conditions affecting men and women. Examples from the literature of those who have used gender as a control variable include Titler et al. (2007) Daidone and Street (2013), Castelli et al. (2015b) and Oh et al. (2016).

In terms of classification, patients can be classified in terms of whether they are an in- or out-patient or an accident or emergency case. Out-patients do not stay in trust overnight with clinical supervision, while in-patients are those who stay at least one night in trust (NHS, 2019a). trusts try to reduce the LOS as a cost-cutting strategy and numbers of out-patient surgical procedures are therefore rising (Crawford et al., 2015). considerable differences in cost between in-patient and out-patient surgeries have been noted. For example, Marla and Stallard (2009) have highlighted the savings that can be made when radical mastectomy is carried out as an outpatient procedure rather than keeping patients in trust overnight. Meanwhile, Crawford et al, 2015 found that orthopedic surgeries were less expensive when carried out as outpatient procedures. Using outpatient procedures also reduces scheduling delays and avoids overnight costs (Aronowitz et al., 1998). Oh et al. (2016) found that outpatient costs were 54% lower than those for in-patients for foot and ankle surgery. Huang et al. (2017) demonstrated significant cost savings for out-patients receiving total knee arthroplasty, while ankle surgery following fracture has also been shown, in selected patients, to be more cost-effective than in-patient procedures (Bettin et al., 2018).

Furthermore, uterine polypectomy has been shown to be more cost-effective in an outpatient setting (Diwakar et al., 2016), while breast cancer surgery on an out-patient basis resulted in a 40% saving over an in-patient treatment over two to three days (Marla and Stallard, 2009). However, out-patients were shown to be twice as likely to need further surgery and were three times as likely to suffer post-discharge complications (Lovecchio et al., 2016). Accident and emergency admissions have been shown to result in longer stays in trust than for those patients admitted directly to trust. However, costs may not be significantly lower (Castelli et al., 2015b). A number of reasons have been identified for increased UK Accident and Emergency costs, including the four-hour target-time for waiting for assessment and new General Practitioner (GP) contracts to reduce their out-of-hours obligations (National Audit Office, 2006).

4.5. Methodology

In order to investigate the relationship between the RCI and region, type, size and foundation status, OLS regression was run. The model is as follows:

71
$Cost_t = \beta_0 + \beta_1 Region_j + \beta_2 Type_j + \beta_3 Foundation_t + \beta_4 Size_t + \beta_5 MalePatients_t$

$+ \beta_6 InPatients_t + \beta_7 OutPatients_t + Year FE_t + \epsilon_t$

Where the *Cost* is the RCI, *Region_j* is a dummy variable equals 1 if the trust is located in *j* region and 0 otherwise, *Type_j* is a dummy variable equals 1 if the trust is categorised as *j* type and 0 otherwise, *Foundation* is a dummy variable equals 1 if the trust is foundation and 0 if it is not, *Size* is the natural log of the total number of available beds overnight, *Male Patients* equals total number of male patients divided by the total number of patients (i.e., male and female), *In-Patients* equals total number of in-patients divided by the total number of all patients, *Out-Patients* equals total number of out-patients divided by the total number of all patients.

The common practice with dummy variables, is to take one of each variable separately, as a reference, for each calculation. Thus, the results represent a comparison between those variables entered into the calculation and those retained as a reference, producing a single comparison each time, rather than comparing one to all the others. This procedure was unsuited to the current study. For example, if we took Greater London out, as a reference, the results would be a comparison between Greater London and each individual region, rather than London as compared to the rest of the regions combined. Therefore, to produce a comparison between one region and all the others, the procedure was reversed so that just one region at a time was entered into each calculation and all the rest retained as the reference point. The same procedure was applied for type. This means that ten regressions were run for the regions, five for type and one for foundation status.

Two time periods were chosen for data analysis. The first one was from 2009 to 2016. It would have been preferable to start the period from the introduction of the Reference Cost system in 1997, but at that point, no data was available on patient gender, as this information

was only introduced in 2009. The second one was from 2012 to 2016 which covers the period following the introduction of HRG4+.

5. Results:

Table 1 provides descriptive statistics of the main variables of interest to this study. The sample used was from 2009 to 2016 and the number of observations included in the model was 1,118. The minimum RCI was 78 and the maximum was 139, where the mean was 99. There were more female than male patients by almost 40% (i.e., female patients = 57%; male patients = 43% on average). In terms of patient classification, it was noted that on average, 66% of patients were classified as out-patients, while only 16 % were in-patients, as shown in Table 1.

	RCI	Size	Male Patients %	In-Patients %	Out-Patients %
N	1,118				
Mean	99	691	43	16	66
Std. Deviation	7	368	4	4	10
Minimum	78	0	6	0	1
Maximum	139	2,196	57	27	98

Table 1. Descriptive Statistics of RCIs, Size and the Control Variables from 2009 to 2016

Note: RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients, Out-Patients= the total number of out-patients divided by the total number.

According to **H1**, it was expected that different regions would vary in terms of RCI. The results in Table 2 show that for the overall time period, 2009 - 2016, the North West, East Midlands and Yorkshire and The Humber had RCIs significantly higher when compared with the other regions ($\beta = 2.976$, p-value ≤ 0.01 ; $\beta = 1.649$, p-value ≤ 0.05 ; $\beta = 4.214$, p-value \leq 0.01, respectively). The East of England is the most efficient region being furthest from the RCIs of other regions ($\beta = -2.254$, p-value ≤ 0.01). The South East and South West also fell significantly below the other regions ($\beta = -1.931$, p-value ≤ 0.01 ; $\beta = -1.723$, p-value ≤ 0.01 , respectively). The other regions, including Greater London, were not significantly different. Between 2012 and 2016 (see Appendix 3, Table A), the significant regions remained significant and those which had not been significantly different did not change, except for Wales which became significantly lower than the other nine regions. Based on the results of Table 2, we find support for **H1**.

RCI was expected to differ between trust type, as suggested by **H2**. The results for type (see Table 3) between 2009 and 2016, show that acute and mental health fell statistically significantly below the other types ($\beta = -2.700$, p-value ≤ 0.01 and $\beta = -4.683$, p- value ≤ 0.01 , respectively). Results suggest that the least efficient type were specialist trusts as it presents a coefficient that was higher than the others ($\beta = 6.139$, p-value ≤ 0.01), while the values for the community were significantly higher ($\beta = 2.466$, p-value ≤ 0.10). Between 2012 and 2016 (see Appendix 3, Table B), Specialist remained significantly higher than the national average, despite the introduction of HRG4+, although the coefficient decreased to $\beta = 3.345$. Based on these results, **H2** is supported.

Table 4 presents results for H3, which suggests that FTs were expected to be more efficient than Non-FTs, and H4, which considers the differences in RCIs between trusts of varying sizes. The results show that FTs were more efficient than Non-FTs (β = -2.183, p-value \leq 0.01). These results remain the same between 2012-2016 (see Appendix 3, Table C). The potential effect of size on RCI was significant. Results suggest that the larger the trust, the higher the RCI (β = 4.153, p-value \leq 0.01). Size remains significantly higher between 2012-2016, although the coefficient decreased to β = 2.528. Therefore, H3 and H4 are supported.

Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
West Midlands	0.337									
	(0.493)									
Wales		-3.234								
		(-1.483)								
South West			-1.723***							
			(-2.914)							
South East				-1.931***						
				(-3.535)						
North West					2.976***					
					(6.313)					
North East						-0.025				
						(-0.036)				
Greater London							-0.692			
							(-1.205)			
East of England								-2.254***		
								(-3.544)		
East Midlands									1.649**	
									(2.526)	
Yorkshire and The Humber										4.214***
										(3.014)
Acute	-8.424***	-8.467***	-7.745***	-8.574***	-7.812***	-8.465***	-8.432***	-8.051***	-8.020***	-8.693***
	(-7.086)	(-7.138)	(-6.414)	(-7.260)	(-6.673)	(-7.120)	(-7.106)	(-6.788)	(-6.704)	(-7.335)
Community	-3.801**	-3.863**	-3.241**	-3.454**	-3.392**	-3.839**	-4.338***	-3.632**	-4.173***	-4.577***
	(-2.419)	(-2.464)	(-2.055)	(-2.208)	(-2.198)	(-2.446)	(-2.674)	(-2.326)	(-2.657)	(-2.893)
										(Continued)

Table 2. Results of OLS Regression Test of RCI Relations with Regions from 2009 to 2016

(\mathbf{n})	· •	1
	nfini	ied)
		aca)

Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Mental Health	-12.677***	-12.654***	-11.654***	-11.791***	-12.541***	-12.718***	-13.180***	-12.283***	-12.246***	-12.974***
	(-7.513)	(-7.510)	(-6.782)	(-6.949)	(-7.571)	(-7.506)	(-7.622)	(-7.308)	(-7.240)	(-7.716)
Foundation Trusts	-1.651***	-1.751***	-1.628***	-1.795***	-1.899***	-1.683***	-1.739***	-1.696***	-1.753***	-1.762***
	(-4.001)	(-4.288)	(-4.017)	(-4.430)	(-4.742)	(-4.087)	(-4.257)	(-4.198)	(-4.318)	(-4.346)
Size	4.925***	4.920***	4.256***	4.912***	5.006***	4.918***	5.011***	4.682***	4.554***	5.145***
	(5.305)	(5.306)	(4.473)	(5.323)	(5.490)	(5.213)	(5.381)	(5.060)	(4.864)	(5.547)
Male Patients	0.175***	0.176***	0.199***	0.188***	0.213***	0.177***	0.163***	0.181***	0.169***	0.182***
	(3.947)	(4.021)	(4.497)	(4.313)	(4.901)	(4.046)	(3.609)	(4.156)	(3.855)	(4.175)
In-patients	-0.368***	-0.364***	-0.334***	-0.349***	-0.427***	-0.368***	-0.409***	-0.349***	-0.375***	-0.381***
L	(-5.213)	(-5.150)	(-4.685)	(-4.952)	(-6.098)	(-5.182)	(-5.203)	(-4.954)	(-5.324)	(-5.403)
Out-patients	-0.061***	-0.058**	-0.054**	-0.045*	-0.046**	-0.060***	-0.066***	-0.052**	-0.058**	-0.060***
I	(-2.654)	(-2.555)	(-2.355)	(-1.948)	(-2.041)	(-2.620)	(-2.840)	(-2.298)	(-2.561)	(-2.658)
Intercept	96.542***	96.391***	95.857***	95.090***	93.716***	96.449***	98.100***	95.949***	97.358***	96.060***
Ĩ	(28.516)	(28.533)	(28.405)	(28.097)	(27.977)	(28.475)	(26.919)	(28.512)	(28.716)	(28.505)
Year FE	Yes									
Ν	1,118	1,118	1,118	1,118	1,118	1,118	1,118	1,118	1,118	1,118
Adj R ²	0.119	0.121	0.126	0.129	0.150	0.119	0.120	0.129	0.124	0.126

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and regions during the period from 2009 to 2016. Colum 1 (2 to 10) West Midlands (Wales, South West, South East, North West, North East, Greater London, East of England, East Midlands and Yorkshire and The Humber, respectively) equals 1 if the trust is located in West Midlands (other regions), 0 otherwise. RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients, Out-Patients= the total number of out-patients divided by the total number of all patients. t-statistics of coefficients are in parentheses. *, **, *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.

Variable	(1)	(2)	(3)	(4)
Acute	-2.700***			
	(-2.908)			
Community		2.466*		
		(1.898)		
Mental Health			-4.683***	
			(-3.637)	
Specialist				6.139***
				(5.290)
West Midlands	-3.820**	-3.355**	-3.590**	-4.680***
	(-2.550)	(-2.212)	(-2.400)	(-3.131)
Wales	-7.838***	-7.231***	-7.266***	-8.607***
	(-3.096)	(-2.835)	(-2.873)	(-3.423)
South West	-6.192***	-5.921***	-5.908***	-6.485***
	(-4.283)	(-4.048)	(-4.085)	(-4.526)
South East	-6.188***	-5.615***	-5.492***	-6.722***
	(-4.301)	(-3.863)	(-3.812)	(-4.696)
North West	-1.580	-1.091	-1.249	-2.372*
	(-1.124)	(-0.766)	(-0.890)	(-1.691)
North East	-3.440**	-3.160**	-3.586**	-4.324***
	(-2.293)	(-2.087)	(-2.395)	(-2.892)
Greater London	-4.074***	-3.811***	-4.546***	-5.317***
	(-2.832)	(-2.592)	(-3.175)	(-3.712)
East of England	-6.365***	-6.070***	-6.251***	-6.938***
	(-4.316)	(-4.065)	(-4.245)	(-4.739)
East Midlands	-1.922	-1.505	-1.670	-2.655*
	(-1.299)	(-1.016)	(-1.134)	(-1.800)
Foundation Trusts	-2.592***	-2.237***	-1.998***	-2.713***
	(-6.479)	(-5.464)	(-4.845)	(-6.846)
Size	1.276	0.476	0.900	2.882***
	(1.510)	(0.616)	(1.156)	(3.179)
Male Patients	0.256***	0.263***	0.250***	0.230***
	(5.654)	(5.795)	(5.514)	(5.083)
In-Patients	-0.143**	-0.214***	-0.410***	-0.306***
	(-2.264)	(-3.879)	(-5.915)	(-5.735)
Out-Patients	0.050**	0.035*	-0.009	0.011
	(2.474)	(1.800)	(-0.395)	(0.542)
Intercept	91.178***	92.200***	98.045***	91.200***
	(26.141)	(26.305)	(28.265)	(27.292)
Year FE	Yes	Yes	Yes	Yes
Ν	1,118	1118	1118	1118
Adi R ²	0.151	0.147	0.154	0.165

Table 3. Results of OLS Regression Test of RCI Relations with Trust Type fro	om 2009 to 2016

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and types during the period from 2009 to 2016. Colum 1 (2 to 4) Acute (Community, Mental Health and Specialist, respectively) equals 1 if the trust is Acute (other types), 0 otherwise. RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients. t-statistics of coefficients are in parentheses. *, **, *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.

Variable	Coefficient	t-Statistic
Foundation Trusts	-2.183***	-5.293
Size	4.153***	4.354
West Midlands	-4.154***	-2.780
Wales	-7.739***	-3.088
South West	-5.740***	-3.997
South East	-5.979***	-4.175
North West	-1.823	-1.300
North East	-4.100***	-2.746
Greater London	-4.854***	-3.333
East of England	-6.330***	-4.317
East Midlands	-2.607*	-1.780
Acute	-6.837***	-5.731
Community	-3.950**	-2.455
Mental Health	-10.535***	-6.037
Male Patients	0.221***	4.902
In-Patients	-0.407***	-5.255
Out-Patients	-0.026	-1.120
Intercept	98.194***	25.966
Year FE	Yes	
Ν	1,118	
Adj R ²	0.178	

 Table 4. Results of OLS Regression Test of RCI Relations with Foundation Status and Size from 2009 to 2016

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and foundation status and size during the period from 2009 to 2016. RCI=reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of out-patients divided by the total number of all patients. *, **, *** Indicate statistical significance at the p-value ≤ 0.10 , ≤ 0.05 and ≤ 0.01 levels, respectively.

6. Discussion

In healthcare both location and provision of specialised services are expected to affect costs according to Palmer (2005). The results set out in Table 2 indicate that there are differences in RCI between trusts, in terms of region. This is in line with the findings of Busse et al. (2008) which suggested that factors such as region or wage levels may play a larger part than the similarity of treatment patterns in cost differences for each care episode. The existence of regional differences supports the view of scholars who argued that taking average costs for the whole of England cannot work due to regional cost discrepancies (e.g., Allen, 2009). In the current study, the results did not show Greater London and the South East to have higher costs than other regions, as had been suggested by Northcott and Llewellyn (2003) and Daidone and Street (2013), who noted that costs remained higher for Greater London despite regional adjustment. This

discrepancy between the results of these two studies and those of the current research can be explained by a number of factors. In the case of Northcott and Llewellyn (2003), the study predated the introduction of HRG4 and the MFF has improved since then. Furthermore, the study was qualitative. Daidone and Street (2013) only used 19 HRGs in their study, so did not cover the full range of activities, and focused on a single year.

The results bear out **H2**, in line with the work of Palmer (2005), demonstrating that specialist trusts have higher costs than other trust types. This may reflect the greater complexity of the service they provide (Daidone and D'Amico, 2009) and the need to account for the treatment of more costly patients (Bojke et al., 2018), despite the idea of specialist trusts providing a "focused factory" (Skinner, 1974). Robinson (2005) suggested that specialist trusts' success (at least in the US) may be due to their ability to choose their markets and the services they provide. One potential solution to the higher costs of providing specialist care is to make a surcharge, similar to the MFF regional adjustment, which is extra to the prospective price if specialist care is found to be more costly compared to other patients within the same payment category (Daidone and Street, 2013). Furthermore, since it is possible that traditional accounting methods under-cost specialist services, the adoption of the ABC technique and PLICS could provide more accurate cost data which might help to address the problem, as discussed by Llewellyn et al. (2016).

FTs were, as expected in **H3**, found to have lower RCIs than Non-FTs. FTs' greater economic freedom allow them to treat private patients and use any tools and equipment purchased for their benefit on publicly funded patients. This might lead to the situation highlighted by Kessler (2007) and Bai (2016) in California, where trusts reduce their public patient numbers because they can charge more for private patients¹⁸.

H4 investigated the potential impact of size of trusts on costs and the results bear out the significance of the size issue, suggesting that the larger the trust the higher its RCI. These results are in line with many other studies which have suggested size to be an important factor to explain trusts' costs. However, results of other studies vary as to whether larger or smaller trusts are more efficient. The current study is consistent with the findings of Asmild et al. (2013), who found larger size to be a problematic, possibly due to congestion problems in larger institutions, where high numbers may lead to bottlenecks in certain treatment areas. In this vein, Giancotti et al. (2017) found that economies of scale are beneficial to efficiency, but only if trusts have between 200 and 300 beds. Additionally, Czypionka et al. (2014) found that small to medium trusts (200-400 beds) were more scale efficient than larger ones, especially those with over 1,000 beds.

6.1 Theoretical Contribution

The findings of Study 2 overall are supported by contingency theory which recognises that cost management practices of organisations are affected by factors such as location, size, type and structure. As a result, no single accounting system to standardise costing can work for all trusts regardless of their circumstances (e.g., Otley, 1980; Haldma and Lääts, 2002; Gerdin and Greve, 2004). The control systems which are applied must be appropriate to the organisation (Islam and Hu, 2012). The results have illustrated that a trust's location, its type, size and foundation status are all associated with its RCI. HRG4+ and the MFF may have gone some way towards addressing this issue. Although this study only examines associations between trust characteristics and RCIs, the results might suggest that work remains to be done to tackle discrepancies, to avoid issues of

¹⁸ Although these studies relate to the differences between private and public trusts in California, the system of governance is comparable to foundation and non-foundation status respectively as discussed in the literature section on page 65.

unfairness arising for the management of certain trusts. If the current system continues to be used without further refinements, some trusts might continue to have above average RCIs because of their characteristics.

As discussed earlier, contingency theory suggests that there is no one ideal way to standardize costing systems which works for all types of organisation. This study makes an important contribution to our understanding of contingency theory because by applying it to healthcare data, I have illustrated and reinforced how variations between organisations mean that "one size fits all" costing systems are problematic, The results show that different types of NHS trusts vary in their capacity to work within the efficiency requirements of the Reference Cost system. Furthermore, trusts in different regions, and with different organisational status vary as to their RCIs and how well they can keep within the necessary cost restrictions. This is in line with contingency theory, but a further issue is that some overarching system is necessary to curb spiraling costs in a healthcare system, even if the structures and needs of different types of trusts make this challenging to achieve. This study has therefore contributed to the theoretical work by highlighting the need to find a system for ensuring efficiency which also takes into account the diversity of organisations. This study does not provide any clear answers to this issue, although it does indicate that the Reference Cost system may be working towards success in this area, despite its limitations. More work is needed to discover how such a system could be further improved, bearing in mind what we learn from contingency theory.

7. Conclusion

The Reference Cost system has been providing cost efficiency data for the NHS since 1997. This study has examined the differences in trust characteristics and RCIs, over the period from 2009 to 2016. It has taken contingency as its theoretical framework, as it suggests that no one costing standardisation process can be applied regardless of an organisation's characteristics. This theory has underpinned the development of the hypotheses tested in the study. The four hypotheses focused on the characteristics of trusts and the differences in their RCI. The data was drawn from National Archives of the Department of Health, NHS Reference Cost Collections, NHS Digital and England NHS websites and analysed using OLS regression. **H1**, that predicted that RCIs differ for trusts in different regions, was supported. Results showed that the East of England was the most efficient region. The results for Greater London suggest that the MFF adjustment for regional differences appear to be working in the region. **H2** was that the RCIs differ for trusts of different types. This hypothesis was supported by the results, which indicated that mental health was the most efficient type of trust and specialist the least of different sizes differ as to their RCI. The results supported **H4**, since the larger the trust, the higher its RCI. The results are in line with contingency theory, which underpinsthis study, demonstrating that variations between trusts are important and that, although the last version of HRG (HRG4+) has been associated with improvements in the RCI system, a greater adjustment may be needed to take into account other variations in trusts.

The results contribute to contingency theory by illustrating that in the healthcare sector there is no one ideal costing standardisation system which is appropriate for all organisations. and that this needs to be taken into account when assessing the cost-effectiveness of trusts with varied remits, of different sizes and structures and in different areas of the country. The Reference Cost system may have gone some way towards addressing this but there are still adjustments to be made to improve it yet further.

One limitation of this study is that it does not take into account the quality of care provided. However, this will be addressed in the next chapter, which will consider this variable in addition to those investigated in the current chapter.

Chapter 4: Reference Cost Index Reduction

and Care Quality of National Health

Service Healthcare

Reference Cost Index Reduction and Care Quality of National Health Service Healthcare

Context: NHS in England introduced PbR in 2002, to enhance the cost efficiency nationally and at individual trusts.

Objective: To examine the relationship between RCI reduction and the quality of trust services provided, using seven quality measures (i.e., mortality rates, SDD, infection rates, LOS, emergency and in-patient and out-patient waiting times).

Data Sources: Data for years between 2010 and 2016 were extracted from two NHS websites. Data for RCIs and foundation status were obtained from the NHS Reference Costs Collection website. Data for the dependent and other control variables were gathered from the National Archives of the Department of Health and the NHS Digital websites.

Study Design: Panel fixed effect regression.

Main Findings: RCI reduction is positively associated with care quality indicators, which are, the trust infection rate and the mean waiting time for in-patients. In terms of PbR trusts with RCIs lower than the national average are more financially efficient without compromising healthcare quality.

Conclusion: The results indicate that there is no negative association between RCI reduction and the healthcare quality measures used in this study. However, a significant limitation of this study is that it only considers acute trusts.

Key Words: NHS, RCI, Payment by results, Cost efficiency, Quality.

1. Introduction

In Chapters 2 and 3 of this thesis, I have discussed healthcare worldwide and examined specifically healthcare costs in the NHS in England and Wales. I have also discussed the financial system introduced in 1997, which was the Reference Cost system and PbR, which followed in 2002. The latter is also known as a fixed-price payment system and is applied in many countries worldwide. In England and Wales, the system is based on the NRCI, which is the total of the averages of the costs of treating patients in specific diagnostic categories (for more details, please see Chapters 1 and 2). Additionally, the relationship between NHS trust characteristics and cost was examined. An underlying assumption in these previous chapters was that there has been a move towards standardisation of RCIs which could be financially beneficial to the healthcare system. However, I did not examine the relationship between RCI reduction and quality in healthcare after the introduction of the Reference Cost system and PbR. These systems were introduced to improve both efficiency and the quality of healthcare in England and Wales. This chapter examines the reduction of RCIs and their relationship with healthcare quality. The study will discuss whether RCI reduction is related to lower healthcare quality.

This chapter contributes to the knowledge in the field of accounting, costing and healthcare quality by investigating the relationship between cost and the quality of services provided. To the researcher's knowledge, this is the first study that has used trust infection rates as a measure of quality both in accounting related studies and healthcare studies investigating the relationship between cost and quality. It is also the first to evaluate quality while making a distinction between those trusts that have lower RCIs than the national average and those which have higher RCIs. Data analysis relies on the panel fixed effect regression with region and year fixed effect. Seven dependent variables are used to measure healthcare quality. Those are mortality rates, the number of patients who were discharged on the same day,

infection rate, mean LOS, and the percentage of emergency patients waiting fewer than four hours for treatment, mean in-patient waiting time and mean out-patient waiting time. The independent variable is cost, measured according to the RCI. Data from the period 2010-2016 were obtained from the NHS Digital website.

The following section provides the background. Section three provides quality definition and measures. Section four provides a review of previous research in this field and discusses the various studies which have assessed healthcare quality. Section five explains and justifies the data and the methodology of the current study. Section six displays the results. Section seven presents the discussion, and the final section provides conclusions and suggestions for future research.

2. Background

The healthcare industry is a rapidly growing sector that accounts for a large proportion of the Gross Domestic Product (GDP) of nations worldwide (Hod, 2016). Governments have therefore tried to introduce measures to curb costs (Figueroa et al., 2016; Kaya Samut and Cafri, 2016), particularly since the global financial crisis which occurred in 2008, adversely affecting different economies and industries all over the world, with the healthcare sector suffering particularly badly (Cantor and Poh, 2018). Countries within the Organisation for Economic Co-operation and Development (OECD) found it necessary to implement various measures to reduce health expenditure to safeguard the sector's future. These strategies included reducing the prices of medical products, cutting healthcare workers' wages, and setting limits on publicly funded healthcare budgets. The effect of the global economic crisis has spurred research into new methods for scrutinizing efficiency and effectiveness in how public money is spent and ensuring control of resource usage (Rondeau and Wagar, 2003).

It is self-evident that healthcare organisations not only need to treat patients but to do so to a high standard. However, there is evidence of considerable variations in the quality of care provided by different trusts and variations in patient experience (Chen et al., 2005; Coelli et al., 2005; Chen, 2006). Treatment quality can be included as an output measure. Thus, a healthcare organisation providing its patients with a superior quality of care can be considered to have produced a higher level of output than one delivering lower quality of care (Castelli et al., 2015a). Public healthcare service aims to optimise its use of resources inpursuit of the nation's health (Hollingsworth, 2008).

It is necessary for healthcare organisations to maximise efficiency and effectiveness, and to lower costs while simultaneously providing quality health services (Kounetas and Papathanassopoulos, 2013). Research has shown that it is often the question of cost which predominates despite efforts to introduce governance systems that are more quality-oriented. For instance, Cardinaels and Soderstrom (2013) have shown that, while patient groups try to put pressure on trusts to include quality, such initiatives are consistently overridden by economic considerations.

The accounting literature has examined the problems and tensions which have arisen, particularly when it has been suggested that accounting practices have effectively altered healthcare's social objectives because the control systems they establish fail to categorise patients appropriately (Miller and O'Leary, 1987; Hayes, 1995; Llewellyn and Northcott, 2005; Kurunmaki and Miller, 2006; Scarparo, 2011). Balancing cost and quality is still problematic (Cardinaels and Soderstrom, 2013; Hassabelnaby, 2014), but despite this, healthcare systems worldwide are introducing quality initiatives. For example, the state of Maryland in the US recently introduced the integration of a quality aspect into its system for pricing and budgeting (Patel et al., 2015; Malmmose and Fouladi, 2019). In Norway, quality

indicators have also been implemented (Østergren, 2009), while in the US customer surveys have been conducted (Pflueger, 2016). Meanwhile, in Iran, an accreditation model has been introduced for their health care providers (Agrizzi et al., 2016).

Dillard et al. (2004) demonstrated that when the continued existence of a public institution requires them to fulfil financial targets, they have no choice but to safeguard their organisation. For trusts this means a concentration on the basic financial issues which are imposed on them by law. The core practice of a healthcare organisation is usually thought to be the care of its patients, but in this case, the organisation is failing to protect this, prioritizing instead the accounting measures of costs and the pursuit of efficiency. The implication of this case is that organisations no longer regard patient treatment as their core service, and the difference between core practices and those which are supportive lacks clarity (Malmmose and Kure, 2020).

In 2008, the Darcy Report set out a number of strategies for the improvement of the NHS's quality of care. This included the drafting of an NHS constitution, and ensuring that any changes were patient-centred, locally-led, and based on clinical needs. Change was also required to address the challenges facing the NHS and the need to improve care quality to provide better value for money for taxpayers. The need to increase patient influence over NHS resources was also recognised. The PbR system meant that resource allocation for trusts was already influenced by patient choices but this was to be enhanced via the payment mechanism's recognition of quality and by strengthening individual control. Greater clarity about what was meant by quality and how it was measured was also recommended, as was the publication of quality data and rewarding of quality practices. From 2010, Quality Accounts were to be presented, taking into account safety, experiences and outcomes. The NHS Choices website was to provide comparative information while The Care Quality Commission (CQC) undertook

independent validation of the performance of service providers using nationally-agreed quality indicators (Department of Health, 2008). From April 2009 the system of payment for NHS service providers aimed to reflect clinical practice, while recognising complexity of care and to encourage innovation. In 2010, the Best Practice Tariffs (BPT) programme was introduced, based on clear evidence of what constituted best practice and to ensure that the setting of tariffs focused on best clinical practice not average cost (Department of Health, 2012).

Commissioning for Quality and Innovation (CQUIN) allowed for extra quality payments to be made as appropriate, according to the NHS Standard Contract. Calculation of a CQUIN scheme's full year financial value was as a percentage of the value for the whole year, taking into account all the healthcare services which were commissioned via the NHS Standard Contract. Payment was conditional upon achievement of the agreed CQUIN goals. Commissioners were required to state what proportion of payment referred to each CQUIN indicator and to be clear about the basis for payments. The CQUIN system encouraged care providers to improve both quality and innovation over and above the Standard Contract's baseline requirements. CQUIN schemes were to be realistic but challenging and it was expected that providers would earn a large percentage of the CQUIN annually. Providers who did not participate in an appropriate national CQUIN scheme did not receive payment (NHS Cambridgeshire Community Services, 2022).

3. Quality Definition and Measures

Issues of characterising quality in healthcare services have received much attention from providers (Sewell, 1997; Alotaibi et al., 2015). However, finding one clear and satisfactory definition of quality, particularly for the healthcare sector, is challenging, due to the range of people involved in the development and delivery of this type of service (Rocha et al., 2013; Rezaee and Karimdadi, 2015), with different levels of community and numerous stakeholders

being involved, from medical and administrative staff (Alotaibi et al., 2020), to taxpayers and regulators (Chang et al., 2002; Smith, 2002; Devlin and Sussex, 2011). There are different perspectives, levels of awareness and expectations, so contradictions can arise (Alotaibi et al., 2020). Furthermore, the professional autonomy of clinicians can make controlling and reviewing their activities difficult (Aidemark, 2001; Zelman et al., 2003; Llewellyn and Northcott, 2005). In addition, trusts are focused on long-term outcomes (Zelman et al., 2003), and financial performance is a means to service provision, not an end. Thus, the numerous healthcare objectives make an evaluation of quality a complex issue (Dixit, 2002; Besley and Ghatak, 2003; Zelman et al, 2003; Llewellyn and Northcott, 2005), and various performance dimensions must be taken into account (Eddy, 1998; McIntyre et al., 2001). These may be contradictory, thus, performing well in one dimension may be to the detriment of performance in another, so it is challenging to establish common, explicit valuations covering each performance dimension (Gutacker and Street, 2018). In order to enhance their performance and optimise outcomes for their patients, healthcare institutions need to investigate the best ways of quantifying performance and how those determinants which are part of the function of health provision can be identified (Cantor and Poh, 2018).

Previous studies have produced controversial results, and this could have arisen due to the lack of generally acceptable tools to measure healthcare performance. Kaplan and Porter (2011) argue that the main problem with healthcare is that the wrong things are evaluated in the wrong ways. Some studies have included quality as a measure of healthcare performance, but others have only used indicators that are cost-oriented and used to highlight financial success. Precise performance evaluation is also hindered by the inconsistent application of inputs and outputs when estimating healthcare performance (Cylus and Dickensheets, 2007). According to Choi et al. (2017), if the evaluation of performance is to be precise, both quality of care and cost indicators should be taken into account in any evaluation.

Three main methods have been used to evaluate the quality of care (Romano and Mutter, 2004). The first uses structural variables which illustrate the circumstances in which care provision is given. This includes both material and human resources, as well as the characteristics of the organisation (Romano and Mutter, 2004). Some researchers have tried to apply structural measures as quality variables, such as teaching status (Taylor et al., 1999; Romano and Mutter, 2004; Rosko and Mutter, 2008) or system-membership status (Rosko, 2001; Bernet et al., 2008). Such organisational characteristics as numbers of medical staff in each department (Chen, 2006), and in-patient numbers have also been used (Linna and Häkkinen, 2006). However, modification of structural measures is challenging, and they do not explain much about the variability which can be noted in processes and outcomes (Mitchell and Shortell, 1997). Neither do they encompass the complex and dynamic characteristics which make up quality care (Romano and Mutter, 2004).

Quality evaluation may also take into consideration those process measures which encompass healthcare content and cover all the activities of healthcare providers (Romano and Mutter, 2004). There are two main categories into which these process measures can be placed, explicit measures and implicit measures. Explicit measures are usually gathered by monitoring whether patients who require specific evidence-based care are being given the necessary treatment. This is usually done by asking patients several yes-or-no questions following their treatment. One example of the development of quality indicators comes from the Centers for Medicare and Medicaid Services (CMS)'s the Health Care Quality Improvement Program, which commenced in 1992 in the United States. The program aimed to help managers of healthcare institutions to identify areas in which the quality of care could be improved (Nayar and Ozcan, 2008). Implicit measures are concerned with the comprehensive assessment of quality by means of experienced professionals conducting a semi-structured review of medical records. Using process measures raises certain practical issues, although the measure can help explain the enigmatic nature of the treatment process. Process measures are more costly to gather than either structural measures or outcome measures. In addition, explicit measures are frequently lacking in predictive validity. On the other hand, implicit measures often lack reliability or suffer from bias arising from the perspective of the reviewers (Romano and Mutter, 2004).

Outcome measures have also been used as quality variables. Such measures explain changes that can be ascribed to healthcare and usually focus on mortality or morbidity as an ultimate outcome measure, which is a matter of great concern to patients (Romano and Mutter, 2004; Rosko and Mutter, 2008). Outcome measures reveal both actions taken for patients and how well these were conducted. Thus, measuring quality is crucial to maintaining healthcare standards in the face of pressures on costs, but research to date has used a variety of metrics for this purpose. This study takes mortality rates, SDD, trust infection rates, LOS, and waiting times as its quality measures. These are fully elucidated in the methodology section below, which also discusses the data used for the analysis.

4. Literature Review

A healthcare provider's performance will probably be affected by its patient profile (Laudicella et al., 2013; Barnett et al., 2015; Greysen et al., 2015). In particular, those providers which have high numbers of patients from low-income groups or many state-funded patients may perform poorly. Earlier work has demonstrated that there is a connection between the socioeconomic status of patients and the outcomes of service use, including readmission (Weissman et al., 1994; Joynt and Jha, 2011; Hu et al., 2014). Prior studies have highlighted the importance of appropriate patient education, and a number of other interventions, which can reduce readmissions (Benbassat and Taragin, 2000). Furthermore, over 50% of the variation in trust readmissions can be attributed to community factors (Herrin et al., 2015).

Much research has considered the relationship between the performance of healthcare institutions and the number of patients with a particular diagnosis or requiring the same procedure. It has been shown that the severity of illness (Andersen and Newman, 2005), as assessed by a weighted severity HRG, may impact negatively on performance (Conway et al., 2015).

Research has elucidated the connections between patient volume and some of the ways in which operational performance improvement has been evaluated, for example, reduction in expenditure and low mortality rates. Links of this type have been explained as "practice makes perfect" (Luft et al., 1987; Theokary and Ren, 2011). Theokary and Ren (2011) have likened this to the economies-of-scale effect and the learning effect. The healthcare industry now generally acknowledges economies of scale, and a vast body of research now indicates that larger-volume trusts perform better financially than those which are smaller (Preyra and Pink, 2006). Learning-by-doing comes from cumulative experience and often drives improved performance. Porter and Teisberg (2004) have shown that the learning effect may account for the superior financial performances of healthcare institutions with larger volumes over those with smaller volumes, arguing that doctors with more experience produce better outcomes and reduce costs at the same time. Within the healthcare sector, there are contradictory opinions about the volume effect (Choi et al., 2017). Theokary and Ren (2011) considered both the positive and negative aspects and discovered that the link between volume and process quality formed an inverted U-shape. While an increase in volume has benefits due to the effect of "practice make perfect", if patient volume is increased beyond a certain level, it may prove detrimental to care quality within the trust. However, Nayar et al. (2013)'s results indicate that, in the United States, small trusts were more efficient and offered higher quality than large trusts.

In order to improve quality, European countries have improved the range of private provision available in their healthcare systems, since the privatisation of healthcare services has been put forward as a way of increasing both quality and efficiency (Kruse et al., 2018). In England and Wales, the government has actively encouraged a policy of NHS trust mergers, as a means of enhancing patients' treatment and experience (Gaynor et al., 2012).

The question of healthcare provider ownership is a major issue, especially the effect on quality (Leys and Toft, 2015). Private trusts might be driven to improve quality to attract more patients, although if demand does not depend on quality, private trusts might not prioritise it (Sloan, 2000; Glaeser and Shleifer, 2001; Brekke et al., 2014). Staff attracted to work in public trusts may be more altruistic workers and focused on quality (Lakdawalla and Philipson, 2006). Evidence from a range of European countries indicates that public healthcare institutions have as good if not better levels of efficiency than private ones. It is difficult to reach a definite conclusion because there are inconsistencies in the quality-of-care evidence which is currently available. It appears that, in Europe, the increase in private healthcare provision is not connected to performance improvement. Mixed evidence was also cited by Eggleston et al. (2008) in their analysis of variations in the quality of care. An overview of systematic reviews concerning quality in public, NFP and FP service providers were produced by Herrera et al. (2014). One of their conclusions was that mortality rates are higher for FP providers. The evidence concerning technical efficiency indicates that no clear conclusion can be reached regarding the cost and/or technical efficiency of FP and NFP and public healthcare institutions. However, public healthcare institutions appear to be equally if not more efficient than private institutions (Berta et al., 2010; Kruse et al., 2018).

Trusts' mortality rates in Germany were examined by (Tiemann and Schreyögg, 2009). Their results indicated that, when controls were in place to account for casemix differences, the mortality rates in FP and NFP trusts were better than those in the public sector. In France, studies showed that the private sector performed worse in terms of quality outcomes. Variations were discovered in the mortality rates depending on the type of healthcare institution. The mortality rates in public (non-teaching) trusts were lower than those of FP trusts (Gobillon and Milcent, 2016). In Italy, higher levels of funding for public provision of healthcare services were associated with a reduction in levels of avoidablemortality (Quercioli et al., 2013).

In the UK, there is little or no difference in LOS, for most diagnostic groups, between private Independent Sector Treatment Centers (ISTCs) and public NHS trusts (Kruse et al., 2018). However, it was shown that, on average, ISTCs had a better level of performance for all dimensions (Gutacker and Street, 2018). These results concur with other research which has shown that ISTCs produce health outcomes that are superior to those of NHS trusts (Browne et al., 2008; Chard et al., 2011), and that patients are discharged sooner (Street et al., 2010; Siciliani et al., 2013). UK patient experiences were the subject of work by Owusu-Frimpong et al. (2010) and Pérotin et al. (2013). Owusu-Frimpong et al. (2010) found that satisfaction rates for patients in ISTCs were higher than for those in publicinstitutions, for example, when trying to attract the attention of doctors. However, no significant difference was discovered by Pérotin et al. (2013) between public and private providers in terms of overall experiences reported by patients. Where variations occurred, these appeared to be associated with other variables, for example, patient characteristics (Pérotin et al., 2013). Furthermore, Fenton et al. (2012); Shoemaker (2012) and Stanowski et al. (2015) have all linked patient satisfaction vith cost and found that better patient satisfaction is associated with higher costs.

The effect of mergers on a wide range of outcomes, such as financial performance, productivity, clinical quality and waiting times, was analysed. The evidence did not suggest any gains from mergers except a reduction in activity (Gaynor et al., 2012). Mergers may not

be the best way of tackling the problem of trusts which under-perform because of the resulting reduction in the potential for competition between trusts (Gaynor et al., 2012). Mergers lead to a reduction in the scale of trusts, overall activity and total staffing levels. Apart from this capacity reduction, the evidence does not suggest that mergers enhance performance. Financial performance often declines after a merger, while patient waiting times lengthen. Furthermore, clinical quality appears not to improve. The results by Gaynoret al. (2012) concur with those which have considered the effects of many healthcare institution mergers in the 1990s, in the US (e.g., Ho and Hamilton, 2000; Krishnan, 2001; Town et al., 2006; Dafhy, 2009) finding scant benefit from consolidation and merger.

There is evidence within the literature that under certain circumstances, such as price regulation and observable quality, it is possible for competition to positively impact quality (Gaynor and Town, 2011). Quality does seem to have been improved in UK NHS trusts as a result of reforms that favour competition (Cooper et al., 2011), and it has been demonstrated that competition leads to improved NHS trust management, and hence to improvements in other outcomes (Bloom et al., 2015). Gaynor et al. (2012) found that there have been indications that quality of care has fallen. LOS is not affected by merger. Four years after merger, mean waiting time increased as well the share of patients waiting for an elective treatment for over 180 days.

PbR is a strategy that has been applied to many healthcare systems globally to increase cost efficiency, provide fair treatment payment and ensure high-quality service and good outcomes for patients (Busse, 2012; Grašič et al., 2015; Bojke et al., 2017; Longo et al., 2017). The issue of ensuring fairness was pioneered in 1983 by the US Medicare system's use of DRG (Jackson, 2001), the British version of this being HRGs, which worked towards fairness in the payment system as discussed by (Grašič et al., 2015; Bojke et al., 2017; Longo et al., 2017; Longo et al., 2017; Longo et al., 2017; Longo et al., 2015; Bojke et al., 2017; Longo et al., 2017; Longo et al., 2017; Longo et al., 2017; Longo et al., 2015; Bojke et al., 2017; Longo et al.

2017), taking into account variations in clinical and running costs (Northcott and Llewellyn, 2002). Thus, payment adjustments of this kind allow additional payments to be made when specialised care is provided. Adjustments to the payments also take into account variations in capital and staffing costs across the country and help to ensure fairness in the reimbursement system, rewarding trusts for care provision rather than their circumstances. Comparable systems are used in numerous other countries around the world (see Lowe and Doolin, 1999; Roeder et al., 2002; Lungen and Lapsley, 2003; Grašič et al., 2015; Holzhacker et al., 2015; Longo et al., 2017). Details of PbR and HRGs were discussed fully in Chapters 1 and 2 of this thesis.

In England and Wales, PbR was introduced in 2002, in line with the "Reforming NHS Financial Flows" policy, which aimed to change decision making and behaviours, improve quality and volume and incentivise cost efficiency. Concerns have been raised about the risks of compromising care quality when fixed-price payment systems require cost reduction measures (Farrar et al., 2009), which may lead to undesirable behaviours when unit costs are used as targets (Dawson and Street, 1998; Appleby and Thomas, 2000). Providers might decrease resources allocated for patient care, negatively impacting quality, or select only patients whose care requirements are less resource-intensive, although the system intends to reduce unit costs by increasing efficiency. Farrar et al. (2009) did not find evidence that quality of care changed following the introduction of PbR. However, in the long term (two years) it was noted that in NHS FTs the quality of care improved, in terms of reduced mortality rates. Evidence that care quality can be adversely affected by fixed-price payment systems has been found in the US, where researchers suggest that quality may have reduced when providers decrease spending to meet average cost targets (Propper et al., 2005). Smith (2002) asserts that the healthcare system in the UK has become one of low cost and poor quality due to the pressure to perform efficiently. The arguments surrounding whether low cost, low quality, or high cost, high quality is best, have been discussed more fully by Culyer (2006), indicating that the best solutions will only be found when both efficiency and quality are taken into consideration.

Appleby et al. (2012) have pointed to the need to evaluate PbR systems alongside other policy instruments, since PbR is only one measure among many in healthcare policy. One payment system is not appropriate to all services, so the system needs to cover a flexible compromise of different approaches and adjustments made according to their impact. They also point to the inevitable trade-offs between objectives, in particular the conflict between cost, quality and supply maintenance. If tariffs are reduced to too low a level, it becomes impossible even for efficient providers to sustain the high quality of their services. PbR may, for example incentivise compromising quality or discharging patients prematurely (Bevan et al., 2014). The loading of increasing numbers of objectives onto the payment system may give rise to unpredictable conflicts and make evaluation difficult (Appleby et al., 2012) and it is problematic to distinguish the impact of PbR from other reforms introduced at the time (Jacobs, 2014). High-quality data and analysis are needed to support the PbR system to avoid lack of compliance or undesired side effects. As Glasziou et al. (2012) have noted, it is possible for financial incentives to enhance clinical quality, but there is also a risk of them becoming a costly diversion. Assessing the impact of PbR in England, Farrar (2010) drew the comparison with Scotland where PbR had not be implemented. However, although LOS in England had reduced rapidly, other factors, such as increased availability of resources in England made it impossible to ascertain PbR's impact on volume of activity. While the Audit Commission (2008) claimed that the introduction of PbR had fostered better financial planning and financial and performance management, Appleby et al. (2012) highlight the lack of quantitative evidence to support this viewpoint or to assess the impact of PbR on quality. The Audit Commission (2008) noted an increase in readmissions but this could not be linked to PbR.

The Health Mandate (2012) noted that trusts had not made full use of the potential for extra payments through CQUIN, which implies a limited beneficial impact for this scheme (Appleby et al., 2012), while Monitor (2012) found both under and over-payments within the system. Petersen et al. (2006) posited that the evidence is challenging to assess due to differences in approach and that rigorous assessment of the impact of PbR on quality is lacking. However, quality itself is not incentivised by PbR because the policy makers assumed that pressure to improve quality would arise from patient and commissioner choices (Appleby et al., 2012). However, Dixon et al. (2010) found choice to be a weak driver of quality. Furthermore, conflict may arise between standards set by the CQC and the National Institute for Health and Clinical Excellence (NICE) and budget constraints on services, which may result in limited access. However, no direct evidence suggests that concentrating on cost reduction has been detrimental to quality (Jacobs, 2014).

Costing, productivity, and the impact of HRG on healthcare have been the subject of much critical discussion. Nevertheless, they remain in use as benchmarking tools. This indicates that more studies are required which address issues of agendas that concentrate on productivity, HRGs, and the influence this has on how well alternative projects manage and whether they succeed or fail. If the maximum number of patients are to receive quality but cost-efficient treatment, in a timely fashion, it is essential to balance cost and quality (Malmmose and Kure, 2020). This study will therefore examine the relationship between RCIs and healthcare quality and answer the question as to whether RCI reduction is associated with lower levels of healthcare quality.

Mortality rates, LOS, waiting times and other variables have been shown to be important in this context, so these will be examined in this study, alongside other variables. Mortality rate is taken as an indicator of differences in care quality in health centre provision (Ferrier and Valdmanis, 1996; Marathe et al., 2007) and the links between healthcare costs and mortality have been a focus of earlier research (see Hussey et al., 2013). Mortality rates are an important factor when analysing technical efficiency (Giuffrida et al., 1999; Amado and Santos, 2009). In addition, this particular variable forms a useful proxy for variability in the quality of care provided (Ferrier and Valdmanis, 1996).

LOS has often been analysed as a factor that may impact trust efficiency (Allen, 2009). It is generally agreed that increasing the LOS negatively affects efficiency (Coulam and Gaumer, 1992; Staat, 2006). However, the only researchers to show a positive effect between these variables are Burgess and Wilson (1998). Some researchers have suggested that pressurising healthcare institutions to reduce LOS could adversely affect the quality of care. The argument, known as "quicker and sicker", suggests that premature discharge means more risk of readmission to trust. For example, Kosecoff et al. (1990) discovered that, after PbR was introduced into the US Medicare system, there was evidence supporting the "quicker and sicker" argument. However, some studies have found no relationship between LOS and readmission rates (Baker et al., 2004; Kaboli et al., 2012); others found that LOS is associated with an increase in readmissions (Bueno et al., 2010), while others suggested that there is an association between longer LOS and higher readmission (Westert et al., 2002). Waiting times, health-related quality of life for patients and access to services are also important dimensions when measuring quality, but these have received scant attention in the literature (Gutacker and Street, 2018).

Rosko and Mutter (2008) expected that improved cost efficiency would result from highquality care as leadership engagement is a key to the delivery of high-quality care. Leaders must be committed to implementing strategies to improve care, monitor their progress and constantly strive to implement initiatives to improve care in their trusts. If leaders are committed in this way to improving care quality, better cost efficiency may also be attained. There is, thus, a positive association between quality of care and healthcare institutions efficiency (Choi et al., 2017).

Following the discussion above, I investigate the following research question:

Research question: Is lower RCI associated with lower levels of healthcare quality among NHS acute trusts?

5. Data and Methodology

Secondary quantitative data was used for this study. It was extracted from two NHS websites. Data for MFF adjusted RCIs and foundation status were obtained from the NHS Reference Costs Collection website¹⁹. Data for the dependent and other control variables were gathered from the NHS Digital website²⁰. The number of observations was 846 and these were all drawn from acute trusts.

In order to measure the quality of healthcare service provided by NHS acute trusts, seven variables were examined. Mortality rates, SDD, infection rate, mean LOS, and the percentage of emergency patients waiting fewer than four hours for treatment, mean in-patient waiting time and mean out-patient waiting time.

Mortality rates are measured by the actual number of deaths divided by the expected number, which is calculated based on patient characteristics, including patients who die in the NHS trusts or within 30 days of discharge. The second variable used was SDD. The number of SDD represents how many patients had surgery and returned home the same day. The third

¹⁹ This data is available at: <u>https://www.gov.uk/government/collections/nhs-reference-costs</u>

²⁰ This data is available at: <u>https://digital.nhs.uk/</u>

variable, infection rate, is calculated by dividing the number of patients with trust acquired infections by the number of the population. In 2014 the NHS calculation method was changed to the number of infected patients divided by the number of beds. However, in this study the number of patients infected in trust was divided by the number of in-patient episodes, to standardise the method across the full period of the research. Infection rate²¹ figures were taken from the National Archives of the Department of Health website²², which provides six monthly updates. In this document, there are figures for different types of infection incidents and different categories of harm (i.e., no harm, low harm, moderate harm, severe harm and death). However, for the current study, the annual figures were required, so the figures were matched by organisation code and added together. It would not have been helpful to separate the different types and categories of harm, so the overall figure for all types and level of infection were used.

LOS represents the number of nights patients remain in the trust. Emergency waiting time was measured by the percentage of patients who were treated withinless than four hours (i.e., the number of patients treated within the four hours divided by the total of a number of emergency cases). In-patient waiting time was measured by the number of months patients waited for treatment. Out-patient waiting time was measured in the number of weeks patients waited for treatment. The natural log was used for SDD and total number of episodes to standardise the data.

Quality was taken as the dependent variable and cost as the independent variable. According to a systematic review carried out by Hussey et al. (2013), evidence for the direction

²¹ Trust-acquired infections

²² This data is available at: https://webarchive.nationalarchives.gov.uk/ukgwa/20171030124143/http://www.nrls.npsa.nhs.uk/resources/

of the relationship between healthcare costs and quality is not consistent. For example, Rosko and Mutter (2008) and Choi et al. (2017), have indicated that improved cost efficiency will result from high-quality care. However, Romley et al. (2011) and Romley etal. (2014) used cost as the independent variable. In this study, it has been demonstrated that striving to be cost-efficient can itself improve care quality. It can be argued, given that the focus of the PbR strategy was specifically a financial initiative to cut healthcare costs, that it is the drive towards more cost-efficient healthcare which has led to the influence on quality. Furthermore, lagged cost data was used because trusts make decisions about how to allocate their resources according to their future plans and intentions. There is therefore a time lag between financial outgoings and the outcome of resource allocation. This is consistent with the methodology employed by Rogers et al. (2017).

Cost was captured by two variables. First, cost was measured by the RCI, not the actual amount of money spent. Second, cost was measured as a dummy variable; the sample was divided into two halves. Those in the group at the high end of the RCI scale were given 1; those at the lower end of the scale were given 0. This second measure of cost make it possible to distinguish between those trusts whose RCIs exceed the national average costs and those which maintain their RCIs below the average, to examine whether PbR, which aims to encourage trusts to reduce costs, is negatively associated with quality. This study compared quality data from one year with financial data from the previous year.

Panel fixed effect regression was run for all the data, using SPSS 25, to test the relationship between cost and healthcare quality, one of the advantages of using this technique is that it controls for all time invariant omitted variables such as the age of trust buildings and equipment. following the formula below: $Y_t = \beta_0 + \alpha_i + \beta_1 Cost_{it-1} + \beta_2 Foundation_{it} + \beta_3 Total Episodes_{it} + \beta_4 InPatients_{it}$

+ β_5 OutPatients_{it} + β_6 Male Patients_{it} + Year FE_t + ϵ_{it}

Where Y is the dependent variable (i.e., mortality rate, the log of SDD, infection rate, mean LOS, emergency waiting time, in-patient waiting time and out-patient waiting time). α_i is the term capturing all individual fixed effects. *Cost* is the lagged RCI, in the first model. In the second model it is a dummy variable (*Cost_dummy*). *Foundation* is a dummy variable equals 1 if the trust is foundation and 0 if it is not. *Size* is the natural log of the number of available bed days. *Total Episodes* is the natural log of number of episodes a trust carries out. *In-Patients* is the number of in- patient episodes divided by the total number of episodes. *Out-Patients* is the number of out-patient episodes divided by the total number of episodes. *Male Patients* is the number of male patients divided by the total number of patients and Year FE is years fixed effect.

6. Results

Tables 1 and 2 present a description of the sample. Table 1, Panel A shows the number of acute trusts in each region from the period from 2010 to 2016, with their percentages in brackets. Table 1 Panel B shows the numbers of acute FTs and the cost_dummy which is the number of acute trusts with RCIs above the national average, with the FT percentages, also in brackets, across the overall period examined.

The number of trusts did not change much in each region during the period except for Greater London which lost 25% of its observations (i.e., from 24 trusts in 2010 to 18 trusts in 2016). The highest was North West and the lowest were Wales and Yorkshire and The Humber with 1 and 2 respectively. No trusts were excluded, but the data only showed 1 acute trust for Wales. The total number of observations decreased during the time period by just 4% (i.e., from 123 trusts in 2010 to 118 trusts in 2016). The percentage of FTs increased to 64% of all

trusts by 2016.

Regions	2010	2011	2012	2013	2014	2015	2016	
West Midlands	10 (8)	10 (8)	10 (8)	10 (8)	10 (8)	10 (8)	10 (8)	
Wales	1(1)	1(1)	1(1)	1(1)	1(1)	1(1)	1(1)	
South West	14 (11)	14 (12)	15 (12)	15 (12)	15 (13)	15 (13)	15 (13)	
South East	15 (12)	15 (13)	15 (12)	15 (12)	14 (12)	14 (12)	13 (11)	
North West	24 (20)	24 (20)	24 (20)	24 (20)	24 (20)	24 (20)	24 (20)	
North East	12 (10)	11 (9)	11 (9)	11 (9)	11 (9)	11 (9)	11 (9)	
Greater London	24 (20)	22 (18)	21 (17)	21 (17)	18 (15)	18 (14)	18 (15)	
East of England	12 (10)	13 (11)	13 (11)	13 (11)	13 (11)	13 (11)	13 (11)	
East Midlands	9 (7)	9 (8)	11 (9)	11 (9)	11 (9)	11 (9)	11 (9)	
Yorkshire and The Humber	2 (2)	2 (2)	2 (2)	2 (2)	2 (2)	2 (2)	2 (2)	
Total	123	121	123	123	119	119	118	
Note: Percentages of number of acute trusts in each region are in parentheses.								

 Table 1. Panel A. Number of National Health Service Acute Trusts in Each Region Included in this Study from 2010 to 2016

Table 1. Panel B. Number of National Health Service Acute Trusts Included in this Study from 2010 to 2016 by Foundation Status and Cost_dummy

	2010	2011	2012	2012	2014	2015	2016
	2010	2011	2012	2015	2014	2013	2010
Foundation Trusts	56 (46)	69 (58)	73 (59)	74 (60)	73 (61)	73 (61)	75 (64)
Cost_dummy	56	51	55	63	69	62	61
Note: Cost_dummy= number of acute trusts that there RCI is above 100. Percentages of number of foundation							

Note: Cost_dummy= number of acute trusts that there RCI is above 100. Percentages of number of foundation trusts (number of foundation trusts divided by total number of trusts) are in parentheses.

Table 2. Descriptive Statistics of the Continuous Variables Used in this Study from 2010 to 2016

Tuble II Desemptive Brutisties o	i the continuous	variables esea in this staay if oni 2010 to 2010					
	Mean	Minimum	Maximum	Standard deviation			
Cost	0.98	0.78	1.25	0.06			
Total Episodes	719,537	209,520	2,099,395	329,570			
In-Patients	16.96	8.05	26.85	2.97			
Out-Patients	66.39	37.79	81.18	5.52			
Male Patients	43.15	32.36	48.27	2.29			
Mortality Rate	1.00	0.54	1.25	0.10			
Same Day Discharge	40,816	5	122,892	20,594			
Infection Rate	0.01	0.00	0.09	0.01			
Mean Length of Stay	4.23	2.90	7.60	0.58			
Emergency Waiting Time	0.91	0.00	0.99	0.07			
In-Patient Waiting Time	49.89	0.00	139.03	11.10			
Out-Patient Waiting Time	39.46	12.10	90.50	10.68			

Note: Cost= lagged reference cost index; Total episodes= the natural log of the number of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of out-patients divided by the total number of all patients; Male Patients= the total number of male patients divided by the total number of patients; Mortality rate= the actual number of deaths divided by the expected number; Same day discharge= The natural log of the number of patients who had surgery and returned home the same day; Infection rate= the number of patients infected divided by the number of in-patient episodes; Mean length of stay= the mean number of nights patients remain within the trust; Emergency waiting times= the percentage of emergency patients who were treated within 4 hours divided by the total number of emergency cases; In-patient waiting time= the number of months in-patients waited for treatment; Out-patients waiting time= the number of weeks out-patients waited for treatment.

Table 2 depicts the descriptive statistics of the continuous variable of the model showing the mean, minimum, maximum and standard deviation for each variable during the period 2010 to 2016. Standard deviation for total episodes and SDD are very high due to the large range between the minimum and the maximum. Therefore, the natural log was used for these two variables in the model. The mean for mortality rates is 100%, indicating that the number of people who died in trusts or after 30 days of their discharge was as expected, with a maximum of 25% higher than expected. There are some trusts with no patients treated in less than four hours in emergencies, although some trusts almost achieved the target of treating all their patients (i.e., 99% of emergency patients) within the four-hour time limit.

Table 3 presents the correlation between the variables used in this study. It shows that mortality rates are negatively correlated with the total number of episodes and out-patients and positively with inpatients and male patients. SDD patients is positively associated with foundation status, total number of episodes and male patients but negatively associated with cost. In terms of infection rate, there is a positive correlation with cost and out-patients. Infection rate is negatively correlated with total number of episodes and in-patients. Mean LOS is also positively associated with cost, out-patients and male patients but negatively associated with foundation status and in-patients. All waiting times are negatively associated with foundation status and all positively associated with total number of episodes.

Table 4 explains the results of the study. The left-hand side of the table shows independent and control variables. Columns 1 to 7 show the dependent variables used, mortality rates, SDD, infection rates, mean LOS, emergency waiting times, in- patient waiting times and out-patient waiting times respectively. Also, Table 4 shows the relationship between cost and healthcare quality. There is a positive relationship between cost and infection rate ($\beta = 0.013 \ p$ -value ≤ 0.1) and between cost and in-patient waiting time ($\beta = 18.906 \ p$ -value \leq

0.05). The table also shows that cost had no significant relationship with the other five quality measures. NHS acute trusts which reduce their costs, the percentage of patients acquiring infections is lower, and the number of months patients waited for in-patient treatment is lower. In general, results indicate that there is no negative relationship between reducing cost and the quality of treatment provided.

Table A in Appendix 4 replicates the estimation regression in Table 4. As a robustness test, the system General Method of Moments (GMM) was conducted. This study follows the estimation strategy proposed by Roodman (2009). The results show that there was no negative relationship between trusts with lower RCIs and the quality measures used in this study. The results show that the RCIs are associated negatively with mortality rate and positively with the SDD which indicate that there no negative relationship between reducing cost and the quality of treatment provided as suggested in the main test, indicating that the findings are robust. Overall, my results on the association between RCI reduction and healthcare quality are robust.

In the first model, the relationship between RCI reduction and healthcare quality was examined. This reduction does not necessarily correspond to national averages. Those which reduce costs could be either above or below the national average. However, in the second model, costs were measured as a dummy variable to distinguish between trusts that have lower RCIs than the national average and those whose RCIs exceed the national average. Trusts that are financially efficient (i.e., their RCI is at the lower end of the scale) have lower in-patient waiting times ($\beta = 1.752$, *p*-value ≤ 0.05) (see Table 5). Another test was run, in which the trusts were divided into thirds rather than halves and the middle third deleted, to allow a clearer examination of those trusts which were at either end of the scale. The results concurred with those of the earlier test and showed only one positive relationship, that being between costs and
in-patient waiting times (see Table B in Appendix 4).

	Cost	Foundation Trusts	Total Episodes	In- Patients	Out- Patients	Male Patients	Mortality Rate	SDD	Infection Rate	Mean LOS	Emergency Waiting Time	In-Patients Waiting Time	Out-Patients Waiting Time
Cost	1	-0.159**	0.123**	-0.132**	0.044	0.052	-0.061	-0.141**	0.113**	0.185**	-0.060	0.015	0.048
Foundation Trusts	-	1	-0.110**	0.203**	0.020	0.162**	0.032	0.111**	0.008	-0 140**	0.090**	-0.076*	-0 119**
Total Episodes		1	1	-0.164**	0.351**	0.144**	-0.173**	0.146**	-0.049	0.014	-0.100**	0.096**	0.115**
In-Patients				1	-0.536**	0.215**	0.206**	0.051	-0.218**	-0.164**	0.021	-0.123**	-0.103**
Out-Patients					1	0.117^{**}	-0.102**	0.021	0.124**	0.124**	-0.058	0.156**	0.176^{**}
Male Patients						1	0.164**	0.124^{**}	-0.034	0.249**	0.043	0.055	-0.015
Mortality Rate							1	-0.145**	-0.001	0.132**	-0.021	0.002	-0.060
SDD								1	0.028	0.039	-0.021	0.078^*	0.047
Infection Rate									1	0.228^{**}	-0.052	0.048	0.108^{**}
Mean LOS										1	-0.062	-0.029	0.019
Emergency													
Waiting Time											1	-0.100**	-0.181**
In-Patients												1	0 170**
Waiting Time												1	0.172
Waiting Time													1
Note: This table preser	nts the Pe	arson correlation	ns between the	variables use	d in the regre	ssions. Cost	= lagged refer	ence cost ind	ex of acute true	sts: Total epis	odes= the natur	al log of the nur	nber
of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of out-patients divided by the total number													

Table 3. Results of Correlations between the Variables used in the Regressions from 2010 to 2016

Note: This table presents the Pearson correlations between the variables used in the regressions. Cost= lagged reference cost index of acute trusts; Total episodes= the natural log of the number of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of male patients divided by the total number of patients; Mortality rate= the actual number of deaths divided by the expected number; Same day discharge= The natural log of the number of patients who had surgery and returned home the same day; Infection rate= the number of patients infected divided by the number of in-patient episodes; Mean length of stay= the mean number of nights patients remain within the trust; Emergency waiting times= the percentage of emergency patients who were treated within 4 hours divided by the total number of months in-patients waited for treatment; Out-patients waiting time= the number of weeks out-patients waited for treatment. * and **, correlation is significant at the 0.05 and 0.01 level, respectively.

Variable	Mortality Rate	SDD	Infection Rate	Mean LOS	Emergency Waiting Time	In-Patients Waiting Time	Out-Patients Waiting Time	
variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Cost t-1	-0.038	0.068	0.013*	0.014	0.011	18.906**	3.527	
	(-0.048)	(0.106)	(0.007)	(0.279)	(0.064)	(7.801)	(7.342)	
Foundation Trusts	0.019	-0.022	0.007*	-0.074	0.015	1.299	-0.321	
	(0.015)	(-0.029)	(0.004)	(-0.075)	(0.013)	(3.875)	(-2.096)	
Total Episodes	-0.019	0.032	0.007**	0.090	-0.025	4.792	3.709	
	(-0.039)	(0.081)	(0.003)	(0.197)	(-0.019)	(5.316)	(4.506)	
In-Patients	-0.002	0.005	-0.000*	-0.055***	-0.002	-0.076	0.151	
	(-0.002)	(0.005)	(-0.000)	(-0.014)	(-0.002)	(-0.510)	(0.346)	
Out-Patients	0.000	-0.001	-0.000	-0.001	-0.002**	0.059	-0.031	
	(0.001)	(-0.003)	(-0.000)	(-0.008)	(-0.001)	(0.237)	(-0.135)	
Male Patients	-0.000	0.000	0.000	0.036**	0.001	0.157	0.160	
	(-0.002)	(0.005)	(0.000)	(0.018)	(0.002)	(0.431)	(0.438)	
Intercept	1.152***	4.256***	-0.043	3.218*	1.220***	-8.902	4.937	
	(0.341)	(0.723)	(-0.026)	(1.889)	(0.196)	(-60.255)	(41.400)	
Trust FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Ν	846	846	846	846	846	846	846	
R-within	0.010	0.008	0.047	0.085	0.268	0.043	0.259	
R-between	0.001	0.000	0.002	0.110	0.053	0.000	0.000	
R-overall	0.007	0.003	0.000	0.085	0.199	0.016	0.068	
Rho	0.801	0.959	0.433	0.839	0.303	0.397	0.716	

Table 4. Results of Panel Fixed Effect Regression of Cost Relations with Healthcare Ouality from 2010 to 2016

Note: Columns 1 to 7 are the dependent variables (i.e., mortality rate, number of same-day discharge patients, infection rate, mean length of stay, percentage of emergency patient waited less than four hours, mean in-patient waiting time and mean out- patient waiting time, respectively). Cost= lagged reference cost index of acute trusts; Total episodes= the natural log of the number of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Male Patients= the total number of male patients divided by the total number of patients; Mortality rate= the actual number of deaths divided by the expected number; Same day discharge= The natural log of the number of patients who had surgery and returned home the same day; Infection rate= the number of patients infected divided by the total number of nights patients remain within the trust; Emergency waiting times= the percentage of emergency patients who were treated within 4 hours divided by the total number of emergency cases; In-patient waiting time= the number of months in-patients waited for treatment; Out-patients waiting time= the number of weeks out-patients waited for treatment. t-statistics of coefficients are in parentheses. *, ** and *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.

Variable	Mortality Rate	SDD	Infection Rate	Mean LOS	Emergency Waiting Time	In-Patients Waiting Time	Out-Patients Waiting Time	
Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Cost_dummy t-1	-0.005	0.012	0.001	0.013	0.006	1.752**	0.184	
	(0.005)	(0.011)	(0.001)	(0.027)	(0.007)	(0.693)	(0.730)	
Foundation Trusts	0.019	-0.022	0.007*	-0.072	0.016	0.866	-0.427	
	(0.015)	(0.029)	(0.004)	(0.074)	(0.013)	(4.020)	(2.104)	
Total Episodes	-0.019	0.03	0.007**	0.086	-0.026	4.957	3.79	
	(0.039)	(0.080)	(0.003)	(0.200)	(0.018)	(5.368)	(4.537)	
In-Patients	-0.002	0.005	-0.000*	-0.055***	-0.002	-0.093	0.151	
	(0.002)	0.005	(0.000)	(0.014)	(0.002)	(0.502)	(0.348)	
Out-Patients	0.001	-0.001	-0.000	-0.001	-0.002**	0.045	-0.031	
	(0.001)	(0.003)	(0.000)	(0.008)	(0.001)	(0.231)	(0.135)	
Male Patients	-0.000	-0.000	0.000	0.035**	0.000	0.161	0.165	
	(0.002)	(0.005)	(0.000)	(0.018)	(0.002)	(0.422)	(0.440)	
Intercept	1.108***	4.346***	-0.032	3.282*	1.252***	9.202	7.704	
	(0.334)	(0.708)	(0.026)	(1.904)	(0.185)	(57.437)	(41.800)	
Trust FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Ν	846	846	846	846	846	846	846	
R-within	0.011	0.009	0.044	0.086	0.269	0.042	0.259	
R-between	0.000	0.000	0.003	0.113	0.041	0.001	0.000	
R-overall	0.006	0.003	0.000	0.087	0.193	0.020	0.068	
Rho	0.801	0.959	0.427	0.839	0.310	0.391	0.716	

Table 5. Results of Panel Fixed Effect Regression of Cost dummy Relations with Healthcare Ouality from 2010 to 2016

Note: Columns 1 to 7 are the dependent variables (i.e., mortality rate, number of same-day discharge patients, infection rate, mean length of stay, percentage of emergency patient waited less than four hours, mean in-patient waiting time and mean out- patient waiting time, respectively). Cost_dummy= lagged reference cost index of acute trusts with reference cost index higher than the average; Total episodes= the natural log of the number of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of out-patients divided by the total number of deaths divided by the expected number; Same day discharge= The natural log of the number of patients who had surgery and returned home the same day; Infection rate= the number of patients infected divided by the total number of in-patient episodes; Mean length of stay= the mean number of nights patients remain within the trust; Emergency waiting times= the percentage of emergency patients who were treated within 4 hours divided by the total number of emergency cases; In-patient waiting time= the number of months in-patients; Out-patients waiting time= the number of months in-patients waiting time= the number of weeks out-patients waited for treatment. t-statistics of coefficients are in parentheses. *, ** and *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.

7. Discussion

The results of the study support the arguments of researchers who claim it is possible to maintain the quality of healthcare, even when costs are reduced. A key factor is the quality measures selected. Previous research has given little attention to the issue of waiting times, which is an important dimension because they affect the quality of life for patients (Gutacker and Street, 2018). Therefore, this study took waiting times as one of its dimensions and found that in-patient waiting times were significantly associated with cost, alongside other quality measure, which were mortality rates, SDD, infection rates and mean LOS.

In this study, the results have indicated that accounting practices and the drive to reduce costs and improve efficiency are not negatively associated with the quality of healthcare given to patients. This concurs with a number of researchers, such as Farrar et al. (2009), who found no evidence that the introduction of PbR affected the quality of care provided and Jacobs (2014) who found no evidence to suggest that a focus on cost reduction compromises quality. Likewise, the current results support the argument made by Rosko and Mutter (2008) that, cost efficiency and high quality care are linked due to the key role of leaders who are committed to applying strategies and initiatives which drive up care quality and achieve greater cost efficiency. This leads to the development of a positive relationship between healthcare efficiency and care quality (Choi et al., 2017). Furthermore, the competition between NHS trusts has been found to foster better quality in terms of patient treatment speeds (Cooper et al., 2011), and the current study supports this viewpoint and that of Gaynor and Town (2011) that competition may positively impact quality. The results also support the findings of Cooper et al. (2011) who demonstrated that there have been quality improvements in UK NHS trusts resulting from measures which favour competition, and that competition has improved NHS trust management, and other outcomes (Bloom et al., 2015).

The study findings contrasts with those of several studies, for example, Dillard et al. (2004), which have expressed concern that the imposition of accounting measures alters the focus of healthcare organisations from patient care as a core service to the pursuit of efficiency as the major goal. Neither do the result support Kosecoff et al. (1990)'s faster sicker findings, for RCIs as used in England and Wales. Thus, the concerns of Malmmose and Kure (2020) that organisations may fail to prioritise their core service of patient care over accounting measures and the pursuit of efficiency are not borne out by the results of the current study.

The results of this study provide a marked contrast with those raising concerns about fixed price payment systems compromising care quality. The views expressed by Dawson and Street (1998) and Appleby and Thomas (2000) that PbR systems requiring cost reduction, could encourage providers to reduce resource allocation for patient care, and hence lower quality were also repudiated by the study results, as were those of Smith (2002) who posited that efficiency performance pressure in the UK healthcare system had created a low cost and poorquality system. Bevan et al. (2014) suggested that reducing tariffs to too low a level may prevent even efficient providers from maintaining high quality of care provision and that PbR may, for example incentivise compromising quality or discharging patients prematurely, although the current study did not specifically test for earlier discharges. However, the results suggest that this is not the case and that efficiency and care quality are complementary. Although this study did not specifically examine satisfaction, it could be assumed that at least some aspects of patient satisfaction would be covered by the quality dimensions used in this study. Therefore, this study does not support the results of prior researchers who suggest that patient satisfaction is linked to higher costs (e.g., Fenton et al., 2012; Shoemaker, 2012; Stanowski et al., 2015).

PbR is an important strategy to enhance cost efficiency in healthcare systems worldwide, to ensure fairness in payments for treatment alongside a high-quality service and favourable patient outcomes (Grašič et al., 2015; Bojke et al., 2017; Longo et al., 2017). The results of the second model study suggest that, in England and Wales, when a distinction is made between those trusts whose RCIs are at the higher end of the scale and those whose RCIs are at the lower end, it can be seen that this goal has been accomplished and that healthcare is now more cost-efficient while maintaining and even improving some aspects of quality of care.

8. Conclusion

In view of the spiraling costs of healthcare globally and the need for greater cost efficiency in health service provision, this study has examined the relationship between RCI and the quality of care provided in the NHS in England and Wales. The seven dependent variables used to represent quality were mortality rates, number of patients who were discharged on the same day, infection rates, mean LOS, and the percentage of emergency patients waiting fewer than four hours for treatment, mean in-patient waiting time and mean out-patient waiting time. Panel fixed effect regression was used and the data, for NHS acute trusts, was taken from the NHS Reference Cost Collection and the NHS Digital website for the period 2010-2016. Having reviewed the literature in this field, the study examined whether reducing costs measured by RCI is negatively associated with the quality of healthcare services provided.

The study concurs with the findings of various previous researchers (e.g., Rosko and Mutter, 2008; Farrar et al., 2009; Jacobs, 2014) who found no evidence that care quality is adversely affected by the introduction of PbR and suggests a positive relationship between healthcare efficiency and care quality (Choi et al., 2017). Thus, the study results go some way to allaying the concerns of (Malmmose and Kure, 2020) that PbR may alter the priorities of

organisations such that accounting measures take precedence over their core service of patient care.

It can be therefore concluded that the introduction of the Reference Cost system and PbR in England and Wales has provided a useful strategy to enhance cost efficiency without compromising healthcare quality and that healthcare is now more cost-efficient and has maintained and even improved some aspects of quality of care.

Some limitations must be noted so that they can be addressed in subsequent research. First, this study only examines data from acute NHS trusts. Generalization of the findings of this study to other contexts might be problematic and should be done with caution. Second, this study does not account for readmission rates. Although some data regarding certain specific diagnoses, such as hip or knee replacements, are available, lack of availability of data regarding general readmission rates did not allow this variable to be considered. Thirdly, this study does not take into account clinical negligence costs as one of the quality variables used, since the data is not readily available for all NHS Trusts for all the years required for the sample. Covering this aspect of costs would be a lengthy project in itself, and it would require many months to collect this data by hand. This issue would be best addressed by a further, dedicated research project. Furthermore, the study focused exclusively on quantitative data. Future research could use the methodology employed here to examine the effect of cost reduction on healthcare quality in other types of trust, such as community, mental health, specialist and ambulance trusts.

Chapter 5: Conclusion

Conclusion

The rationale behind this PhD thesis was to contribute to the knowledge within the field of healthcare accounting, costing and quality in healthcare system. It used NHS trusts in England and Wales as a test bed as it provided one of the largest cost accounting databases for public healthcare in the world. The NHS healthcare budget is large and financed by taxpayers' money. Striving for greater efficiency is therefore of the utmost importance. Furthermore, inefficiency within the healthcare system may result in problems that cause human suffering. This thesis therefore makes an important contribution to the knowledge which can underpin efficiency improvements by considering the trends and patterns produced by costing approaches over a twenty-year period, and by examining the relationships between performance measurement, RCI and healthcare quality. The thesis also contributes to development of contingency theory by applying it to RCI data, since this is the first time, to the author's knowledge, that the theory has been applied to this type of data. Contingency theory suggests that there is no "one size fits all" solution to accounting practices within different organisations, yet the Reference Cost system is based on the assumption that there is a degree of homogeneity between healthcare trusts. Using contingency theory to underpin the investigation in Study 2 has shown that, even in the healthcare sector, there are differences between institutions which need to be taken into account, thus illustrating the usefulness of the theory in this context.

The first study focused on the Reference Cost system, which has been in use in the NHS in England since 1997. The aim of introducing the system was to gather data about treatment costs, to enable comparisons to be made between trusts, efficiency to be improved and any variation in treatment costs to be reduced in line with national average costs. Study 1 sought to describe trends and patterns in cost variations between trusts over the twenty-year period since the introduction of RCIs. The study also examined whether and how RCIs varied across

regions, types and foundation status. Trends were identified by means of descriptive statistics. The National Archives of the Department of Health and the NHS Reference Cost Collections websites were used as a data source and analysis was conducted by means of ANOVA and one-sample t-test. The findings of the study showed that, during those two decades, overall variations between trusts have been reduced. This indicates that since the Reference Cost system was introduced costs between trusts have been standardized. However, amongst the characteristics tested, which were region, type and foundation status, there were found to be significant differences. However, the study did not establish the relationship between organisational characteristics and RCIs, because that would have required further statistical tests to be run on the data. This was to be the focus of the second study in the thesis.

In Study 2, which was rooted in contingency theory, four hypotheses were tested to focus on the relationship of trust characteristics on their RCI. H1 RCIs differ between regions. H2 RCIs differ between trust types. H3 FTs have lower RCIs than Non-FTs. H4 RCIs vary according to trust size. The data source was The National Archives of the Department of Health and the NHS Reference Cost Collections websites and OLS regression was used for the analysis. The findings confirmed H1, indicating that the most efficient region was the East of England. In the case of Greater London, the results suggest that the MFF adjustment to take into account regional differences, appear to be working successfully, despite claims that the specific circumstances of this region have not been fully taken into account. The results of the analysis also support H2, showing that the most efficient trust type was mental health and the least efficient was specialist. As posited by H3, FTs have lower RCIs than Non-FTs. The results also support H4, showing that the larger the trust, the greater the average costs. The findings concur with contingency theory, showing that differences between trusts are an important factor, although the introduction of the last version of HRG (HRG4+) into the RCI system has made a partial improvement. However, the results of this study indicate that further adjustments are required to take into consideration other differences between trusts.

Study 2 makes an important contribution to contingency theory by illustrating that, even within the public healthcare sector, account must be taken of variations between trusts because there is no perfect "one size fits all" structure suitable for all organisations within the sector. Assessment of cost-effectiveness must consider such factors as trust size and location as well as their different remits. The MFF system has made some progress in addressing this but further adjustments could improve the system.

A limitation of Study 2 was that it did not address issues of the quality of care provided in the NHS. Therefore, Study 3 examines whether reducing RCIs, as required by PbR, is negatively associated with the quality of healthcare services provided. This issue was investigated using seven dependent variables to represent quality: mortality rates, number of patients who were discharged on the same day, infection rates, mean LOS, and the percentage of emergency patients waiting fewer than four hours for treatment, mean in-patient waiting time and mean out-patient waiting time. The focus of the third study was acute trusts and the data were collected from the NHS Reference Cost Collections and NHS Digital websites covering the period 2010-2016. The data were analysed using panel fixed effect regression.

The results of Study 3 demonstrate that cost reduction may be achieved without compromising quality and that a concern with finance and accounting does not relegate patient care to a secondary concern. This finding concurs with earlier researchers who have shown that a focus on efficiency may help to maintain or improve healthcare quality. The results show that limiting costs as required by the PbR system is positively associated with some quality

measures, such as speed of treatment and infection rate, which suggests that PbR is to encouraging efficiency without adversely affecting care quality. The study helps to allay those fears expressed by researchers who have suggested that a focus on efficiency risks reducing the core service of patient care to a secondary consideration.

Some limitations to this research have been identified. Only data from acute NHS trusts was used, so it would be difficult to generalise these results to other contexts. Neither does this study take into consideration readmission rates, due to a lack of availability of data which made it impossible to analyse this variable. Furthermore, the study is exclusively quantitative. Future research might utilise the methodology of Study 3 and apply it to other trust types, such as community, mental health, specialist and ambulance trusts, to assess the association between RCI reduction and healthcare quality in those areas. Other variables might be considered, such as the associations between RCIs and clinical negligence payments or being a Private Finance Initiative (PFI) trust. It would also be useful to find a method of controlling for the effect of other systems, for example best practice tariffs, to help identify the relationship between PbR and trust behaviour. Conducting a cluster analysis, for example comparing teaching and nonteaching trusts or those which provide children's or elderly persons' services with those who do not, would also provide further useful information. Future qualitative research could consider in depth the differences between a trust with an above average RCI and one with a lower than average RCI. It would also be useful to conduct qualitative research on how well clinical staff understand the importance of balancing professionalism and ethics with the need to be cost efficient. This is essential to safeguard the future survival of individual trusts and the NHS system as a whole.

Overall, this thesis has shown that trusts' healthcare costs in England and Wales are

becoming increasingly standardised. However, work still needs to be done in order to make trusts more efficient, taking into account the several uncontrollable variables investigated in this thesis in the form of trust characteristics. It is important to ensure that the drive towards cost reduction and standardisation does not come at the expense of healthcare quality.

References

Abernethy, M. A. (1996) 'Physicians and resource management: the role of accounting and nonaccounting controls', *Financial Accountability and Management*, 12(2), pp. 141-156. doi: 10.1111/j.1468-0408.1996.tb00419.x.

Agrizzi, D., Agyemang, G. and Jaafaripooyan, E. (2016) 'Conforming to accreditation inIranian hospitals', *Accounting Forum*, 40(2), pp. 106-124. doi: 10.1016/j.accfor.2016.02.002.

Aidemark, L. G. (2001) 'The meaning of balanced scorecards in the health care organisations', *Financial Accountability and Management*, 17(1), pp. 23-40. doi: 10.1111/1468-0408.00119.

Aldogan, M., Austill, A. D. and Kocakülâh, M. C. (2014) 'The excellence of activity-based costing in cost calculation: case study of a private hospital in Turkey', *Journal of Health Care Finance*, 41(1)., pp. 1-27

Allen, P. (2009) "Payment by results" in the English NHS: the continuing challenges', *Public Money & Management*, 29(3), pp. 161-166. doi: 10.1080/09540960902891681.

Alotaibi, G., Yusoff, Z. and Taib, C. A. (2020) 'Total quality management: perspective of Saudi healthcare organizations', *Journal of Business Management and Accounting*, 9(1), pp. 89-93 doi: 10.32890/jbma2019.9.1.8769.

Alotaibi, G., G., Mohd Mokhtar, S. S., Taib, C. A. and Yusoff, R. Z. B. (2015) 'The influence of organizational culture and leadership behavior on innovation performance', *International Journal of Business Research*, 15(5), pp. 47-56. doi: 10.18374/ijbr-15-5.5.

Amado, C. A., Dae. F. and Dos Santos, S. P. (2009) 'Challenges for performance assessment and improvement in primary health care: the case of the Portuguese health centres', *Health Policy*, 91(1), pp. 43-56. doi: 10.1016/j.healthpol.2008.11.008.

Andersen, R. and Newman, J. F. (2005) 'Societal and individual determinants of medical care utilization in the United States', *Milbank Quarterly*, 83(4), pp. 95-124. doi: 10.1111/j.1468-0009.2005.00428.x.

Appleby, J., Harrison, T., Hawkins, L., and Dixon, A. (2012) *Payment by results: How can payment systems help to deliver better care?* London: The King's Fund.

Appleby, J. and Thomas, A. (2000) 'Measuring performance in the NHS: what really matters?', *British Medical Journal*, 320(7247), pp. 1464-1467.

Aronowitz, E. R., and Kleinbart, F. A. (1998) 'Outpatient ACL reconstruction using intraoperative local analgesia and oral postoperative pain medication', *Orthopedics*, 21(7), pp. 781-784.

Asmild, M., Hollingsworth, B. and Birch, S. (2013) 'The scale of hospital production in different settings: one size does not fit all', *Journal of Productivity Analysis*, 40(2), pp. 197-206. doi: 10.1007/s11123-012-0332-9.

Atella, V., Belotti, F., Bojke, C., Castelli, A., Grasic, K., Kopinska, J., Mortari, A. and Street,

A (2019) 'How health policy shapes healthcare sector productivity? Evidence from Italy and the UK', *Health Policy*, 123(1), pp. 27-36. doi: 10.1016/j.healthpol.2018.10.016.

Athanassopoulos, A. D., Gounaris, C. and Sissouras, A. (1999) 'A descriptive assessment of the production and cost efficiency of general hospitals in Greece', *Health Care Management Science*, 2(2), pp. 97-106. doi: 10.1023/a:1019023408924.

Audit Commission (2008) *The right result? Payment by results 2003–07*. London. Audit Commission. Available at: https://www.gov.uk/government/organisations/audit-commission

Awashi, V. N. (1994) 'ABC's of activity-based costing', *Industrial Management*, 36(4), pp. 8-11.

Azevedo, H., and Mateus, C. (2014) 'Cost effects of hospital mergers in Portugal', *The European Journal of Health Economics*, 15(9), pp. 999-1010. doi: 10.1007/s10198-013-0552-6.

Baber, W. R., Daniel, P. L. and Roberts, A. A. (2002) 'Compensation to managers of charitable organizations: an empirical study of the role of accounting measures of program activities', *The Accounting Review*, 77(3), pp. 679-693.

Bai, G. (2016) 'Applying variance analysis to understand California hospitals' expense recovery status by patient groups', *Accounting Horizons*, 30(2), pp. 211-223.

Baker, D., Einstadter, D., Husak, S. and Cebul, R. (2004) 'Trends in post discharge mortality and readmissions: has length of stay declined too far?', *Archives of Internal Medicine*, 164(5), pp. 538-544. doi: 10.1001/archinte.164.5.538.

Barbetta, G. P., Turati, G. and Zago, A. M. (2007) 'Behavioral differences between public and private not-for-profit hospitals in the Italian national health service', *Health Economics*, 16(1), pp. 75-96. doi: 10.1002/hec.1143.

Barnett, M. L., Hsu, J. and McWilliams, J. M. (2015) 'Patient characteristics and differences in hospital readmission rates', *Jama Internal Medicine*, 175(11), pp. 1803-1812. doi: 10.1001/jamainternmed.2015.4660.

Barros, C. P., de Menezes, A. G. and Vieira, J. C. (2013) 'Measurement of hospital efficiency, using a latent class stochastic frontier model', *Applied Economics*, 45(1), pp. 47-54. doi: 10.1080/00036846.2011.579061.

Becker, E. R. and Sloan, F. A. (1985) 'Hospital ownership and performance', Economic Inquiry,

23(1), pp. 21-36. doi: 10.1111/j.1465-7295.1985.tb01750.x.

Beenstock, M. and Felsenstein, D. (2007) 'Mobility and mean reversion in the dynamics of regional inequality', *International Regional Science Review*, 30(4), pp. 335-361. doi: 10.1177/0160017607304542.

Beenstock, M. and Felsenstein, D. (2008) 'Regional heterogeneity, conditional convergence and regional inequality', *Regional Studies*, 42(4), pp. 475-488. doi: 10.1080/00343400701291500.

Benbassat, J. and Taragin, M. (2000) 'Hospital readmissions as a measure of quality of health care: advantages and limitations', *Archives of Internal Medicine*, 160(8), pp. 1074-1081. doi: 10.1001/archinte.160.8.1074.

Bernet, P. M., Moises, J. and Valdmanis, V. G. (2011) 'Social efficiency of hospital care delivery: frontier analysis from the consumer's perspective', *Medical Care Research and Review*, CA, 68(1_suppl), pp. 36s-54s. doi: 10.1177/1077558710366267.

Bernet, P. M., Rosko, M. D. and Valdmanis, V. G. (2008) 'Hospital efficiency and debt', *Journal* of *Health Care Finance*, 34(4), pp. 66-88.

Berta, P., Callea, G., Martini, G. and Vittadini, G. (2010) 'The effects of upcoding, cream skimming and readmissions on the Italian hospitals efficiency: a population-based investigation', *Economic Modelling*, 27(4), pp. 812-821. doi: 10.1016/j.econmod.2009.11.001.

Besley, T. and Ghatak, M. (2003) 'Incentives, choice, and accountability in the provision of public services', *Oxford Review of Economic Policy*, 19(2), pp. 235-249. doi: 10.1093/oxrep/19.2.235.

Bettin, C., Nelson, R., Rothberg, D., Barg, A., Lyman, M. and Saltzman, C. (2018) 'Cost comparison of surgically treated ankle fractures managed in an inpatient versus outpatient

setting', *Journal of the American Academy of Orthopaedic Surgeons*, 27(3), pp. e127-e134. doi: 10.5435/jaaos-d-16-00897.

Bevan, G., Karanikolos, M., Exley, J., Nolte, E., Connolly, S., and Mays, N. (2014) *The four health systems of the UK: How do they compare? Research report*. Nuffield Trust and Health Foundation. Available at: https://www.nuffieldtrust.org.uk/research/the-four-health-systems-of-the-uk-how-do-they-compare (Accessed: 1 June 2022).

Blanchflower, D. G. and Oswald, A. J. (2005) 'Regional wages and the need for a better area cost adjustment', *Public Money & Management*, 25(2), pp. 86-88. doi: 10.1111/j.1467-9302.2005.00457.x

Bloom, N., Propper, C., Seiler, S. and Van Reenen, J. (2015) 'The impact of competition on management quality: evidence from public hospitals', *The Review of Economic Studies*, 82(2), pp. 457-489. doi: 10.1093/restud/rdu045.

Bojke, C., Castelli, A., Grasic, K. and Street, A. (2017) 'Productivity growth in the English national health service from 1998/1999 to 2013/2014', *Health Economics*, 26(5), pp. 547-565. doi: 10.1002/hec.3338.

Bojke, C., Grašič, K. and Street, A. (2018) 'How should hospital reimbursement be refined to support concentration of complex care services?', *Health Economics*, 27(1), pp. E26-E38. doi: 10.1002/hec.3525.

Bourn, M. and Ezzamel, M. (1986a) 'Costing and budgeting in the national health service', *Financial Accountability and Management*, 2(1), pp. 53-71. doi: 10.1111/j.1468-0408.1986.tb00256.x.

Bourn, M. and Ezzamel, M. (1986b) 'Organisational culture in hospitals in the national health

service', *Financial Accountability and Management*, 2(3), pp. 203-225. doi: 10.1111/j.1468-0408.1986.tb00265.x.

Brekke, K. R., Gravelle, H., Siciliani, L. and Straume, O. R. (2014) Patient choice, mobility and competition among health care providers, in Levaggi, R. and Montefiori, M. (eds) *Health Care Provision and Patient Mobility*. Milan: Springer, pp. 1–26. doi: 10.1007/978-88-470-5480-6_1.

Brien, P. (2018) *Public spending by country and region*. Westminster, UK: House of Commons Library.

Brimson, J. A., and Antos, J. (1994) Activity-based management: for service industries, government entities, and nonprofit organizations. Hoboken, New Jersey: Wiley.

Broadbent, J. (1992) 'Change in organisations: a case study of the use of accounting information in the NHS', *The British Accounting Review*, 24(4), pp. 343-367. doi: 10.1016/s0890-8389(05)80044-7.

Broadbent, J., Laughlin, R. and Read, S. (1991) 'Recent financial and administrative changes in the NHS: a critical theory analysis', *Critical Perspectives on Accounting*, 2(1), pp. 1-29. doi: 10.1016/1045-2354(91)90017-8.

Browne, J., Jamieson, L., Lewsey, J., Van der Meulen, J., Copley, L. and Black, N. (2008) 'Casemix and patients' reports of outcome in independent sector treatment centres: Comparison with NHS providers', *BMC Health Services Research*, 8(1), pp. 1-7. doi: 10.1186/1472-6963-8-78.

Bueno, H., Ross, J. S., Wang, Y., Chen, J., Vidan, M. T., Normand, S. L. T. and Krumholz, H.
M. (2010) 'Trends in length of stay and short-term outcomes among Medicare patients hospitalized for heart failure, 1993-2006', *Journal of the American Medical Association*, 303(21), pp. 2141-2147. doi: 10.1001/jama.2010.748.

Burgess, J. F. and Wilson, P. W. (1998) 'Variation in inefficiency among us hospitals', *Information Systems and Operational Research*, 36(3), pp. 84-102. doi: 10.1080/03155986.1998.11732348.

Busse, R. (2012) 'Do diagnosis-related groups explain variations in hospital costs and length of stay? - analyses from the EuroDRG project for 10 episodes of care across 10 European countries', *Health Economics*, 21(S2), pp. 1-5. doi: 10.1002/hec.2861.

Busse, R., Schreyögg, J. and Smith, P. C. (2008) 'Variability in healthcare treatment costs amongst nine EU countries – results from the health basket project', *Health Economics*, 17(s1), pp. S1-S8.

Cannavacciuolo, L., Illario, M., Ippolito, A., and Ponsiglione, C. (2015) 'An activity-based costing approach for detecting inefficiencies of healthcare processes', *Business Process Management Journal*, 21(1), pp. 55-79. Doi: 10.1108/BPMJ-11-2013-0144

Cantor, V. J. M. and Poh, K. L. (2018) 'Integrated analysis of healthcare efficiency: asystematic review', *Journal of Medical Systems*, 42(1), pp. 1-23. doi: 10.1007/s10916-017-0848-7.

Cao, P., ToYabe, S. I., and Akazawa, K. (2006) 'Development of a practical costing method for hospitals', *The Tohoku Journal of Experimental Medicine*, 208(3), pp. 213-224.

Cardinaels, E. and Soderstrom, N. (2013) 'Managing in a complex world: accounting and governance choices in hospitals', *European Accounting Review*, 22(4), pp. 647-684.

Castelli, A., Street, A., Verzulli, R. and Ward, P. (2015a) 'Examining variations in hospital productivity in the English NHS', *The European Journal of Health Economics*, 16(3), pp. 243-254. doi: 10.1007/s10198-014-0569-5.

Castelli, A., Daidone, S., Jacobs, R., Kasteridis, P. and Street, A. D. (2015b) 'The determinants

of costs and length of stay for hip fracture patients', *PLoS One*, 10(7), p. E0133545. doi: 10.1371/journal.pone.0133545.

Chalkley, M. and Sussex, J. (2018) *Private provision of publicly funded health care: the economics of ownership*. London: Office of Health Economics. Available at: https://www.ohe.org/publications/private-provision-publicly-fundedhealth-care-economicsownership

Chang, H., Cheng, M. A. and Das, S. (2004) 'Hospital ownership and operating efficiency: evidence from Taiwan', *European Journal of Operational Research*, 159(2), pp. 513-527. doi: 10.1016/s0377-2217(03)00412-0.

Chang, L. C., Lin, S. W. and Northcott, D. N. (2002) 'The NHS performance assessment framework: a "balanced scorecard" approach?', *Journal of Management in Medicine*, 16(4–5), pp. 345-358. doi: 10.1108/02689230210446526.

Chapman, C., Kern, A. and Laguecir, A. (2014) 'Costing practices in healthcare', *Accounting Horizons*, 28(2), pp. 353-364. doi: 10.2308/acch-50713.

Chard, J., Kuczawaski, M., Black, N. and Van der Meulen, J. (2011) 'Outcomes of elective surgery undertaken in independent sector treatment centres and NHS providers in England: audit of patient outcomes in surgery', *BMJ (online)*, 343 (7831), p. 981. doi: 10.1136/bmj.d6404.

Chen, A., Hwang, Y. and Shao, B. (2005) 'Measurement and sources of overall and input inefficiencies: evidences and implications in hospital services', *European Journal of Operational Research*, 161(2), pp. 447-468. doi: 10.1016/j.ejor.2003.09.017.

Chen, A., Sabharwal, S., Akhtar, K., Makaram, N., and Gupte, C. M. (2015) 'Time-driven activity based costing of total knee replacement surgery at a London teaching hospital', *The*

Knee, 22(6), pp. 640-645.

Chen, S. N. (2006) 'Productivity changes in Taiwanese hospitals and the national health insurance', *Service Industries Journal*, 26(4), pp. 459-477. doi: 10.1080/02642060600622355.

Choi, J. H., Park, I., Jung, I. and Dey, A. (2017) 'Complementary effect of patient volume and quality of care on hospital cost efficiency', *Health Care Management Science*, 20(2), pp. 221-231. doi: 10.1007/s10729-015-9348-9.

Chow, C. W., Ganulin, D. and Haddad, K. (1998) 'The balanced scorecard: a potent tool for energizing and focusing healthcare organization management', *Journal of Healthcare Management*, 43(3), pp. 263-280. doi: 10.1097/00115514-199805000-00010.

Chua, W. F. (1995) 'Experts, networks and inscriptions in the fabrication of accounting images: a story of the representation of three public hospitals', *Accounting, Organizations and Society*, 20(2–3), pp. 111-145.

Cinzia, R. (2010) 'Real wages, wage inequality and the regional cost-of-living in the UK', *Empirical Economics*, 52(4), PP. 1309-1335.

Coelli, T. J., Prasada Rao, D. S., O'Donnell, C. J. and Battese, G. (2005) An Introduction to Efficiency and Productivity Analysis. New York: Springer, PP. XVII-349. doi: 10.1007/b136381.

Conway, R., Byrne, D., O'Riordan, D. and Silke, B. (2015) 'Emergency readmissions are substantially determined by acute illness severity and chronic debilitating illness: a single centre cohort study', *European Journal of Internal Medicine*, 26(1), pp. 12-17. doi: 10.1016/j.ejim.2014.12.013.

Cooper, R., and Kaplan, R. S. (1991) 'Profit priorities from activity-based costing', Harvard

Business Review, 69(3), pp. 130-135.

Cooper, Z., Gibbons, S., Jones, S. and McGuire, A. (2011) 'Does hospital competition save lives? Evidence from the English NHS patient choice reforms', *Economic Journal*, 121(554), pp. F228-F260. doi: 10.1111/j.1468-0297.2011.02449.x.

Coulam, R. F. and Gaumer, G. L. (1992) 'Medicare's prospective payment system: a critical appraisal', *Health Care Financing Review*, 1991(S) pp. 45-77.

Cowing, T. G. and Holtmann, A. G. (1983) 'Multiproduct short-run hospital cost functions: empirical evidence and policy implications from cross-section data', *Southern Economic Journal*, 49(3), pp. 637-653.

Crawford, D. C., Li, C. S., Sprague, S. and Bhandari, M. (2015) 'Clinical and cost implications of inpatient versus outpatient orthopedic surgeries: a systematic review of the published literature', *Orthopedic Reviews*, 7(4), p. 6177. doi: 10.4081/or.2015.6177.

Crémieux, P. Y. and Ouellette, P. (2001) 'Omitted variable bias and hospital costs', *Journal of Health Economics*, 20(2), pp. 271-282. doi: 10.1016/s0167-6296(00)00085-0.

Culyer, A. J. (2006) 'The bogus conflict between efficiency and vertical equity', *Health Economics*, 15(11), pp. 1155-1158. doi: 10.1002/hec.1158.

Custer, W. S. and Willke, R. J. (1991) 'Teaching hospital costs: the effects of medical staff characteristics', *Health Services Research*, 25(6), pp. 831-57.

Cylus, J. D. and Dickensheets, B. A. (2007) 'Hospital multifactor productivity: a presentation and analysis of two methodologies', *Health Care Financing Review*, 29(2), pp. 49-64.

Czypionka, T., Kraus, M., Mayer, S. and Rohrling, G. (2014) 'Efficiency, ownership, and

financing of hospitals: the case of Austria', *Health Care Management Science*, 17(4), pp. 331-347. doi: 10.1007/s10729-013-9256-9.

Dafhy, L. (2009) 'Estimation and identification of merger effects: an application to hospital mergers', *Journal of Law and Economics*, 52(3), pp. 523-550. doi: 10.1086/600079.

Daidone, S. and d'Amico, F. (2009) 'Technical efficiency, specialization and ownership form: evidences from a pooling of Italian hospitals', *Journal of Productivity Analysis*, 32(3), pp. 203-216. doi: 10.1007/s11123-009-0137-7.

Daidone, S. and Street, A. (2013) 'How much should be paid for specialized treatment?', *Social Science & Medicine*, 84, pp. 110-118. doi: 10.1016/j.socscimed.2013.02.005.

Dawson, D. and Street, A. (1998) *Reference costs and the pursuit of efficiency in the NHS*. York: The University of York. Available at: https://www.york.ac.uk/che/pdf/DP161.pdf

Dent, J. F. (1990) 'Strategy, organization and control: some possibilities for accounting research', *Accounting, Organizations and Society*, 15(1-2), pp. 3-25. doi: 10.1016/0361-3682(90)90010-r.

Department of Health. (1997) The new NHS 1998 reference costs. UK: Department of Health. Available at: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/f ile/266003/newnhs.pdf

Department of Health. (2002) A short guide to NHS Foundation Trusts. UK: Department of Health. Available at:

https://www.nuffieldtrust.org.uk/files/2019-11/foundation.pdf

Department of Health. (2005) *A short guide to NHS Foundation Trusts*. UK: Department of Health. Available at: https://glostext.gloucestershire.gov.uk/Data/Health%20Overview%20and%20Scrutiny%20Co mmittee/20051107/Agenda/Consultation%20on%20Gloucestershire%20Partnership%20Trust %20s%20Application%20for%20Foundation%20Trusts%20Status%20-%20attachment%202.pdf

Department of Health. (2008) *High quality care for all: NHS next stage review final report*. Available at: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/f ile/228836/7432.pdf (Accessed: 1 June 2022).

Department of Health. (2012) *A simple guide to payment by results*. UK: Department of Health. Available at: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/f ile/213150/PbR-Simple-Guide-FINAL.pdf

Department of Health. (2013) Payment by results PbR and the market forces factor (MFF) in 2013–14. UK: Department of Health. Available at: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/f ile/214906/PbR-and-the-MFF-in-2013-14.pdf

Department of Health. (2014) *Reference costs 2013–14*. UK: Department of Health. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/f ile/380322/01_Final_2013-14_Reference_Costs_publication_v2.pdf

Devlin, N. J. and Sussex, J. (2011) Incorporating multiple criteria in HTA. London: Methods

and Processes.

Dillard, J. F., Rigsby, J. T. and Goodman, C. (2004) 'The making and remaking of organization context: duality and the institutionalization process', *Accounting, Auditing & Accountability Journal*, 17(4), pp. 506-542. doi: 10.1108/09513570410554542.

Diwakar, L., Roberts, T. E., Cooper, N. A. M., Middleton, L., Jowett, S., Daniels, J. and OPT Trial Collaborative Group. (2016) 'An economic evaluation of outpatient versus inpatient polyp treatment for abnormal uterine bleeding', *International Journal of Obstetrics & Gynaecology*, 123(4), pp. 625-631. doi: 10.1111/1471-0528.13434.

Dixit, A. (2002) 'Incentives and organizations in the public sector: an interpretative review', *Journal of Human Resources*, 37(4), pp. 696-727. doi: 10.2307/3069614.

Dixon, A., Robertson, R., Appleby, J., Burge, P., Devlin, N., and Magee, H. (2010) *Patient choice: How patients choose and how providers respond*. London: The King's Fund.

Dixon, J., Street, A. and Allwood, D. (2018) 'Productivity in the NHS: why it matters and what to do next', *British Medical Journal*, 363, p. K4301. doi: 10.1136/bmj.k4301.

Doolin, B. (1999) 'Casemix management in a New Zealand hospital: rationalisation and resistance', *Financial Accountability and Management*, 15(3&4), pp. 397-417. doi: 10.1111/1468-0408.00091.

Dowless, R. M. (1997) 'Using activity-based costing to guide strategic decision making', *Healthcare Financial Management*, 51(6), pp. 86-89.

Dragija, M., and Lutilsky, I. D. (2017) ABC method in healthcare institutions based on business processes, in Bilgin, M. Danis, H. Demir, E. and Can, U. (eds) *Country experiences in economic development, management and entrepreneurship.* Cham: Springer, pp. 539-551. doi:

10.1007/978-3-319-46319-3_33.

Dranove, D. (1988) 'Pricing by non-profit institutions: The case of hospital cost-shifting', *Journal of Health Economics*, 7(1), pp. 47-57. doi: 10.1016/0167-6296(88)90004-5.

Dredge, B. (2003) 'Finance. Spell it out', The Health Service Journal, 113(5879), p. 35.

Drury, C. M. (2013) Management and cost accounting. London: Springer.

Eastaugh, S. R. (1999) 'Managing risk in a risky world', *Journal of Health Care Finance*, 25(3), pp. 10-16.

Eddy, D. M. (1998) 'Performance measurement: problems and solutions', *Health Affairs*, 17(4), pp. 7-25. doi: 10.1377/hlthaff.17.4.7.

Eggleston, K., Shen, Y. C., Lau, J., Schmid, C. H. and Chan, J. (2008) 'Hospital ownership and quality of care: what explains the different results in the literature?', *Health Economics*, 17(12), pp. 1345-1362. doi: 10.1002/hec.1333.

Elba, F., Ciappei, C., Rialti, R. and Zollo, L. (2017) 'Evaluating the quality and effectiveness of regional healthcare systems: a non-parametric analysis of Istat health for all data', *Proceedings of 20th Toulon-Verona Conference*, Verona: University of Verona.

Eldenburg, L. and Kallapur, S. (1997) 'Changes in hospital service mix and cost allocations in response to changes in Medicare reimbursement schemes', *Journal of Accounting and Economics*, 23(1), pp. 31-51.

Eldenburg, L. and Wolcott, S. (2005) Cost management: Measuring, monitoring, and motivating. *Wolcott Lynch Associates*. New Jersey: Wiley.

Ellwood, S., Chambers, N., Llewellyn, S., Begkos, C. and Wood, C. (2015) Debate: Achieving

the benefits of patient-level costing—open book or can't look?. *Public Money & Management*, 35(1), pp. 69-70. doi: 10.1080/09540962.2015.986893.

Emmanuel, C., Otley, D. and Merchant, K. (1990) *Accounting for management control*. US: Springer. doi: 10.1007/978-1-4899-6952-1.

Farrar, S., Yi, D., Sutton, M., Chalkley, M., Sussex, J. and Scott, A. (2009) 'Has payment by results affected the way that English hospitals provide care? Difference-in-differences analysis', *British Medical Journal*, 339, p. B3047. doi: 10.1136/bmj.b3047.

Farsi, M. and Filippini, M. (2008) 'Effects of ownership, subsidization and teaching activities on hospital costs in Switzerland', *Health Economics*, 17(3), pp. 335-350. doi: 10.1002/hec.1268.

Fenton, J. J., Jerant, A. F., Bertakis, K. D. and Franks, P. (2012) 'The cost of satisfaction: a national study of patient satisfaction, health care utilization, expenditures, and mortality', *Archives of Internal Medicine*, 172(5), pp. 405-411. doi: 10.1001/archinternmed.2011.1662.

Ferrari, A. (2006) 'Market oriented reforms of health services: a non-parametric analysis', *The Service Industries Journal*, 26(1), pp. 1-13. doi: 10.1080/02642060500358720.

Farrar, S., Chalkley, M., Yi, D., and Ma, A. (2010) *Payment by results-consequences for key outcomes measures and variations across HRGs, providers and patients: Report to Department of Health.* UK: Department of Health.

Ferrier, G. D. and Valdmanis, V. (1996) 'Rural hospital performance and its correlates', *Journal of Productivity Analysis*, 7(1), pp. 63-80. doi: 10.1007/bf00158477.

Fetter, R. B. and Freeman, J. L. (1986) 'Diagnosis related groups: product line management within hospitals', *Academy of Management Review*, 11(1), pp. 41-54.

Fetter, R. B., Shin, Y., Freeman, J. L., Averill, R. F. and Thompson, J. D. (1980) 'Case mix definition by diagnosis-related groups', *Medical Care*, 18(2), pp. I-53.

Figueroa, J. F., Wang, D. E. and Jha, A. K. (2016) 'Characteristics of hospitals receiving the largest penalties by US pay-for-performance programmes', *BMJ Quality and Safety*, 25(11), pp. 898-900. doi: 10.1136/bmjqs-2015-005040.

Fries, J. F. (1989) 'The compression of morbidity: near or far?', *The Milbank Quarterly*, 67(2), pp. 208-232 doi: 10.2307/3350138.

Gaughan, J., Gutacker, N., Grasic, K., Kreif, N., Siciliani, L. and Street, A. (2018) *Paying for efficiency: Incentivising same-day discharges in the English NHS*. York: Centre for Health Economics, University of York.

Gaynor, M., Laudicella, M. and Propper, C. (2012) 'Can governments do it better? Mergermania and hospital outcomes in the English NHS', *Journal of Health Economics*, 31(3), pp. 528-543. doi: 10.1016/j.jhealeco.2012.03.006.

Gaynor, M. and Town, R. J. (2011) 'Competition in health care markets', in Culyer, A. J. and Newhouse, J. P. (eds.) *Handbook of health economics*. Amsterdam: Elsevier b.v., pp. 499-637. doi: 10.1016/b978-0-444-53592-4.00009-8.

Gebreiter, F. and Ferry, L. (2016) 'Accounting and the 'insoluble' problem of health-care costs', *European Accounting Review*, 25(4), pp.719-733.

Geissler, A., Quentin, W., Scheller-Kreinsen, D. and Busse, R. (2011) Introduction to DRGs in Europe: common objectives across different hospital systems, European observatory on health system and policies series. Maidenhead, Berkshire, UK: Open University Press.

Gerdin, J. and Greve, J. (2004) 'Forms of contingency fit in management accounting research -

a critical review', Accounting, Organizations and Society, 29(3-4), pp. 303-326. doi: 10.1016/s0361-3682(02)00096-x.

Giancotti, M., Guglielmo, A. and Mauro, M. (2017) 'Efficiency and optimal size of hospitals: results of a systematic search', *PLoS One*, 12(3), p. E0174533. doi: 10.1371/journal.pone.0174533.

Giancotti, M. and Mauro, M. (2015) 'Scale efficiency of public hospital: a content analysis of 20 years of research', *World Review of Business Research*, 5(2), pp. 24-38.

Giuffrida, A., Gravelle, H. and Roland, M. (1999) 'Measuring quality of care with routine data: avoiding confusion between performance indicators and health outcomes', *British Medical Journal*, 318(7202), pp. 94-98. doi: 10.1136/bmj.319.7202.94.

Glaeser, E. L. and Shleifer, A. (2001) 'Not-for-profit entrepreneurs', *Journal of Public Economics*, 81(1), pp. 99-115. doi: 10.1016/s0047-2727(00)00130-4.

Glasziou, P. P., Buchan, H., Del Mar, C., Doust, J., Harris, M., Knight, R., and Stockwell, A. (2012) 'When financial incentives do more good than harm: a checklist', *British Medical Journal*, 345(aug13 2), pp. e5047-e5047. doi: 10.1136/bmj.e5047

Gobillon, L. and Milcent, C. (2016) 'Evaluating the effect of ownership status on hospital quality: the key role of innovative procedures', *Annals of Economics and Statistics*, 121/122, p. 161. doi: 10.15609/annaeconstat2009.121-122.161.

Gok, M. S. and Altindağ, E. (2015) 'Analysis of the cost and efficiency relationship: experience in the Turkish pay for performance system', *The European Journal of Health Economics*, 16(5), pp. 459-469. doi: 10.1007/s10198-014-0584-6.

Gosselin, M. (1997) 'The effect of strategy and organizational structure on the adoption and

implementation of activity-based costing', Accounting, Organizations and Society, 22(2), pp. 105-122.

Grašič, K., Mason, A. R. and Street, A. (2015) 'Paying for the quantity and quality of hospital care: the foundations and evolution of payment policy in England', *Health Economics Review*, 5(1), p. 15. doi: 10.1186/s13561-015-0050-x.

Greysen, S. R., Cenzer, I. S., Auerbach, A. D. and Covinsky, K. E. (2015) 'Functional impairment and hospital readmission in Medicare seniors', *Jama Internal Medicine*, 175(4), pp. 559-565. doi: 10.1001/jamainternmed.2014.7756.

Gruca, T. S. and Nath, D. (2001) 'The technical efficiency of hospitals under a single payer system: the case of Ontario community hospitals', *Health Care Management Science*, 4(2), pp. 91-101. doi: 10.1023/a:1011401510010.

Gutacker, N. and Street, A. (2018) 'Multidimensional performance assessment of public sector organisations using dominance criteria', *Health Economics*, 27(2), pp. E13-E27. doi: 10.1002/hec.3554.

Guven-uslu, P. (2005) 'Benchmarking in health services', *Benchmarking: An International Journal*, 12(4), pp. 293-309.

Guven-uslu, P. and Conrad, L. (2008) 'Uses of management accounting information for benchmarking in NHS trusts', *Public Money and Management*, 28(4), pp.239-246.

Haldma, T. and Lääts, K. (2002) 'Contingencies influencing the management accounting practices of Estonian manufacturing companies', *Management Accounting Research*, 13(4), pp. 379-400. doi: 10.1006/mare.2002.0197.

Harrison, S., Pollitt, C. and Simon, T. (1994) 'Controlling health professionals the future of work

and the organization in the NHS', *Physiotherapy*, 80(12), p. 869. doi: 10.1016/s0031-9406(10)60174-5.

Harvey, J. W. and McCrohan, K. F. (1988) 'Fundraising costs societal implications for philanthropies and their supporters', *Business & Society*, 27(1), pp. 15-22.

Hassabelnaby, H. R. (2014) 'Health care efficiency and quality: a global challenge', *International Journal of Management Accounting Research*, 4(1), pp. 1-4.

Hayes, P. (2005) 'Estimating UK regional price indices, 1974–96', *Regional Studies*, 39(3), pp. 333-344. doi: 10.1080/00343400500087216.

Hayes, R. S. (1995) 'The case of McDonnell Douglas Corporation and FASB 106', *Accounting, Auditing & Accountability Journal*, 8(3), pp. 12-33. doi: 10.1108/09513579510094679.

Health Mandate. (2012) Paying for quality: An analysis of the impact of the 2010/11 commissioning for quality and improvement schemes in London. London: Health Mandate.

Hefty, T. R. (1969) 'Returns to scale in hospitals: a critical review of recent research', *Health Services Research*, 4(4), pp. 267-280.

Helmig, B. and Lapsley, I. (2001) 'On the efficiency of public, welfare and private hospitals in Germany over time: a sectoral data envelopment analysis study', *Health Services Management Research*, 14(4), pp. 263-274.doi: 10.1177/095148480101400406.

Herr, A. (2008) 'Cost and technical efficiency of German hospitals: does ownership matter?', *Health Economics*, 17(9), pp. 1057-1071. doi: 10.1002/hec.1388.

Herr, A., Schmitz, H. and Augurzky, B. (2011) 'Profit efficiency and ownership of German hospitals', *Health Economics*, 20(6), pp. 660-674. doi: 10.1002/hec.1622.

Herrera, C. A., Rada, G., Kuhn-Barrientos, L. and Barrios, X. (2014) 'Does ownership matter? An overview of systematic reviews of the performance of private for-profit, private not-forprofit and public healthcare providers', *PLoS One*, 9(12), p. E93456. doi: 10.1371/journal.pone.0093456.

Herrin, J., St Andre, J., Kenward, K., Joshi, M. S., Audet, A. M. J. and Hines, S. C. (2015) 'Community factors and hospital readmission rates', *Health Services Research*, 50(1), pp. 20-39. doi: 10.1111/1475-6773.12177.

Herwartz, H. and Strumann, C. (2012) 'On the effect of prospective payment on local hospital competition in Germany', *Health Care Management Science*, 15(1), pp. 48-62. doi: 10.1007/s10729-011-9180-9.

Ho, V. and Hamilton, B. H. (2000) 'Hospital mergers and acquisitions: does market consolidation harm patients?', *Journal of Health Economics*, 19(5), pp. 767-791. doi: 10.1016/s0167-6296(00)00052-7.

Hod, R. (2016) 'The relationships between hospital service quality and operational efficiency: an alternative model', *Health Economics & Outcome Research: Open Access*, 02(03), pp. 1-7. doi: 10.4172/2471-268x.1000119.

Hoerger, T. J. (1991) "Profit" variability in for-profit and not-for-profit hospitals', *Journal of Health Economics*, 10(3), pp. 259-289.

Hollingsworth, B. (2008) 'The measurement of efficiency and productivity of health care delivery', *Health Economics*, 17(10), pp. 1107-1128. doi: 10.1002/hec.1391.

Holzhacker, M., Krishnan, R. and Mahlendorf, M. D. (2015) 'The impact of changes in regulation on cost behavior', *Contemporary Accounting Research*, 32(2), pp. 534-566. doi:

10.1111/1911-3846.12082.

Homburg, C. (2004) 'Improving activity-based costing heuristics by higher-level cost drivers', *European Journal of Operational Research*, 157(2), pp. 332-343.

Hood, C. (1995) 'The "new public management" in the 1980s: variations on a theme', *Accounting, Organizations and Society*, 20(2–3), pp. 93-109. doi: 10.1016/0361-3682(93)e0001-w.

Hopwood G. A. (1992) 'Accounting calculation and the shifting sphere of the economic', *European Accounting Review*, 1(1), pp. 125-143.

Hsu, J. (2010) *The relative efficiency of public and private service delivery*', *World Health Organization*, 39, pp. 1-9.

Hu, J., Gonsahn, M. D. and Nerenz, D. R. (2014) 'Socioeconomic status and readmissions: evidence from an urban teaching hospital', Health Affairs, 34(5), pp. 778-785. doi: 10.1377/hlthaff.2013.0816.

Huang, A., Ryu, J. J. and Dervin, G. (2017) 'Cost savings of outpatient versus standard inpatient total knee arthroplasty', *Canadian Journal of Surgery*, 60(1), pp. 57-62. doi: 10.1503/cjs.002516.

Hussey, P. S., Wertheimer, S. and Mehrotra, A. (2013) 'The association between health care quality and cost: A systematic review', *Annals of Internal Medicine*. US: American College of Physicians, 158(1), pp. 27-33. doi: 10.7326/0003-4819-158-1-201301010-00006.

Islam, J. and Hu, H. (2012) 'A review of literature on contingency theory in managerial accounting', *African Journal of Business Management*, 6(15), pp. 5159-5164. doi: 10.5897/ajbm11.2764.

Jackson, A. and Lapsley, I. (2003) 'The diffusion of accounting practices in the new "managerial" public sector', *International Journal of Public Sector Management*, 16(4–5), pp. 359-372. doi: 10.1108/09513550310489304.

Jackson, T. (2001) 'Using hybridization patient-level costing data for setting DRG weights: the Victorian (Australia) cost weight studies', *Health Policy*, 56(2), pp. 149-163. doi: 10.1016/s0168-8510(00)00148-2.

Jacobs, K. (1995) 'Budgets: a medium of organizational transformation', *Management Accounting Research*, 6(1), pp. 59-75. doi: 10.1006/mare.1995.1004.

Jacobs, R. (2014) 'Payment by results for mental health services: economic considerations of case-mix funding', *Advances in Psychiatric Treatment*, 20(3), pp. 155-164.

Jones, C. S. (1999) 'Developing financial accountability in British acute hospitals', *Financial Accountability and Management*, 15(1), pp. 1-20. doi: 10.1111/1468-0408.00071.

Jones, C. S. and Dewing, I. P. (1997) 'The attitudes of NHS clinicians and medical managers towards changes in accounting controls', *Financial Accountability and Management*, 13(3), pp. 261-280. doi: 10.1111/1468-0408.00037.

Jones, M. J. and Mellett, H. J. (2007) 'Determinants of changes in accounting practices: accounting and the UK health service', *Critical Perspectives on Accounting*, 18(1), pp. 91-121.

Joynt, K. E. and Jha, A. K. (2011) 'Who has higher readmission rates for heart failure, and why? Implications for efforts to improve care using financial incentives', *Circulation: Cardiovascular Quality and Outcomes*, 4(1), pp. 53-59. doi: 10.1161/circoutcomes.110.950964.

Kaboli, P. J., Go, J. T., Hockenberry, J., Glasgow, J. M., Johnson, S. R., Rosenthal, G. E. and Vaughan-Sarrazin, M. (2012) 'Associations between reduced hospital length of stay and 30-day
readmission rate and mortality: 14-year experience in 129 veterans affairs hospitals', *Annals of Internal Medicine*, 157(12), pp. 837-845. doi:10.7326/0003-4819-157-12-201212180-00003.

Kaplan, R. S. (1990) 'Contribution margin analysis: No longer relevant/strategic cost management: The new paradigm', *Journal of Management Accounting Research*, 2(1), pp. 1-32.

Kaplan, R. S. and Norton, D. P. (1992) 'The balanced scorecard – measures that drive performance', *Harvard Business Review*, 70(1), pp. 71-79.

Kaplan, R. S. and Porter, M. E. (2011) 'How to solve the cost crisis in health care', *Harvard Business Review*, 89(9), pp. 46-52.

Kaya Samut, P. and Cafri, R. (2016) 'Analysis of the efficiency determinants of health systems in OECD countries by DEA and panel tobit', *Social Indicators Research*, 129(1), pp. 113-132. doi: 10.1007/s11205-015-1094-3.

Kessler, D. P. (2007) 'Cost shifting in California hospitals: what is the effect on private payers?' *Sacramento: California Foundation for Commerce and Education*, 7, PP. 1-20.

Khumawala, S. B. and Gordon, T. P. (1997) 'Bridging the credibility of GAAP: Individual donors and the new accounting standards for nonprofit organizations', *Accounting Horizons*, 11(3), p. 45.

Kirton, R., and Hazlehurst, M. (1991) *Activity based costing at the Luton and Dunstable Hospital*. CIMA Mastercourse Presentation, London.

Kohlbacher, M. (2010) 'The effects of process orientation: a literature review', *Business Process Management Journal*, 16(1), pp. 135-152. doi: 10.1108/14637151011017985.

Kosecoff, J., Kahn, K. L., Rogers, W. H., Reinisch, E. J., Sherwood, M. J., Rubenstein, L. V. and Brook, R. H. (1990) 'Prospective payment system and impairment at discharge: the "quicker-and-sicker" story revisited', *Journal of the American Medical Association*, 264(15), pp. 1980-1983. doi: 10.1001/jama.1990.03450150080035.

Kounetas, K. and Papathanassopoulos, F. (2013) 'How efficient are Greek hospitals? A case study using a double bootstrap DEA approach', *The European Journal of Health Economics*, 14(6), pp. 979-994. doi: 10.1007/s10198-012-0446-z.

Krishnan, R. (2001) 'Market restructuring and pricing in the hospital industry', *Journal of Health Economics*, 20(2), pp. 213-237.

Krishnan, R. and Yetman, M. H. (2011) 'Institutional drivers of reporting decisions in nonprofit hospitals', *Journal of Accounting Research*, 49(4), pp. 1001-1039.

Kruse, F. M., Stadhouders, N. W., Adang, E. M., Groenewoud, S. and Jeurissen, P. P. (2018) 'Do private hospitals outperform public hospitals regarding efficiency, accessibility, and quality of care in the European Union? A literature review', *The International Journal of Health Planning and Management*, 33(2), pp. E434-E453. doi: 10.1002/hpm.2502.

Kuntz, L., Scholtes, S. and Vera, A. (2008) 'DRG cost weight volatility and hospital performance', *OR Spectrum*, 30(2), pp. 331-354.

Kurunmaki, L., Lapsley, I. and Melia, K. (2006) *Costs, care and rationing: a comparative study of intensive care in the UK and Finland*. London: Chartered Institute of Management Accountants.

Kurunmaki, L. and Miller, P. (2006) 'Modernising government: the calculating self, hybridization and performance measurement', *Financial Accountability and Management*,

22(1), pp. 87-106. doi: 10.1111/j.0267-4424.2006.00394.x.

Kurunmäki, L. and Miller, P. (2008) 'Counting the costs: the risks of regulating and accounting for health care provision', *Health, Risk & Society*, 10(1), pp. 9-21. doi: 10.1080/13698570701782395.

Labro, E. (2015) 'Health-care costs: discussion of "the impact of changes in regulation on cost behavior", *Contemporary Accounting Research*, 32(2), pp. 567-574.

Lakdawalla, D. and Philipson, T. (2006) 'The nonprofit sector and industry performance', *Journal of Public Economics*, 90(8–9), pp. 1681-1698. doi: 10.1016/j.jpubeco.2005.11.004.

Lanen, W., Anderson, S., and Maher, M. (2013) *Fundamentals of cost accounting*. New York: McGraw-Hill Education.

Lapsley, I. (2001) 'Accounting, modernity and health care policy', *Financial Accountability & Management*, 17(4), pp. 331-350.

Lapsley, I. and Wright, E. (2004) 'The diffusion of management accounting innovations in the public sector: a research agenda', *Management Accounting Research*, 15(3), pp. 355-374. doi: 10.1016/j.mar.2003.12.007.

Laudicella, M., Li Donni, P. and Smith, P. C. (2013) 'Hospital readmission rates: signal of failure or success?', *Journal of Health Economics*, 32(5), pp. 909-921. doi: 10.1016/j.jhealeco.2013.06.004.

Lave, J. R. and Lave, L. B. (1970) 'Hospital cost functions', *The American Economic Review*, 60(3), pp. 379-395.

Lawson, R. A. (2005) 'The use of activity based costing in the healthcare industry: 1994 vs.

2004', Research in Healthcare Financial Management, 10(1), pp. 77-94.

Lee, K. H., Yang, S. B. and Choi, M. (2009) 'The association between hospital ownership and technical efficiency in a managed care environment', *Journal of Medical Systems*, 33(4), pp. 307-315. doi: 10.1007/s10916-008-9192-2.

Leys, C. and Toft, B. (2015) *How safe are NHS patients in private hospitals? Learning from the care quality commission.* Centre for Health and the Public Interest.

Lindlbauer, I. and Schreyögg, J. (2014) 'The relationship between hospital specialization and hospital efficiency: do different measures of specialization lead to different results?', *Health Care Management Science*, 17(4), pp. 365-378. doi: 10.1007/s10729-014- 9275-1.

Linna, M. and Häkkinen, U. (2006) 'Reimbursing for the costs of teaching and research in Finnish hospitals: a stochastic frontier analysis', *International Journal of Health Care* Finance and *Economics*, 6(1), pp. 83-97. doi: 10.1007/s10754-006-6256-z.

Liu, L. Y., and Pan, F. (2007) 'The implementation of activity-based costing in China: An innovation action research approach', *The British Accounting Review*, 39(3), pp. 249-264.

Llewellyn, S. (1998) 'Boundary work: costing and caring in the social services. Accounting, organizations and society', *Pergamon*, 23(1), pp. 23-47. doi: 10.1016/s0361-3682(96)00036-0.

Llewellyn, S. and Northcott, D. (2005) 'The average hospital', *Accounting, Organizations and Society*, 30(6), pp. 555-583.

Llewellyn, S., Chambers, N., Ellwood, S., Begkos, C. and Wood, C. (2016) Patient-level information and costing systems (PLICSs): a mixed-methods study of current practice and future potential for the NHS health economy. *NIHR Journals Library, Southampton (UK)*, 4(31).

Longo, F., Siciliani, L. and Street, A. (2017) 'Are cost differences between specialist and general hospitals compensated by the prospective payment system?', *The European Journal of Health Economics*, 20(1), pp. 7-26. doi: 10.1007/s10198-017-0935-1.

Lovecchio, F., Alvi, H., Sahota, S., Beal, M. and Manning, D. (2016) 'Is outpatient arthroplasty as safe as fast-track inpatient arthroplasty? A propensity score matched analysis', *The Journal of Arthroplasty*, 31(9), pp. 197-201. doi: 10.1016/j.arth.2016.05.037.

Lowe, A. and Doolin, B. (1999) 'Casemix accounting systems: new spaces for action', *Management Accounting Research*, 10(3), pp. 181-201.

Luft, H. S., Hunt, S. S. and Maerki, S. C. (1987) 'The volume-outcome relationship: practicemakes-perfect or selective-referral patterns?', *Health Services Research*, 22(2), pp. 157-82.

Lungen, M. and Lapsley, I. (2003) 'The reform of hospital financing in Germany: an international solution?', *Journal of Health Organization and Management*, 17(5), PP. 360-372. doi: 10.1108/14777260310505138.

Malmi, T. (1999) 'Activity-based costing diffusion across organizations: an exploratory empirical analysis of Finnish firms', *Accounting, Organizations and Society*, 24(8), pp. 649-672.

Malmmose, M. (2018) 'Accounting research on health care – Trends and gaps', *Financial Accountability & Amp*, 35(1), pp. 90-114. doi: 10.1111/faam.12183

Malmmose, M. and Fouladi, N. (2019) 'Accounting facilitating sociopolitical aims: the case of Maryland hospitals', *Financial Accountability and Management*, 35(4), pp. 413-429. doi: 10.1111/faam.12220.

Malmmose, M. and Kure, N. (2020) 'Putting the patient first? The story of a decoupled hospital management quality initiative', *Critical Perspectives on Accounting*, 80. doi:

10.1016/j.cpa.2020.102233.

Marathe, S., Wan, T. T., Zhang, J. and Sherin, K. (2007) 'Factors influencing community health centers' efficiency: a latent growth curve modeling approach', *Journal of Medical Systems*, 31(5), pp. 365-374. doi: 10.1007/s10916-007-9078-8.

Marla, S. and Stallard, S. (2009) 'Systematic review of day surgery for breast cancer', *International Journal of Surgery*, 7(4), pp. 318-323. doi: 10.1016/j.ijsu.2009.04.015.

Masiye, F. (2007) 'Investigating health system performance: an application of data envelopment analysis to Zambian hospitals', *BMC Health Services Research*, 7(1), p. 58. doi: 10.1186/1472-6963-7-58.

McIntyre, D., Rogers, L. and Heier, E. J. (2001) 'Overview, history, and objectives of performance measurement', *Pub Med*, 22(3), pp. 7-21.

McKee, M. (2002) *Hospitals in a Changing Europe*. Buckingham; Philadelphia: Open University Press.

Menke, T. J. (1997) 'The effect of chain membership on hospital costs. Health services research', *Health Research & Educational Trust*, 32(2), pp. 177-96.

Miller, P. and O'Leary, T. (1987) 'Accounting and the construction of the governable person', *Accounting, Organizations and Society*, 12(3), pp. 235-265. doi: 10.1016/0361-3682(87)90039-0.

Mitchell, P. H. and Shortell, S. M. (1997) 'Adverse outcomes and variations in organization of care delivery', *Medical Care*, 35(11 suppl. 1), pp. NS19-NS32. doi: 10.1097/00005650-199711001-00003.

Mitropoulos, P., Kounetas, K. and Mitropoulos, I. (2016) 'Factors affecting primary health care centers' economic and production efficiency', *Annals of Operations Research*, 247(2), pp. 807-822. doi: 10.1007/s10479-015-2056-5.

Mitropoulos, P., Mitropoulos, I. and Giannikos, I. (2013) 'Combining DEA with location analysis for the effective consolidation of services in the health sector', *Computers & Operations Research*, 40(9), pp. 2241-2250. doi: 10.1016/j.cor.2012.01.008.

Modell, S. (2004) 'Performance measurement myths in the public sector: a research note', *Financial Accountability and Management*, 20(1), pp. 39-55. doi: 10.1111/j.1468-0408.2004.00185.x.

Monitor. (2012) An Evaluation of the Reimbursement System for NHS Funded Care. London: Monitor.

Mossialos, E., Dixon, A., Figueras, J., and Kutzin, J. (2002) *Funding Health Care: Options for Europe*. Buckingham; Philadelphia: Open University Press.

Mutter, R. and Rosko, M. (2007) 'The impact of ownership on the cost-efficiency of U.S. hospitals', in Blank, J. and Valdmanis, V. (eds) *Evaluating hospital policy and performance: contributions from hospital policy and productivity research*. New York: Emerald Group Publishing Limited, pp. 113-138. doi: 10.1016/s0731- 2199(07)00007-6.

National Audit Office. (2006) The provision of out-of-hours care in England. London: National Audit Office. Available at: https://www.nao.org.uk/report/the-provision-of-out-of-hours-care-in-england/

Nayar, P., Ozcan, Y. A., Yu, F. and Nguyen, A. T. (2013) 'Benchmarking urban acute care hospitals: efficiency and quality perspectives', *Health Care Management Review*, 38(2), pp.

137-145. doi: 10.1097/hmr.0b013e3182527a4c.

Nayar, P. and Ozcan, Y. A. (2008) 'Data envelopment analysis comparison of hospital efficiency and quality', *Journal of Medical Systems*, 32(3), pp. 193-199. doi: 10.1007/s10916-007-9122-8.

NHS. (2019a) *Going into hospital as an inpatient or outpatient*. Available at: https://www.nhs.uk/using-the-nhs/nhs-services/hospitals/going-into-hospital-as-an-inpatient-or-outpatient/ (Accessed: 22 January 2020).

NHS. (2019b) *NHS business definitions: NHS Trust.* Available at: https://www.datadictionary.nhs.uk/data_dictionary/nhs_business_definitions/n/nhs_trust_de.a sp?shownav=1 (Accessed: 4 September 2019).

NHS Cambridgeshire Community Services. (2022) *Commissioning for quality and innovation*. Available at: https://www.cambscommunityservices.nhs.uk/about-us/priorities-and-how-we-are-doing/commissioning-for-quality-and-innovation-(cquin)HHS (Accessed: 1 June 2022).

NHS Specialised Services. (2019) *Specialised Services*. Available at: https://www.england.nhs.uk/commissioning/spec-services/ (Accessed: 10 August 2019).

Northcott, D. and Llewellyn, S. (2002) 'Challenges in costing health care services', *International Journal of Public Sector Management*, 15(3), pp. 188-203. doi:10.1108/09513550210423361.

Northcott, D. and Llewellyn, S. (2003) 'The "ladder of success" in healthcare: the UK national reference costing index', *Management Accounting Research*, 14(1), pp. 51-66.

Northcott, D. and Llewellyn, S. (2004) *Cost variability in health care*. London: Chartered Institute of Management Accountants.

Office for National Statistics. (2018a) Employee earnings in the UK 2018 employment and

labourmarket.Availableat:https://www.ons.gov.uk/employmentandlabourmarket/peopleinwork/earningsandworkinghours/bulletins/annualsurveyofhoursandearnings/2018 (Accessed: 30 September 2019).

Office for National Statistics. (2018b) Relative regional consumer price levels of goods and
services,UK:2016.Availableat:https://www.ons.gov.uk/economy/inflationandpriceindices/articles/relativeregionalconsumerpricelevelsuk/2016 (Accessed: 19 October 2019).19 October 2019).

Office for National Statistics. (2019) *UK house price index, publications*. Available at: https://www.gov.uk/government/publications/about-the-uk-house-price-index (Accessed: 30 September 2019).

Oh, J., Perlas, A., Lau, J., Gandhi, R. and Chan, V. W. (2016) 'Functional outcome and costeffectiveness of outpatient vs inpatient care for complex hind-foot and ankle surgery. A retrospective cohort study', *Journal of Clinical Anesthesia*, 35, pp. 20-25. doi: 10.1016/j.jclinane.2016.07.014.

O'Reilly, J., Busse, R., Häkkinen, U., Or, Z., Street, A., and Wiley, M. (2012) 'Paying for hospital care: the experience with implementing activity-based funding in five European countries', *Health Economics, Policy and Law*, 7(1), pp. 73-101.

Østergren, K. (2009) 'Management control practices and clinician managers: the case of the Norwegian health sector', *Financial Accountability & Management*, 25(2), pp. 167-195. doi: 10.1111/j.1468-0408.2009.00473.x.

Otley, D. (2016) 'The contingency theory of management accounting and control: 1980–2014', *Management Accounting Research*, 31, pp. 45-62. doi: 10.1016/j.mar.2016.02.001.

Otley, D. T. (1980) 'The contingency theory of management accounting: achievement and prognosis', *Accounting, Organizations and Society*, 5(4), pp. 413-428. doi: 10.1016/0361-3682(80)90040-9.

Owusu-Frimpong, N., Nwankwo, S. and Dason, B. (2010) 'Measuring service quality and patient satisfaction with access to public and private healthcare delivery', *International Journal of Public Sector Management*, 23(3), pp. 203-220. doi:10.1108/09513551011032455.

Palmer, K. (2005) *NHS reform: will it deliver the desired health outcomes*. London: The King's Fund, London.

Patel, A., Rajkumar, R., Colmers, J. M., Kinzer, D., Conway, P. H. and Sharfstein, J. M. (2015)
'Maryland's global hospital budgets – preliminary results from an all-payer model', *New England Journal of Medicine*, 373(20), pp. 1899-1901. doi: 10.1056/nejmp1508037.

Pérotin, V., Zamora, B., Reeves, R., Bartlett, W. and Allen, P. (2013) 'Does hospital ownership affect patient experience? An investigation into public-private sector differences in England', *Journal of Health Economics*, 32(3), pp. 633-646. doi: 10.1016/j.jhealeco.2013.03.003.

Petersen, L. A., Woodard, L. D., Urech, T., Daw, C., and Sookanan, S. (2006) 'Does pay-forperformance improve the quality of health care?', *Annals of Internal Medicine*, 145(4), pp. 265-272.

Pflueger, D. (2016) 'Knowing patients: the customer survey and the changing margins of accounting in healthcare', *Accounting, Organizations and Society*, 53, pp. 17-33. doi: 10.1016/j.aos.2016.08.002.

Pollitt, C., Harrison, S., Hunter, D. and Marnoch, G. (1988) 'The reluctant managers: clinicians and budgets in the NHS', *Financial Accountability and Management*, 4(3), pp. 213-233. doi:

10.1111/j.1468-0408.1988.tb00299.x.

Popesko, B. (2013) 'Specifics of the activity-based costing applications in hospital management', *International Journal of Collaborative Research on Internal Medicine & Public Health*, 5(3), p. 179.

Porter, M. E. and Teisberg, E. O. (2004) Redefining competition in health care. *Harvard Business Review*, 82, pp. 64-77.

Preston, A. M., Cooper, D. J. and Coombs, R. W. (1992) 'Fabricating budgets: a study of the production of management budgeting in the National Health Service', *Accounting, Organizations and Society*, 17(6), pp. 561-593. doi: 10.1016/0361-3682(92)90014-j.

Preyra, C. and Pink, G. (2006) 'Scale and scope efficiencies through hospital consolidations', *Journal of Health Economics*, 25(6), pp. 1049-1068. doi: 10.1016/j.jhealeco.2005.12.006.

Propper, C., Eachus, J., Chan, P., Pearson, N. and Smith, G. D. (2005) 'Access to health care resources in the UK: the case of care for arthritis', *Health Economics*, 14(4), pp. 391-406. doi: 10.1002/hec.978.

Quercioli, C., Messina, G., Basu, S., McKee, M., Nante, N. and Stuckler, D. (2013) 'The effect of healthcare delivery privatisation on avoidable mortality: longitudinal cross-regional results from Italy, 1993-2003', *Journal of Epidemiology and Community Health*, 67(2), pp. 132-138. doi: 10.1136/jech-2011-200640.

Rezaee, M. J. and Karimdadi, A. (2015) 'Do geographical locations affect in hospitals performance? A multi-group data envelopment analysis', *Journal of Medical Systems*, 39(9), p. 85. doi: 10.1007/s10916-015-0278-3.

Rienzo, C. (2017) 'Real wages, wage inequality and the regional cost-of-living in the UK',

Empirical Economics, 52(4), pp. 1309-1335. doi: 10.1007/s00181-016-1122-4.

Robinson, J. C. (2005) 'Entrepreneurial challenges to integrated health care', in Mechanic, D. Rogut, LB. Colby, DC. and Knickman, JR (eds) *Policy Challenges in Modern Health Care*. New Brunswick, NJ: Rutgers University Press, pp. 53-70.

Robra, B. and Werblow, A. (2006) 'Einsparpotenziale im medizinfernen bereich deutscher krankenhäuser – eine regionale effizienzfront-analyse', *Das Gesundheitswesen*, 68(07), p. A105. doi: 10.1055/s-2006-948661.

Rocha, A., Martins, A., Junior, J. C. F., Boulos, M. N. K., Vicente, M. E., Feld, R. andRodríguez-Molinero, A. (2013) 'Innovations in health care services: The CAALYX system', *International Journal of Medical Informatics*, 82(11), pp. E307-E320. doi: 10.1016/j.ijmedinf.2011.03.003.

Roeder, N., Rohell, B. and Hindle, D. (2002) 'Per case payment in Germany: all in a mess', *Australian Health Review*, 25(6), p. 206.

Rogers, A. T., Bai, G., Lavin, R. A. and Anderson, G. F. (2017) 'Higher hospital spending on occupational therapy is associated with lower readmission rates', *Medical Care Research and Review*, 74(6), pp. 668-686. doi: 10.1177/1077558716666981.

Roh, C. Y., Moon, M. J. and Jung, K. (2013) 'Efficiency disparities among community hospitals in Tennessee: do size, location, ownership, and network matter?', *Journal of Health Care for the Poor and Underserved*, 24(4), pp. 1816-1833. doi: 10.1353/hpu.2013.0175.

Romano, P. S. and Mutter, R. (2004) 'The evolving science of quality measurement for hospitals: implications for studies of competition and consolidation', *International Journal of Health Care Finance and Economics*, 4(2), pp. 131-157. doi: 10.1023/b:ihfe.0000032420.18496.a4.

Romley, J. A., Chen, A. Y., Goldman, D. P. and Williams, R. (2014) 'Hospital costs and inpatient mortality among children undergoing surgery for congenital heart disease', *Health Services Research*, 49(2), pp. 588-608. doi: 10.1111/1475-6773.12120.

Romley, J. A., Jena, A. B. and Goldman, D. P. (2011) 'Hospital spending and inpatient mortality: evidence from California', *Annals of Internal Medicine*, 154(3), pp. 160-167. doi: 10.7326/0003-4819-154-3-201102010-00005.

Rondeau, K. V. and Wagar, T. H. (2003) 'Downsizing and organizational restructuring: what is the impact on hospital performance?', *International Journal of Public Administration*, 26(14), pp. 1647-1668. doi: 10.1081/pad-120024416.

Roodman, D. (2009) 'How to do xtabond2: An introduction to difference and system GMM in Stata', *The Stata journal*, 9(1), pp. 86-136.

Rosko, M. D. (1999) 'Impact of internal and external environmental pressures on hospital inefficiency', *Health Care Management Science*, 2(2), pp. 63-74. doi: 10.1023/a:1019031610741.

Rosko, M. D. (2001) 'Cost efficiency of US hospitals: A stochastic frontier approach', *Health Economics*, 10(6), pp. 539-551. doi: 10.1002/hec.607.

Rosko, M. D. and Chilingerian, J. A. (1999) 'Estimating hospital inefficiency: Does case mix matter?', *Journal of Medical Systems*, 23(1), pp. 57-71. doi: 0.1023/a:1020823612156.

Rosko, M. D. and Mutter, R. L. (2008) 'Review: stochastic frontier analysis of hospital inefficiency – a review of empirical issues and an assessment of robustness', *Medical Care Research and Review*, (65)2, pp. 131-166. doi: 10.1177/1077558707307580.

Russell, H., Street, A. and Ho, V. (2016) 'How well do all patient refined-diagnosis-related

groups explain costs of pediatric cancer chemotherapy admissions in the United States?', *Journal of Oncology Practice*, VA, 12(5), pp. E564-E575. doi: 10.1200/jop.2015.010330.

Saltman, R. B. (2003) 'Melting public-private boundaries in European health systems', *European Journal of Public Health*, 13(1), pp. 24-29. doi: 10.1093/eurpub/13.1.24.

Scarparo, S. (2006) 'The integration of clinical and costing information: a comparative study between Scotland and Sweden', *Financial Accountability and Management*, 22(2), pp. 133-155. doi: 10.1111/j.0267-4424.2006.00396.x.

Scarparo, S. (2011) 'Clinical audit, guidelines and standards: a productive relation formanaging clinical practices', *Financial Accountability and Management*, 27(1), pp. 83-101. doi: 10.1111/j.1468-0408.2010.00517.x.

Schneider, J. E., Miller, T. R., Ohsfeldt, R. L., Morrisey, M. A., Zelner, B. A. and Li, P. (2008) 'The economics of specialty hospitals', *Medical Care Research and Review*, 65(5), pp. 531-553. doi: 10.1177/1077558708316687.

Schreyögg, J., Stargardt, T., Tiemann, O. and Busse, R. (2006) 'Methods to determine reimbursement rates for diagnosis related groups (DRG): A comparison of nine European countries', *Health Care Management Science*, 9(3), pp. 215-223. doi: 10.1007/s10729-006-9040-1.

Sewell, N. (1997) 'Continuous quality improvement in acute health care: creating a holistic and integrated approach', *International Journal of Health Care Quality Assurance*, 10(1), pp. 20-26. doi: 10.1108/09526869710159598.

Shactman, D. (2005) 'Specialty hospitals, ambulatory surgery centers, and general hospitals: Charting a wise public policy course', *Health Affairs*, 24(3), pp. 868-873. doi: Shen, Y. C., Eggleston, K., Lau, J. and Schmid, C. H. (2007) 'Hospital ownership and financial performance: what explains the different findings in the empirical literature?', *Inquiry: The Journal of Health Care Organization, Provision, and Financing*, 44(1), pp. 41-68. doi: 10.5034/inquiryjrnl_44.1.41.

Shleifer, A. (1985) 'A theory of yardstick competition', *The Rand Journal of Economics*, 16(3), pp. 319-327.

Shoemaker, W. (2012) 'The cost of quality: how VBP scores correlate with hospital costs', *Healthcare Financial Management: Journal of the Healthcare Financial Management Association*, 66(10), pp. 50-56.

Siciliani, L., Sivey, P. and Street, A. (2013) 'Differences in length of stay for hip replacement between public hospitals, specialised treatment centres and private providers: selection or efficiency?', *Health Economics* (United Kingdom), 22(2), pp. 234-242. doi: 10.1002/hec.1826.

Skinner, W. (1974) 'The focused factory', Harvard Business Review, 52(3), pp. 113-121.

Sloan, F. A. (2000) 'Not-for-profit ownership and hospital behavior', in Culyer A. J. and Newhouse J. P. (eds) *Handbook of health economics*. New York: Elsvier, pp. 1141-1174.

Sloan, F. A., Morrisey, M. A. and Valvona, J. (1988) 'Effects of the Medicare prospective payment system on hospital cost containment: an early appraisal', *The Milbank Quarterly*, 66(2), pp. 191-220.

Smith, P. C. (2002) 'Measuring health system performance', *The European Journal of Health Economics*, 3(3), pp. 145-148. doi: 10.1007/s10198-002-0138-1.

Sommersguter-Reichmann, M. and Stepan, A. (2015) 'The interplay between regulation and efficiency: Evidence from the Austrian hospital inpatient sector', *Socio-economic Planning Sciences*, 52, pp. 10-21. doi: 10.1016/j.seps.2015.09.001.

Staat, M. (2006) 'Efficiency of hospitals in Germany: a DEA-bootstrap approach', *Applied Economics*, 38(19), pp. 2255-2263. doi: 10.1080/00036840500427502.

Staat, M. and Hammerschmidt, M. (2000) Benchmarking the health sector in Germany – an application of data envelopment analysis. Institute for Market-Oriented Management, University Mannheim.

Stanowski, A. C., Simpson, K. and White, A. (2015) 'Pay for performance: are hospitals becoming more efficient in improving their patient experience?', *Journal of Healthcare Management*, 60(4), pp. 268-284. Doi: 10.1097/00115514-201507000-00008.

Steinmann, L. and Zweifel, P. (2003) 'On the (in)efficiency of Swiss hospitals', Applied Economics, 35(3), pp. 361-370. Doi: 10.1080/00036840210167183.

Street, A., Sivey, P., Mason, A., Miraldo, M. and Siciliani, L. (2010) 'Are English treatment centres treating less complex patients?', *Health Policy*, 94(2), pp. 150-157. doi: 10.1016/j.healthpol.2009.09.013.

Street, A. and Maynard, A. (2007) 'Activity based financing in England: the need for continual refinement of payment by results', *Health Economics, Policy and Law*, 2(4), pp. 419-427.

Taylor, D. H., Whellan, D. J. and Sloan, F. A. (1999) 'Effects of admission to a teaching hospital on the cost and quality of care for Medicare beneficiaries', *New England Journal of Medicine*, 340(4), pp. 293-299. doi: 10.1056/nejm199901283400408.

The World Bank. (2015) Domestic general government health expenditure (% of GDP).

Theokary, C. and Justin Ren, Z. (2011) 'An empirical study of the relations between hospital volume, teaching status, and service quality', *Production and Operations Management*, 20(3), pp. 303-318. doi: 10.1111/j.1937-5956.2011.01228.x.

Tiemann, O. and Schreyögg, J. (2009) 'Effects of ownership on hospital efficiency in Germany', *Business Research*, 2(2), pp. 115-145. doi:10.1007/bf03342707.

Tiemann, O. and Schreyögg, J. (2012) 'Changes in hospital efficiency after privatization', *Health Care Management Science*, 15(4), pp. 310-326. doi: 10.1007/s10729-012-9193-z.

Tillema, S. (2010) 'Public sector benchmarking and performance improvement: What is the link and can it be improved?', *Public Money & Management*, 30(1), pp. 69-75.

Titler, M., Dochterman, J., Kim, T., Kanak, M., Shever, L., Picone, D. M. and Budreau, G. (2007) 'Cost of care for seniors hospitalized for hip fracture and related procedures', *Nursing Outlook*, 55(1), pp. 5-14. doi: 10.1016/j.outlook.2006.06.006.

Town, R., Wholey, D., Feldman, R. and Burns, L. R. (2006) 'The welfare consequences of hospital mergers' National Bureau of Economic Research Working Paper 12244. Cambridge, MA: doi: 10.3386/w12244.

Vita, M. G. (1990) 'Exploring hospital production relationships with flexible functional forms', *Journal of Health Economics*, 9(1), pp. 1-21. doi: 10.1016/0167-6296(90)90038-5.

Vitaliano, D. F. (1987) 'On the estimation of hospital cost functions', *Journal of Health Economics*, 6(4), pp. 305-318. doi: 10.1016/0167-6296(87)90018-x.

Walther, L. M., and Skousen, C. J. (2009) *Managerial and cost accounting*. London: Ventus Publishing ApS.

Weaver, M. and Deolalikar, A. (2004) 'Economies of scale and scope in Vietnamese hospitals', *Social Science & Medicine*, 59(1), pp. 199-208. doi: 10.1016/j.socscimed.2003.10.014.

Wegmann, G., and Stephen, N. (2009) 'The activity-based costing method developments: stateof-the art and case study', *The IUP Journal of Accounting Research and Audit Practices*, 8(1), pp. 7-22.

Weisbrod, B. A. and Dominguez, N. D. (1986) 'Demand for collective goods in private nonprofit markets: can fundraising expenditures help overcome free-rider behavior?', *Journal of Public Economics*, 30(1), pp. 83-96.

Weissman, J. S., Stern, R. S. and Epstein, A. M. (1994) 'The impact of patient socioeconomic status and other social factors on readmission: a prospective study in four Massachusetts hospitals', *Inquiry*, 31(2), pp. 163-172.

Westert, G. P., Lagoe, R. J., Keskimäki, I., Leyland, A. and Murphy, M. (2002) 'An international study of hospital readmissions and related utilization in Europe and the USA', *Health Policy*, 61(3), pp. 269-278. doi: 10.1016/s0168-8510(01)00236-6.

Weygandt, J. J., Kimmel, P. D., and Kieso, D. E. (2009) *Managerial accounting: Tools for business decision making*. Hoboken, New Jersey: John Wiley & Sons.

Wilson, G. W. and Jadlow, J. M. (1982) 'Competition, profit incentives, and technical efficiency in the provision of nuclear medicine services', *The Bell Journal of Economics*, 13(2), pp. 472-482.

Xirasagar, S. and Lin, H. C. (2006) 'Effects of payment incentives, hospital ownership and competition on hospitalization decisions for ambulatory surgical procedures', *Health Policy*, 76(1), pp. 26-37. doi: 10.1016/j.healthpol.2005.04.011.

Zelman, W. N., Pink, G. H. and Matthias, C. B. (2003) 'Use of the balanced scorecard in health care', *Journal of Health Care Finance*, 29(4), pp. 1-16.

Zuckerman, S., Hadley, J. and Iezzoni, L. (1994) 'Measuring hospital efficiency with frontier cost functions', *Journal of Health Economics*, 13(3), pp. 255-280. doi: 10.1016/0167-6296(94)90027-2.

Zweifel, P., Felder, S. and Meiers, M. (1999) 'Ageing of population and health care expenditure: a red herring?', *Health Economics*, 8(6), pp. 485-496.

Appendix 1

		•	0 7 0		1000 / 001/
Table A.	. The De	evelopment	of Keference	Cost Index from	1998 to 2016

Year	Change
	- To eliminate some of the cost differences found in the 1998 publication, the NHS Executive introduced a new Costing Manual as the basis for costing the 1998/99 submissions. The more prescriptive approach adopted assisted in eliminating the variations caused by different costing methodologies. This process was continued to ensure that any opportunities to shift costs between services were removed.
1998	- In addition, a reconciliation of the reference cost financial statements was undertaken with the final accounts for each NHS trust included in reference costs, to ensure that all relevant costs were included in this database.
	- The first publication concentrated on providing information on over 5000 surgical procedures covering almost 5 million episodes of acute care in 1997/98. That financial year saw this rise to almost 20 million inpatient and outpatient episodes.
1999	- This publication covered 69.4 million inpatient, outpatient and Accident and Emergency episodes and an increased range of surgical and medical treatments. This was an increase of 250% over the previous financial year. The document introduced information on Accident and Emergency services and a significant range of outpatient services for the first time. Small elements of community-based services, were taken into account because most community services were excluded from the Index as only small areas of their total expenditure were included and this would not have been consistent or comparable across all providers.
2000	- Service coverage was further extended to include more detail about acute services and a wider range of community services.
2000	- £21 billion of NHS expenditure was covered, representing 83% of expenditure on hospital and community health services.
2001	- A further extended range of services was included, covering £25 billion of NHS expenditure representing almost 89% of expenditure on hospital and community health services.
	- 'Payment by Results' was introduced.
2002	- Payment, at a predetermined national tariff, was required for 15 specific procedures (HRGs) where activity exceeded the level of the previous year. Prior to this, prices were negotiated locally. 6 clinical specialties were to be contracted using national case-mix adjustments, to facilitate comparisons of relative workload, taking into account not just patient numbers but the severity of their conditions.
	- The publication covered NHS expenditure details for in excess of £30 billion, increasing the scope of the Reference Cost system by £5 billion from the previous year.
	- This was the occasion on which a spell, covering a patient's entire stay from admission to discharge, rather than a finished consultant episode (FCE) was used as the basis for the tariff.

2003	 This publication covered £33 billion of NHS expenditure, increasing by £3 billion the Reference Costs coverage year. This means that more treatments were added. The HRG version from 1997 was HRG v3. In 2003 the new version was HRG v3.5.
2004	- Between 2003 and 2004 changes were made to the way the increases in pensions indexation was treated.
2005	- This year was the first in which some trusts were granted foundation status.
	- Major changes were made to data gathering, usage and Reference Costs Indices calculation. £41 billion of NHS expenditure was covered for 2006/07 including cost data from more than 400 NHS providers.
	- For the first time, the Healthcare Resource Group version 4 (HRG4) was used. The HRG4 currency supported the policy of Payment by Results (PbR) and revised the system to account for complex cases and expand the groupings used previously.
2006	- The design of HRG4 included:
2006	• Increased Clinical Coverage, including Chemotherapy, Radiotherapy, Specialist Palliative Care, Diagnostic Imaging, Rehabilitation and Critical Care.
	Improved accounting for severity and case complexity by using Complications and Co-morbidities.
	• Enhancing HRGs' representation of costs by separating activities to account for expensive drugs and other costly aspects of treatments. This also allowed HRGs to be independent of setting to facilitate healthcare funding.
	- Adult Critical Care data was not included because this was the first year it was the 'number of organs supported' was used as the basis for calculation instead of 'levels of critical care'.
2010	- Best Practice Tariffs was introduced.
2012	- For the first time, the Healthcare Resource Group version 4+ (HRG v4+) was used
Note:	This table shows the main changes to the Reference Cost Index system between 1998 and 2016 as documented by the National Health Service websites.

Appendix 2

Table A. Descriptive Statistics of RCIs of all Trusts from 1997 to 2016

	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	162	186	154	212	199	173	147	153	153	158	162	148	157	134	167	353	149	141	136	133
95 Percentile	119	127	117	141	127	125	123	118	120	119	115	117	115	115	123	117	115	114	114	115
Median	99	100	96	98	100	100	99	99	100	101	99	99	99	98	99	100	101	100	100	100
5 Percentile	82	84	81	80	82	83	84	84	87	85	87	88	89	89	88	88	87	89	89	88
MIN	67	67	71	54	46	28	42	51	50	69	81	75	80	81	74	63	71	75	80	72
STD	14	16	12	21	17	15	12	12	12	12	11	9	9	8	11	19	9	9	8	9
Average	100	103	98	102	102	101	101	100	101	101	100	100	100	100	102	102	101	101	100	101
Number of Trusts	196	216	221	313	284	273	268	267	265	240	238	239	236	243	248	244	244	239	237	234
Number of Trusts within 99-101	10	11	14	18	21	27	13	28	26	14	16	30	31	28	23	26	31	23	31	20
%	5	5	6	6	7	10	5	10	10	6	7	13	13	12	9	11	13	10	13	9
Number of Trusts below 101	113	114	151	188	154	153	146	155	143	120	135	145	145	152	137	138	131	128	136	124
%	57	53	68	60	54	56	54	58	54	50	57	61	61	63	55	57	54	54	57	53
Number of new Trusts	197	20	11	112	31	57	7	0	1	18	2	1	2	8	9	3	1	1	0	0
Number of disappeared Trusts	0	1	6	20	60	68	12	1	3	43	4	0	5	1	4	7	1	6	2	3

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was below 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the previous year; Number of disappeared trusts= number of trust that their data did not exist in the given year.

West Midlands	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	141	134	154	176	166	156	147	133	134	128	116	118	139	118	133	118	117	114	115	119
95 Percentile	135	121	115	138	128	130	119	115	121	116	113	115	116	115	127	115	113	111	112	113
Median	100	97	99	102	105	99	99	100	98	99	100	100	101	101	102	103	102	102	101	99
5 Percentile	87	89	93	85	86	86	81	80	87	86	87	90	93	92	94	91	93	91	91	90
MIN	87	87	84	82	77	78	72	69	57	81	81	88	92	88	92	78	91	82	91	84
STD	15	11	13	20	16	16	13	13	14	10	8	8	9	8	11	9	7	7	6	8
Average	103	100	102	107	105	102	100	98	99	100	100	100	103	102	106	103	102	101	101	100
Number of Trusts	22	22	23	37	35	32	29	29	29	28	26	27	27	28	28	28	28	27	27	26
Number of Trusts within 99-101	1	1	2	0	2	3	2	4	3	2	2	4	3	5	2	4	2	4	4	1
%	5	5	9	0	0	9	7	14	10	7	8	15	11	18	7	14	7	15	15	4
Number of Trusts below 101	12	13	15	18	14	20	17	17	18	16	14	15	13	14	13	12	12	13	15	14
%	55	59	65	49	40	63	59	59	62	57	54	56	48	50	46	43	43	48	56	54
Number of new Trusts	22	0	1	17	2	5	2	0	0	2	0	1	0	1	1	0	0	0	0	0
Number of disappeared Trusts	0	0	0	3	4	8	5	0	0	3	2	0	0	0	1	0	0	1	0	1
Acute	17	17	18	19	18	16	16	16	16	16	15	15	15	15	15	15	15	14	14	14
Specialist	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	2
Community	0	0	0	4	2	0	0	0	0	0	0	0	0	1	2	2	2	2	2	2
Ambulance	0	0	0	0	0	4	4	4	4	2	1	1	1	1	1	1	1	1	1	1
Mental health	2	2	2	11	12	9	6	6	6	7	7	8	8	8	7	7	7	7	7	7
Foundation	0	0	0	0	0	0	0	0	2	5	10	10	12	13	13	14	14	13	13	13

Table B. Descriptive Statistics of RCIs of all Trusts in West Midlands from 1997 to 2016

					-															
Wales	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	123	118	122	114	116	107	125	120	119	141	129	116	108	105	113	111	111	109	108	109
95 Percentile	122	117	120	112	114	107	123	119	118	139	127	115	107	105	113	111	111	109	108	108
Median	111	109	106	97	97	103	107	105	106	116	108	107	104	102	106	109	109	108	106	108
5 Percentile	100	101	93	96	94	99	91	92	94	94	89	100	100	99	99	96	102	96	91	94
MIN	99	100	91	96	94	99	89	90	92	91	86	99	100	98	98	95	101	95	89	92
STD	17	13	21	10	12	6	26	21	19	35	30	12	5	5	7	9	5	8	10	9
Average	111	109	106	102	102	103	107	105	106	116	108	107	104	102	106	105	107	104	101	103
Number of Trusts	2	2	2	3	3	2	2	2	2	2	2	2	2	2	3	3	3	3	3	3
Number of Trusts within 99-101	0	1	0	0	0	0	0	0	0	0	0	1	1	0	0	0	0	0	0	0
%	0	50	0	0	0	0	0	0	0	0	0	50	50	0	0	0	0	0	0	0
Number of Trusts below 101	1	1	1	2	2	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1
%	50	50	50	67	67	50	50	50	50	50	50	50	50	50	33	33	0	33	33	33
Number of new Trusts	2	0	0	1	0	0	1	0	0	0	0	0	0	0	1	0	0	0	0	0
Number of disappeared Trusts	0	0	0	0	0	1	1	0	0	0	0	0	0	0	0	0	0	0	0	0
Acute	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Specialist	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Community	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1
Ambulance	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Mental health	0	0	0	1	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Foundation	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1

Table C. Descriptive Statistics of RCIs of all Trusts in Wales from 1997 to 2016

South West	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	118	151	101	129	125	140	134	132	134	147	148	128	122	134	167	135	127	128	115	116
95 Percentile	110	116	101	112	115	126	129	119	127	112	123	118	117	129	131	125	124	118	114	113
Median	95	96	92	91	95	103	101	99	99	98	98	99	96	97	96	100	99	98	99	97
5 Percentile	75	81	82	74	87	82	86	87	88	84	87	91	90	89	86	89	90	90	88	90
MIN	68	70	77	70	84	76	81	85	85	69	85	85	84	86	79	85	86	90	86	85
STD	12	15	6	13	10	14	13	11	12	13	13	10	9	12	18	12	11	10	9	8
Average	95	98	93	93	98	102	102	100	101	98	101	101	99	102	102	101	102	102	100	99
Number of Trusts	23	24	25	30	28	31	31	31	31	28	28	28	28	28	27	27	28	27	26	26
Number of Trusts within 99-101	1	0	1	4	1	1	1	4	3	2	1	3	1	2	2	4	3	2	1	1
%	4	0	4	13	4	3	3	13	10	7	4	11	4	7	7	15	11	7	4	4
Number of Trusts below 101	15	18	23	24	18	13	15	19	18	17	17	17	18	18	18	17	16	17	14	16
%	65	75	92	80	64	42	48	61	58	61	61	61	64	64	67	63	57	63	54	62
Number of new Trusts	23	1	2	5	2	7	0	0	0	2	0	0	0	0	0	1	1	0	0	0
Number of disappeared Trusts	0	0	1	0	4	4	0	0	0	5	0	0	0	0	1	1	0	1	1	0
Acute	20	20	20	21	21	19	19	19	19	19	19	19	19	19	18	18	18	18	18	18
Specialist	0	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	0	0
Community	2	2	2	3	0	0	0	0	0	0	0	0	0	0	0	1	2	2	1	1
Ambulance	0	0	0	0	0	5	5	5	5	2	2	2	2	2	2	1	1	1	1	1
Mental health	1	1	2	5	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6
Foundation	0	0	0	0	0	0	0	0	4	11	16	16	17	18	18	18	18	17	17	18

Table D. Descriptive Statistics of RCIs of all Trusts in South West from 1997 to 2016

South East	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	148	150	117	149	140	122	123	119	132	138	133	123	123	117	127	113	104	106	107	113
95 Percentile	128	127	115	135	125	120	113	112	118	120	113	115	111	111	110	107	104	104	106	109
Median	100	103	99	101	102	99	98	98	97	99	97	98	98	98	98	96	98	97	99	98
5 Percentile	84	89	85	83	92	88	88	86	87	82	92	87	90	89	87	88	92	89	91	84
MIN	82	86	84	57	90	84	81	81	83	81	85	86	84	81	82	78	89	87	86	82
STD	16	14	9	18	11	11	9	8	10	13	9	9	8	7	9	8	5	5	5	8
Average	103	106	99	103	104	101	99	98	100	100	100	99	99	98	99	97	98	98	98	98
Number of Trusts	22	25	26	39	35	33	33	33	31	26	26	26	26	27	27	27	27	26	26	25
Number of Trusts within 99-101	1	1	3	2	4	1	2	1	3	2	2	2	3	4	4	2	2	1	4	2
%	5	4	12	5	11	3	6	3	10	8	8	8	12	15	15	7	7	4	15	8
Number of Trusts below 101	12	10	17	20	16	19	21	22	19	15	16	18	17	19	18	18	17	16	18	15
%	55	40	65	51	46	58	64	67	61	58	62	69	65	70	67	67	63	62	69	60
Number of new Trusts	22	4	1	15	2	9	0	0	0	4	0	0	1	1	1	1	0	0	0	0
Number of disappeared Trusts	0	1	0	2	6	11	0	0	2	9	0	0	1	0	1	1	0	1	0	1
Acute	17	19	20	20	20	17	17	17	17	16	16	16	16	16	16	17	17	16	16	15
Specialist	2	2	2	2	2	2	2	2	2	2	2	2	2	2	1	1	1	1	1	1
Community	1	3	3	9	4	2	2	2	2	2	2	2	2	3	3	2	2	2	2	2
Ambulance	0	0	0	0	0	4	4	4	4	1	1	1	1	1	1	1	1	1	1	1
Mental health	2	1	1	8	9	8	8	8	6	5	5	5	5	5	6	6	6	6	6	6
Foundation	0	0	0	0	0	0	0	0	2	6	9	10	12	12	14	14	15	14	14	16

Table E. Descriptive Statistics of RCIs of all Trusts in South East from 1997 to 2016

North West	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	162	133	146	212	137	173	135	153	150	138	128	122	123	117	131	112	113	122	119	115
95 Percentile	119	124	118	146	125	116	123	131	132	133	119	118	119	115	113	109	109	111	112	112
Median	95	100	98	98	100	101	102	102	102	103	99	99	99	100	99	100	100	102	100	102
5 Percentile	82	80	81	80	76	80	85	83	87	86	87	88	90	89	88	86	88	91	86	86
MIN	80	75	75	54	61	67	77	75	82	80	83	83	89	86	74	81	71	75	80	72
STD	15	13	13	24	15	15	12	15	15	14	11	9	8	8	10	7	8	8	8	9
Average	97	101	99	103	101	102	102	103	104	104	100	100	101	101	100	100	100	101	100	100
Number of Trusts	41	44	43	61	53	46	45	45	45	41	42	42	42	44	45	44	44	44	44	43
Number of Trusts within 99-101	1	4	2	5	5	4	2	3	5	0	3	3	4	6	3	4	8	3	5	4
%	2	9	5	8	9	9	4	7	11	0	7	7	10	14	7	9	18	7	11	9
Number of Trusts below 101	28	24	26	38	28	22	21	22	21	14	25	25	25	23	24	24	25	21	24	19
%	68	55	60	62	53	48	47	49	47	34	60	60	60	52	53	55	57	48	55	44
Number of new Trusts	41	3	0	20	7	13	2	0	0	1	1	0	0	2	1	0	0	0	0	0
Number of disappeared Trusts	0	0	1	2	15	20	3	0	0	5	0	0	0	0	0	1	0	0	0	1
Acute	26	27	27	27	28	27	26	26	26	26	26	26	26	26	26	25	25	25	25	25
Specialist	5	5	4	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6
Community	1	2	2	16	12	0	0	0	0	0	0	0	0	2	3	3	3	3	3	3
Ambulance	0	0	0	0	0	4	4	4	4	1	1	1	1	1	1	1	1	1	1	1
Mental health	9	10	10	12	7	9	9	9	9	8	9	9	9	9	9	9	9	9	9	8
Foundation	0	0	0	0	0	0	0	0	5	13	25	25	28	30	30	30	30	30	30	34

North East	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	155	118	113	127	152	157	139	113	116	123	127	118	111	111	135	129	141	130	130	127
95 Percentile	130	117	110	115	123	133	128	112	115	113	113	112	111	110	112	112	111	112	116	126
Median	105	103	96	98	98	101	102	100	103	107	102	102	102	100	102	98	100	99	98	98
5 Percentile	90	82	90	63	86	81	88	88	95	95	96	92	89	90	85	93	92	93	92	92
MIN	90	77	88	61	84	74	76	72	87	84	90	88	85	86	74	85	85	87	89	88
STD	16	12	7	16	15	17	13	9	7	8	7	7	6	7	12	9	11	9	9	10
Average	107	102	99	96	100	103	103	99	104	105	103	102	101	101	102	101	102	101	100	100
Number of Trusts	15	18	20	27	26	27	27	27	27	22	22	22	22	22	22	21	21	21	21	21
Number of Trusts within 99-101	0	2	2	0	1	6	1	3	1	2	2	4	5	6	1	2	2	2	3	2
%	0	11	10	0	4	22	4	11	4	9	9	18	23	27	5	10	10	10	14	10
Number of Trusts below 101	6	8	13	15	15	15	11	15	6	7	9	10	10	14	10	13	11	12	15	15
%	40	44	65	56	58	56	41	56	22	32	41	45	45	64	45	62	52	57	71	71
Number of new Trusts	15	3	2	9	2	9	0	0	0	4	0	0	0	0	1	0	0	0	0	0
Number of disappeared Trusts	0	0	0	2	3	8	0	0	0	9	0	0	0	0	1	1	0	0	0	0
Acute	10	12	14	14	14	15	15	15	15	14	14	14	14	14	14	13	13	13	13	13
Specialist	2	2	2	2	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Community	1	1	1	1	1	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1
Ambulance	0	0	0	0	0	4	4	4	4	2	2	2	2	2	2	2	2	2	2	2
Mental health	2	3	3	10	10	8	8	8	8	6	6	6	6	6	5	5	5	5	5	5
Foundation	0	0	0	0	0	0	0	0	4	9	11	11	15	15	16	16	16	16	16	16

Table G. Descriptive Statistics of RCIs of all Trusts in North East from 1997 to 2016

Greater London	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	136	158	145	201	199	149	141	149	153	158	162	148	157	132	135	137	132	133	136	129
95 Percentile	119	151	124	164	152	122	118	114	114	113	110	117	116	119	115	116	117	124	116	120
Median	103	103	96	98	102	99	97	99	99	99	100	101	99	98	101	100	100	101	102	102
5 Percentile	91	87	76	79	80	81	84	86	89	85	86	84	87	89	90	86	83	87	88	89
MIN	67	67	71	54	46	28	42	51	50	78	82	75	80	88	88	63	77	81	83	85
STD	12	19	15	28	26	17	14	13	14	12	14	12	12	10	10	12	11	12	11	10
Average	105	106	100	108	106	99	98	99	100	100	101	100	101	101	101	100	101	102	102	103
Number of Trusts	29	36	37	57	54	50	50	50	50	47	46	46	45	46	47	45	44	42	41	41
Number of Trusts within 99-101	5	1	1	2	1	5	1	6	8	4	3	7	6	2	8	5	9	3	4	6
%	17	3	3	4	2	10	2	12	16	9	7	15	13	4	17	11	20	7	10	15
Number of Trusts below 101	12	17	23	33	24	33	34	30	33	28	24	26	28	29	26	25	25	21	20	19
%	40	47	62	58	44	66	68	60	66	60	52	57	62	63	55	56	57	50	49	46
Number of new Trusts	30	6	4	22	8	7	0	0	1	1	1	0	1	1	1	1	0	1	0	0
Number of disappeared Trusts	0	0	3	2	11	11	0	0	1	4	2	0	2	0	0	3	1	3	1	0
Acute	23	26	28	30	29	28	28	28	28	28	28	28	27	27	27	25	24	22	21	21
Specialist	3	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5
Community	1	1	1	9	5	0	0	0	0	0	0	0	0	1	2	2	2	2	2	2
Ambulance	0	0	0	0	0	5	5	5	5	2	2	2	2	2	2	2	2	2	2	2
Mental health	3	4	3	13	15	12	12	12	12	12	11	11	11	11	11	11	11	11	11	11
Foundation	0	0	0	1	0	0	0	0	5	11	14	14	16	18	20	21	22	22	22	23

Table H. Descriptive Statistics of RCIs of all Trusts in Greater London from 1997 to 2016

East of England	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	115	174	117	159	150	150	128	123	119	115	109	117	119	123	116	122	114	114	111	116
95 Percentile	113	130	111	134	138	115	119	118	113	114	107	110	112	108	113	112	111	110	105	113
Median	95	97	95	98	96	98	97	97	96	98	96	96	98	96	98	96	97	99	99	101
5 Percentile	84	83	81	83	85	83	85	81	87	85	88	84	85	87	87	88	87	88	88	87
MIN	82	78	81	79	74	81	83	76	80	81	81	82	80	82	80	78	85	87	85	86
STD	10	21	9	18	17	14	11	11	9	9	7	8	8	8	9	10	8	7	7	9
Average	96	101	96	101	100	98	98	98	98	99	97	97	98	97	98	98	98	98	98	99
Number of Trusts	17	18	18	25	23	26	25	24	24	22	22	22	21	22	22	22	22	22	22	22
Number of Trusts within 99-101	0	0	1	2	2	4	2	1	0	1	1	2	2	0	2	2	2	5	5	2
%	0	0	6	8	9	15	8	4	0	5	5	9	10	0	9	9	9	23	23	9
Number of Trusts below 101	13	11	14	16	17	17	17	15	15	13	16	17	16	18	15	16	15	15	15	12
%	76	61	78	64	74	65	68	63	63	59	73	77	76	82	68	73	68	68	68	55
Number of new Trusts	17	1	0	10	3	4	1	0	0	1	0	0	0	2	0	0	0	0	0	0
Number of disappeared Trusts	0	0	0	3	5	1	2	1	0	3	0	0	1	1	0	0	0	0	0	0
Acute	14	14	14	16	16	15	14	14	14	14	14	14	13	13	13	13	13	13	13	13
Specialist	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Community	1	2	2	4	0	0	0	0	0	0	0	0	0	2	2	2	2	2	2	2
Ambulance	0	0	0	0	0	2	2	2	2	1	1	1	1	1	1	1	1	1	1	1
Mental health	1	1	1	4	6	8	8	7	7	6	6	6	6	5	5	5	5	5	5	5
Foundation	0	0	0	0	0	0	0	0	3	9	13	13	13	13	14	14	14	14	14	14

Table I. Descriptive Statistics of RCIs of all Trusts in East of England from 1997 to 2016

East Midlands	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	116	186	131	118	113	139	123	114	120	117	115	115	107	109	129	353	149	141	131	133
95 Percentile	112	114	123	114	107	120	114	114	112	115	113	115	105	107	125	135	124	114	113	128
Median	99	104	98	99	99	101	103	101	102	104	101	99	99	99	102	102	101	102	100	101
5 Percentile	87	89	84	89	86	92	96	93	94	95	91	94	94	94	93	93	91	91	93	92
MIN	87	87	76	86	74	89	94	91	93	91	91	91	92	88	92	87	86	90	92	90
STD	9	20	12	8	8	11	7	7	7	7	7	6	4	5	12	53	13	10	9	11
Average	100	107	99	100	97	104	105	103	102	103	101	100	99	100	105	114	105	103	102	103
Number of Trusts	19	21	22	29	22	22	22	22	22	20	20	20	20	20	23	23	23	23	23	23
Number of Trusts within 99-101	0	1	2	2	5	3	2	6	2	1	2	4	6	3	0	2	3	2	4	1
%	0	5	9	7	23	14	9	27	9	5	10	20	30	15	0	9	13	9	17	4
Number of Trusts below 101	10	7	14	18	16	10	6	11	10	7	10	14	15	13	10	11	9	11	12	11
%	53	33	64	62	73	45	27	50	45	35	50	70	75	65	43	48	39	48	52	48
Number of new Trusts	19	2	1	12	5	3	1	0	0	3	0	0	0	0	3	0	0	0	0	0
Number of disappeared Trusts	0	0	0	5	12	3	1	0	0	5	0	0	0	0	0	0	0	0	0	0
Acute	13	14	14	14	15	14	14	14	14	12	12	12	12	12	12	12	12	12	12	12
Specialist	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Community	1	2	2	6	3	1	0	0	0	0	0	0	0	0	3	3	3	3	3	3
Ambulance	0	0	0	0	0	2	2	2	2	1	1	1	1	1	1	1	1	1	1	1
Mental health	4	4	5	8	3	4	5	5	5	6	6	6	6	6	6	6	6	6	6	6
Foundation	0	0	0	0	0	0	0	0	6	11	12	12	13	14	14	14	14	14	14	16

Table J. Descriptive Statistics of RCIs of all Trusts in East Midlands from 1997 to 2016

											-									
Yorkshire and The Humber	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	110	134	97	103	112	103	103	106	105	112	108	105	107	106	106	125	109	114	109	110
95 Percentile	108	124	97	103	108	102	102	104	104	110	107	105	106	105	106	122	109	112	108	109
Median	100	93	90	92	92	93	92	90	101	96	97	99	98	96	102	105	106	105	103	101
5 Percentile	82	84	80	84	72	87	83	88	92	89	87	91	95	92	97	100	99	101	92	97
MIN	77	84	79	83	69	86	82	88	90	88	86	90	95	91	96	100	98	100	90	97
STD	11	18	8	9	16	7	9	9	6	11	9	7	6	7	5	11	5	6	8	6
Average	97	97	89	93	90	94	92	93	99	98	97	98	100	97	102	109	105	106	101	102
Number of Trusts	6	6	5	5	5	4	4	4	4	4	4	4	3	4	4	4	4	4	4	4
Number of Trusts within 99-101	1	0	0	1	0	0	0	0	1	0	0	0	0	0	1	1	0	1	1	1
%	17	0	0	20	0	0	0	0	25	0	0	0	0	0	25	25	0	25	25	25
Number of Trusts below 101	4	5	5	4	4	3	3	3	2	2	3	2	2	3	2	1	1	1	2	2
%	67	83	100	80	80	75	75	75	50	50	75	50	67	75	50	25	25	25	50	50
Number of new Trusts	6	0	0	1	0	0	0	0	0	0	0	0	0	1	0	0	0	0	0	0
Number of disappeared Trusts	0	0	1	1	0	1	0	0	0	0	0	0	1	0	0	0	0	0	0	0
Acute	6	6	5	5	5	4	4	4	4	4	4	4	3	3	3	3	3	3	3	3
Specialist	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Community	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1	1
Ambulance	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Mental health	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Foundation	0	0	0	0	0	0	0	0	1	1	1	1	1	2	2	2	2	2	2	2

Table K. Descriptive Statistics of RCIs of all Trusts in Yorkshire and The Humber from 1997 to 2016

 Table L. Descriptive Statistics of RCIs of all Acute Trusts from 1997 to 2016

Acute	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	155	134	146	139	141	127	118	115	119	138	116	119	114	117	118	125	114	116	117	115
95 Percentile	117	117	110	116	114	111	110	111	111	112	109	110	110	110	109	112	109	109	110	110
Median	97	98	96	96	98	99	99	99	100	99	98	99	98	98	98	99	100	99	99	99
5 Percentile	83	84	82	79	86	87	87	87	88	86	86	88	90	89	90	90	89	90	89	89
MIN	74	75	75	54	69	82	79	72	83	69	81	79	80	86	87	78	83	88	85	83
STD	11	10	9	12	10	8	7	8	7	9	7	6	6	6	6	7	6	6	6	6
Average	99	99	96	96	98	99	99	99	100	99	98	99	99	99	99	99	99	99	99	99
Number of Trusts	147	156	161	167	167	156	154	154	154	150	149	149	146	146	145	142	141	137	136	135
Number of Trusts within 99-101	9	8	9	12	16	19	10	20	17	12	8	25	26	22	19	19	25	15	21	15
%	6	5	6	7	10	12	6	13	11	8	5	17	18	15	13	13	18	11	15	11
Number of Trusts below 101	91	92	124	121	105	101	91	96	90	87	94	102	105	105	98	91	90	83	89	80
%	62	59	77	72	63	65	59	62	58	58	63	68	72	72	68	64	64	61	65	59
Number of new Trusts	147	9	8	18	10	10	3	0	0	1	1	0	2	0	0	2	0	1	0	0
Number of disappeared Trusts	0	0	3	12	10	21	5	0	0	5	2	0	5	0	1	5	1	5	1	1
West Midlands	17	17	18	19	18	16	16	16	16	16	15	15	15	15	15	15	15	14	14	14
Wales	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
South West	20	20	20	21	21	19	19	19	19	19	19	19	19	19	18	18	18	18	18	18
South East	17	19	20	20	20	17	17	17	17	16	16	16	16	16	16	17	17	16	16	15
North West	26	27	27	27	28	27	26	26	26	26	26	26	26	26	26	25	25	25	25	25
North East	10	12	14	14	14	15	15	15	15	14	14	14	14	14	14	13	13	13	13	13
Greater London	23	26	28	30	29	28	28	28	28	28	28	28	27	27	27	25	24	22	21	21
East of England	14	14	14	16	16	15	14	14	14	14	14	14	13	13	13	13	13	13	13	13
East Midlands	13	14	14	14	15	14	14	14	14	12	12	12	12	12	12	12	12	12	12	12
Yorkshire and The Humber	6	6	5	5	5	4	4	4	4	4	4	4	3	3	3	3	3	3	3	3
Foundation	0	0	0	0	0	0	0	0	26	50	69	70	73	77	80	81	83	81	81	83

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the given year; West Midland, Wales, South West, North West, North East, Greater London, East of England, East Midland and Yorkshire and The Humber= number of trusts in those regions; Foundation= number of foundation trusts.

Table M. Descriptive Statistics of RCIs of all Ambulance Trusts from 2002 to 2016

	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
Ambulance	2002	2003	2004	2005	2000	2007	2008	2009	2010	2011	2012	2013	2014	2013	2010
MAX	173	135	138	150	113	109	118	112	112	112	112	111	110	106	113
95 Percentile	145	134	132	132	111	109	116	112	112	112	110	109	107	106	109
Median	100	97	100	102	101	104	107	102	103	107	100	101	102	102	102
5 Percentile	75	77	80	82	83	90	88	88	92	88	89	90	89	90	89
MIN	67	72	69	57	81	90	88	85	90	85	85	85	87	89	88
STD	25	18	17	18	11	7	11	9	8	9	8	7	7	6	8
Average	105	102	102	103	98	101	102	101	102	102	100	100	100	100	100
Number of Trusts	30	30	30	30	12	11	11	11	11	11	10	10	10	10	10
Number of Trusts within 99-101	2	0	2	1	1	0	0	1	0	0	2	1	0	0	0
%	7	0	7	3	8	0	0	9	0	0	20	10	0	0	0
Number of Trusts below 101	16	16	17	15	6	4	5	5	5	5	6	5	4	3	3
%	53	53	57	50	50	36	45	45	45	45	60	50	40	30	30
Number of new Trusts	30	0	0	0	10	0	0	0	0	0	0	0	0	0	0
Number of disappeared Trusts	0	0	0	0	28	1	0	0	0	0	1	0	0	0	0
West Midlands	4	4	4	4	2	1	1	1	1	1	1	1	1	1	1
Wales	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
South West	5	5	5	5	2	2	2	2	2	2	1	1	1	1	1
South East	4	4	4	4	1	1	1	1	1	1	1	1	1	1	1
North West	4	4	4	4	1	1	1	1	1	1	1	1	1	1	1
North East	4	4	4	4	2	2	2	2	2	2	2	2	2	2	2
Greater London	5	5	5	5	2	2	2	2	2	2	2	2	2	2	2
East of England	2	2	2	2	1	1	1	1	1	1	1	1	1	1	1
East Midlands	2	2	2	2	1	1	1	1	1	1	1	1	1	1	1
Yorkshire and The Humber	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Foundation	0	0	0	0	0	0	0	0	2	4	5	5	5	5	5

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the previous year; Number of disappeared trusts= number of trust that their data did not exist in the given year; West Midland, Wales, South West, North West, North East, Greater London, East of England, East Midland and Yorkshire and The Humber= number of trusts in those regions; Foundation= number of foundation trusts.

Community	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	119	186	124	212	199	122	105	109	119	115	115	116	109	118	125	138	120	110	113	133
95 Percentile	115	179	120	183	186	121	104	109	118	115	115	115	108	113	116	120	118	109	111	116
Median	99	108	103	108	109	118	101	103	106	113	110	112	107	101	105	103	105	103	97	102
5 Percentile	82	88	88	81	75	105	98	98	94	111	106	109	106	86	86	87	92	93	88	88
MIN	79	85	88	79	61	103	98	98	93	111	105	109	106	81	82	85	87	88	87	87
STD	13	33	12	32	33	10	5	8	19	3	7	5	2	10	11	12	8	6	8	11
Average	98	126	104	116	115	114	101	103	106	113	110	112	107	100	103	103	104	101	98	101
Number of Trusts	8	13	13	52	27	3	2	2	2	2	2	2	2	10	18	18	19	19	18	18
Number of Trusts within 99-101	0	1	1	1	0	0	0	0	0	0	0	0	0	0	1	1	0	2	0	0
%	0	8	8	2	0	0	0	0	0	0	0	0	0	0	6	6	0	11	0	0
Number of Trusts below 101	4	3	6	20	9	0	1	1	1	0	0	0	0	5	7	8	5	9	10	8
%	50	23	46	38	33	0	50	50	50	0	0	0	0	50	39	44	26	47	56	44
Number of new Trusts	8	5	1	42	0	0	0	0	0	1	0	0	0	8	8	1	1	0	0	0
Number of disappeared Trusts	0	0	1	3	25	24	1	0	0	1	0	0	0	0	0	1	0	0	1	0
West Midlands	0	0	0	4	2	0	0	0	0	0	0	0	0	1	2	2	2	2	2	2
Wales	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1
South West	2	2	2	3	0	0	0	0	0	0	0	0	0	0	0	1	2	2	1	1
South East	1	3	3	9	4	2	2	2	2	2	2	2	2	3	3	2	2	2	2	2
North West	1	2	2	16	12	0	0	0	0	0	0	0	0	2	3	3	3	3	3	3
North East	1	1	1	1	1	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1
Greater London	1	1	1	9	5	0	0	0	0	0	0	0	0	1	2	2	2	2	2	2
East of England	1	2	2	4	0	0	0	0	0	0	0	0	0	2	2	2	2	2	2	2
East Midlands	1	2	2	6	3	1	0	0	0	0	0	0	0	0	3	3	3	3	3	3
Yorkshire and The Humber	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1	1
Foundation	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	6

Table N. Descriptive Statistics of RCIs of all Community Trusts from 1997 to 2016

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of trusts = number of trust that their data did not exist in the given year; West Midland, Wales, South West, North West, North East, Greater London, East of England, East Midland and Yorkshire and The Humber= number of trusts in those regions; Foundation= number of foundation trusts.

Table O. Descriptive Statistics of RCIs of all Mental Health Trusts from 1997 to 2016

Mental health	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	121	150	124	176	152	157	139	153	148	119	148	128	122	134	167	353	149	141	136	130
95 Percentile	118	124	117	135	126	120	121	114	115	114	113	117	115	114	132	131	125	128	122	123
Median	99	99	96	100	100	101	99	99	101	102	100	99	100	99	104	100	101	100	100	99
5 Percentile	80	71	81	79	78	80	83	81	87	84	88	85	88	88	78	80	85	85	86	83
MIN	68	67	77	54	46	28	42	51	50	78	82	75	80	82	74	63	71	75	80	72
STD	13	18	12	20	17	17	14	13	13	9	10	9	9	9	17	37	14	13	11	12
Average	99	101	98	104	102	101	100	99	100	101	100	100	100	101	105	105	102	102	102	101
Number of Trusts	24	26	27	72	69	64	62	61	59	56	56	57	57	56	55	55	55	55	55	54
Number of Trusts within 99-101	1	2	1	3	4	6	3	6	6	1	7	5	3	6	2	4	4	6	10	3
%	4	8	4	4	6	9	5	10	10	2	13	9	5	11	4	7	7	11	18	6
Number of Trusts below 101	15	15	16	39	36	32	34	36	30	24	32	35	30	32	23	29	27	29	31	29
%	63	58	59	54	52	50	55	59	51	43	57	61	53	57	42	53	49	53	56	54
Number of new Trusts	24	3	2	50	21	17	4	0	1	6	1	1	0	0	1	0	0	0	0	0
Number of disappeared Trusts	0	1	1	5	24	22	6	1	3	9	1	0	0	1	2	0	0	0	0	1
West Midlands	2	2	2	11	12	9	6	6	6	7	7	8	8	8	7	7	7	7	7	7
Wales	0	0	0	1	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
South West	1	1	2	5	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6
South East	2	1	1	8	9	8	8	8	6	5	5	5	5	5	6	6	6	6	6	6
North West	9	10	10	12	7	9	9	9	9	8	9	9	9	9	9	9	9	9	9	8
North East	2	3	3	10	10	8	8	8	8	6	6	6	6	6	5	5	5	5	5	5
Greater London	3	4	3	13	15	12	12	12	12	12	11	11	11	11	11	11	11	11	11	11
East of England	1	1	1	4	6	8	8	7	7	6	6	6	6	5	5	5	5	5	5	5
East Midlands	4	4	5	8	3	4	5	5	5	6	6	6	6	6	6	6	6	6	6	6
Yorkshire and The Humber	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Foundation	0	0	0	1	0	0	0	0	0	15	29	29	38	40	40	40	40	40	40	43

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the previous year; Number of disappeared trusts= number of trust that their data did not exist in the given year; West Midland, Wales, South West, South East, North West, North East, Greater London, East of England, East Midland and Yorkshire and The Humber= number of trusts in those regions; Foundation= number of foundation trusts.
Table P. Descriptive Statistics of RCIs of all Specialist Health Trusts from 1997 to 2016

1																				
Specialist	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	162	154	154	173	166	156	147	149	153	158	162	148	157	132	135	137	132	133	134	129
95 Percentile	150	134	145	154	147	150	141	133	135	148	156	124	140	131	135	123	124	124	119	121
Median	111	116	110	108	113	111	114	112	112	116	113	115	111	112	113	110	108	110	108	109
5 Percentile	91	90	81	83	83	86	87	84	91	93	88	88	89	90	96	93	92	94	93	92
MIN	67	88	71	79	80	82	86	75	87	86	87	83	88	90	90	93	88	92	86	86
STD	22	16	19	23	20	20	17	18	18	19	21	14	16	13	12	10	11	10	10	10
Average	116	114	112	112	115	114	115	111	113	118	115	111	112	110	111	109	109	109	108	108
Number of Trusts	17	21	20	22	21	20	20	20	20	20	20	20	20	20	19	19	19	18	18	17
Number of Trusts within 99-101	0	0	3	2	1	0	0	0	2	0	1	0	1	0	1	0	1	0	0	2
%	0	0	15	9	5	0	0	0	10	0	5	0	5	0	5	0	5	0	0	12
Number of Trusts below 101	3	4	5	8	4	4	4	5	7	3	5	3	5	5	4	4	4	3	3	4
%	17	19	25	36	19	20	20	25	35	15	25	15	25	25	21	21	21	17	17	24
Number of new Trusts	30	3	0	2	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Number of disappeared Trusts	0	0	1	0	1	1	0	0	0	0	0	0	0	0	1	0	0	1	0	1
West Midlands	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	3	2
Wales	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
South West	0	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	0	0
South East	2	2	2	2	2	2	2	2	2	2	2	2	2	2	1	1	1	1	1	1
North West	5	5	4	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6	6
North East	2	2	2	2	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Greater London	3	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5	5
East of England	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
East Midlands	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Yorkshire and The Humber	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Foundation	0	0	0	0	0	0	0	0	6	11	13	13	16	16	18	18	18	17	17	16

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts = number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the given year; West Midland, Wales, South West, North West, North East, Greater London, East of England, East Midland and Yorkshire and The Humber= number of trusts in those regions; Foundation= number of foundation trusts.

Table O.	Descriptive	Statistics (of RCIs o	of all Non	-Foundation	Trusts from	2005 to	2016
\mathbf{I} and \mathbf{V} .	DUSCHDUVU	oransulos y		n an 1 10n	-r vunuauvn	II USUS II VIII		AUTO

Tuble Q. Descriptive Studietes of Refs of un room i of	unuunon	11456511										
Non-Foundation	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	153	158	162	148	157	132	135	138	132	133	134	133
95 Percentile	120	119	115	116	112	115	122	115	110	110	112	113
Median	101	101	102	100	100	100	101	101	101	101	101	102
5 Percentile	87	85	86	88	90	90	90	88	87	89	89	89
MIN	50	80	81	79	84	81	74	78	86	82	86	82
STD	12	12	11	9	9	8	10	10	8	8	8	9
Average	101	102	102	101	101	101	101	101	101	101	101	102
Number of Trusts	233	164	127	127	109	108	106	100	98	96	94	81
Number of Trusts within 99-101	24	12	10	19	19	17	10	12	14	9	13	7
%	10	7	8	15	17	16	9	12	14	9	14	9
Number of Trusts below 101	123	80	60	71	62	61	55	54	48	49	50	37
%	53	49	47	56	57	56	52	54	49	51	53	46
Number of new Trusts	1	17	1	1	2	8	9	3	1	1	0	0
Number of disappeared Trusts	35	86	38	1	20	9	11	9	3	3	2	13

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts= number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the given year.

Table R. Descriptive Statistics of RCIs of all Foundation Trusts from 2005 to 2016

-												
Foundation	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016
MAX	125	147	148	128	139	134	167	353	149	141	136	130
95 Percentile	115	115	113	117	119	115	125	117	117	116	114	116
Median	98	99	97	98	98	98	99	99	100	100	99	99
5 Percentile	90	84	87	88	89	88	88	87	87	89	89	87
MIN	88	69	81	75	80	82	74	63	71	75	80	72
STD	9	12	9	9	9	9	12	23	10	9	9	9
Average	100	100	99	99	100	100	102	102	101	101	100	100
Number of Trusts	32	76	111	112	127	135	142	144	146	143	143	153
Number of Trusts within 99-101	2	2	6	11	12	11	13	14	17	14	18	13
%	6	3	5	10	9	8	9	10	12	10	13	8
Number of Trusts below 101	20	40	75	74	83	91	82	84	83	79	86	87
%	63	53	68	66	65	67	58	58	57	55	60	57
Number of new Trusts	32	44	35	1	15	8	7	2	2	0	0	12
Number of disappeared Trusts	0	0	0	0	0	0	0	0	0	3	0	2

Note: RCI= reference cost index; MAX= the highest RCI in a given year; MIN= the lowest RCI in a given year; STD= standard deviation; Number of trusts= number of trusts that their RCI was published; Number of trusts within 99-101= number of trust that their RCI was between 99% and 101%; %= the percentage of the number above it; Number of trusts below 101= number of trust that their RCI was below 101%; Number of new trusts= number of trust that their data did not exist in the given year.

Variable		20	002-2016			2005-2016					
variable	Ν	Mean	t	Mean Difference	Ν	Mean	t	Mean Difference			
West Midlands	419	101.182	2.339	1.182*	329	101.460	2.914	1.460**			
Wales	36	105.332	2.655	5.332*	30	105.383	2.540	5.383*			
South West	425	100.904	1.587	0.904	332	100.777	1.229	0.777			
South East	419	98.944	-2.524	-1.055*	320	98.781	-2.621	-1.219**			
North West	656	101.219	2.901	1.219**	520	100.946	2.225	0.946*			
North East	345	101.798	3.430	1.798**	264	101.798	3.453	1.798**			
Greater London	690	100.493	1.052	0.493	540	100.907	1.834	0.907			
East of England	340	98.019	-4.033	-1.980**	265	97.931	-4.107	-2.068**			
East Midlands	326	103.397	3.728	3.397**	260	103.240	2.911	3.240**			
Yorkshire and The Humber	59	99.530	-0.443	-0.469	47	101.165	1.062	1.164			
Note: RCI= reference cost index. * P-va	alue is ≤ 0.05 , *	** P-value is ≤ 0.01									

Table S: Results of One Sample t Test of the Mean RCI of each Region on the National Average

Table T: Results of One Sample t Test of the Mean RCI of each Type on the National Average

Variable		1	.997-2016		2005-2016					
variable	Ν	Mean	t	Mean Difference	Ν	Mean	t	Mean Difference		
Acute	2992	98.603	-9.456	-1.396**	1730	99.006	-6.152	-0.993**		
Ambulance					147	101.010	1.117	1.010		
Community	250	107.789	5.562	7.789**	130	102.209	2.614	2.209*		
Mental Health	1075	101.295	2.705	1.294**	670	101.492	2.499	1.492*		
Specialist	392	112.240	14.726	12.240**	230	111.302	11.995	11.302**		
Note: RCI= reference cost index. * P-va	alue is ≤ 0.05 , **	* P-value is ≤ 0.01	l							

Table U. Descriptive Statistics for ANOVA Test

Variable	1997-20	016	2002	-2016	2005-2	2016
	Ν	Mean	Ν	Mean	Ν	Mean
West Midlands	558	101.847	419	101.182	329	101.460
Wales	48	105.396	36	105.332	30	105.383
South West	555	99.590	425	100.905	332	100.778
South East	566	100	419	98.944	320	98.781
North West	898	101.083	656	101.221	520	100.946
North East	451	101.433	345	101.798	264	101.798
Greater London	904	101.517	690	100.494	540	100.908
East of England	440	98.243	340	98.019	265	97.932
East Midlands	439	102.640	326	103.398	260	103.240
Yorkshire and The Humber	87	97.871	59	99.530	47	101.165
Total	4946	100.860	3715	100.753	2907	100.797
Acute	2992	98.603	2194	98.929	1730	99.006
Ambulance	237	101.651	237	101.651	147	101.010
Community	250	107.790	137	102.480	130	102.209
Mental health	1075	101.204	857	101.187	670	101.493
Specialist	392	112.241	290	111.719	230	111.303
Total	4946	100.860	3715	100.753	2907	100.797
Non-foundation	1465	100.284	1464	100.352	1464	100.352
Foundation Trusts	3481	101.102	2251	101.014	1443	101.248
Total	4946	100.693	3715	100.683	2907	100.800

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	7688.101	9	854.233	6.709	0.000
Within Groups	471748.910	3705	127.328		
Total	479437.011	3714			
Note: RCI= reference.	cost index				

Table V: ANOVA test to compare between the means of RCIs of regions from 2002 to 2016

Table W: Welch and Brown-Forsythe tests for regions from 2002 to 2016

	Statistic	df 1		df 2	Sig.
Welch	7.40073		9	555.095	0.000
Brown-Forsythe	6.87617	,	9	1258.370	0.000

Table X: ANOVA test to compare between the means of RCIs of regions from 2005 to 2016

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	6093.660	9	677.073	5.773	0.000
Within Groups	339797.087	2897	117.293		
Total	345890.748	2906			
N & DOL 6					

Note: RCI= reference cost index.

Table Y: Welch and Brown-Forsythe tests for regions from 2005 to 2016

	Statistic	df 1		df 2	Sig.
Welch	6.94414		9	447.46	0.000
Brown-Forsythe	5.82549	1	9	954.877	0.000

Types	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	78288.147	4	19572.037	128.766	0.000
Within Groups	751018.678	4941	151.997		
Total	829306.826	4945			
Note: RCI= reference cost in	ndex.				

Table Z: ANOVA test to compare between the means of RCIs of types from 1997 to 2016

Table AA: Welch and Brown-Forsythe tests for types from 1997 to 2016

	Statistic	df 1	df 2	Sig.
Welch	80.3701	4	712.665	0.000
Brown-Forsythe	67.7722	4	1057.660	0.000

Table AB: ANOVA test to compare between the means of RCIs of types from 2005 to 2016

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	31525.445	4	7881.361	72.755	0.000
Within Groups	314365.303	2902	108.327		
Total	345890.748	2906			
Note: RCI= reference c	cost index.				

Table AC: Welch and Brown-Forsythe tests for types from 2005 to 2016 Page 100 - 2005

	Statistic	df 1		df 2	Sig.
Welch	47.2070		4	415.940	0.000
Brown-Forsythe	53.1367		4	989.853	0.000

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	7612.970	8	951.621	5.715	0.000
Within Groups	814066.070	4889	166.510		
Total	821679.041	4897			
Note: RCI= reference cost inde	х.				

Table AD: ANOVA test to compare between the means of RCIs of regions Excluding Wales from 1997 to 2016

Table AE: Welch and Brown-Forsythe tests for regions Excluding Wales from 1997 to 2016

	Statistic	df 1	df 2	Sig.
Welch	6.604	8	1198.931	0.000
Brown-Forsythe	6.211	8	3557.647	0.000

Appendix 3

Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
West Midlands	-0.115									
	(-0.133)									
Wales		-5.102*								
		(-1.837)								
South West			-1.402*							
			(-1.897)							
South East				-2.398***						
				(-3.463)						
North West					3.712***					
					(6.255)					
North East						-1.337				
						(-1.497)				
Greater London							-1.186			
							(-1.607)			
East of England								-2.383***		
								(-2.995)		
East Midlands									2.541***	
									(3.121)	
Yorkshire and The Humber										6.157***
										(3.528)
Acute	-4.719***	-4.724***	-4.105**	-4.828***	-3.915 ***	-4.770***	-4.659 ***	-4.270***	-3.982***	-5.024***
	(-3.121)	(-3.138)	(-2.669)	(-3.226)	(-2.657)	(-3.165)	(-3.093)	(-2.834)	(-2.626)	(-3.353)

Table A. Results of OLS Regression Test of RCI Relations with Regions from 2012 to 2016

Community	-2.780	-2.779	-2.296	-2.469	-2.285	-2.818	-3.651*	-2.576	-3.520*	-3.906**
-	(-1.431)	(-1.435)	(-1.176)	(-1.282)	(-1.209)	(-1.454)	(-1.813)	(-1.335)	(-1.812)	(-2.002)
Mental Health	-9.338***	-9.207***	-8.560***	-8.126 ***	-9.335***	-9.651***	-10.168***	-8.930***	-8.740***	-9.724***
	(-4.634)	(-4.584)	(-4.181)	(-4.013)	(-4.769)	(-4.775)	(-4.897)	(-4.456)	(-4.354)	(-4.868)
Foundation Trusts	-1.728***	-1.829***	-1.667 ***	-1.868 ***	-1.947***	-1.589 ***	-1.763 ***	-1.733***	-1.804***	-1.840***
	(-3.263)	(-3.497)	(-3.207)	(-3.606)	(-3.835)	(-3.017)	(-3.388)	(-3.350)	(-3.484)	(-3.559)
Size	2.862**	2.854**	2.317 **	2.862 **	3.051***	3.181***	3.034***	2.625**	2.302**	3.166***
	(2.532)	(2.531)	(1.992)	(2.554)	(2.774)	(2.770)	(2.678)	(2.331)	(2.025)	(2.818)
Male Patients	0.135**	0.133**	0.152***	0.151***	0.181***	0.134**	0.110*	0.138**	0.119**	0.144***
	(2.374)	(2.366)	(2.658)	(2.676)	(3.266)	(2.372)	(1.890)	(2.446)	(2.103)	(2.568)
In-Patients	-0.423**	-0.415***	-0.403***	-0.403***	-0.508***	-0.437***	-0.494***	-0.408***	-0.447***	-0.445***
	(-4.746)	(-4.655)	(-4.489)	(-4.549)	(-5.782)	(-4.881)	(-4.978)	(-4.599)	(-5.031)	(-5.022)
Out-Patients	-0.061**	-0.059**	-0.058**	-0.043	-0.047*	-0.064**	-0.072***	-0.055**	-0.061**	-0.063**
	(-2.351)	(-2.294)	(-2.271)	(-1.644)	(-1.886)	(-2.4713)	(-2.710)	(-2.136)	(-2.382)	(-2.451)
Intercept	101.468***	101.398***	101.290***	99.753***	98.109***	101.121***	104.172***	101.189***	103.265***	100.961***
	(24.029)	(24.102)	(24.074)	(23.689)	(23.713)	(23.976)	(23.015)	(24.140)	(24.438)	(24.141)
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
n	703	703	703	703	703	703	703	703	703	703
Adi R ²	0.09	0.10	0.10	0.11	0.14	0.10	0.10	0.10	0.11	0.11

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and regions during the period from 2012 to 2016. RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients. *, **, *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively. Colum 1 (2 to 10) West Midlands (Wales, South West, South East, North West, North East, Greater London, East of England, East Midlands and Yorkshire and The Humber, respectively) equals 1 if the trust is located in West Midlands (other regions), 0 otherwise. t-statistics of coefficients are in parentheses.

Variable	(1)	(2)	(3)	(4)
Acute	0.565			
	(0.478)			
Community		0.192		
		(0.124)	4 550444	
Mental Health			-4.550***	
			(-3.133)	2 245**
Specialist				5.545*** (2.255)
West Midlands	-6.297***	-6.272 ***	-6.083***	-6.833***
v ost minimus	(-3.444)	(-3.389)	(-3.348)	(-3.725)
Wales	-11.721***	-11.710***	-11.110***	-12.212 ***
Wales	(-3.759)	(-3.733)	(-3.581)	(-3.925)
South West	-7.716***	-7.651***	-7.996***	-7.790***
	(-4.394)	(-4.306)	(-4.159)	(-4.455)
South East	-8.587***	-8.586***	-7.996***	-9.031***
	(-4.903)	(-4.835)	(-4.572)	(-5.156)
North West	-3.049*	-3.033*	-2.742	-3.559**
	(-1.784)	(-1.750)	(-1.613)	(-2.075)
North East	-6.725***	-6.680***	-6.810***	-7.190***
	(-3.657)	(-3.603)	(-3.729)	(-3.902)
Greater London	-6.665***	-6.545***	-6.738***	-7.135***
	(-3.802)	(-3.654)	(-3.883)	(-4.065)
East of England	-8.448***	-8.385***	-8.151***	-8.684***
C	(-4.702)	(-4.609)	(-4.565)	(-4.845)
East Midlands	-3.340*	-3.385*	-3.387*	-3.932**
	(-1.861)	(-1.887)	(-1.904)	(-2.186)
Foundation Trusts	-2.576***	-2.597***	-2.138***	-2.775***
	(-5.185)	(-5.107)	(-4.191)	(-5.623)
Size	0.374	0.627	1.379	2.019*
	(0.371)	(0.684)	(1.504)	(1.885)
Male Patients	0.209***	0.209***	0.194***	0.193***
	(3.669)	(3.670)	(3.429)	(3.391)
In-Patients	-0.342***	-0.316***	-0.489***	-0.348***
	(-4.379)	(-4.936)	(-6.010)	(-5.623)
Out-Patients	0.018	0.021	-0.020	0.008
	(0.803)	(1.001)	(-0.821)	(0.392)
Intercept	100.085***	99.209***	103.015***	97.815***
-	(24.100)	(23.812)	(25.536)	(24.860)
Year FE	Yes	Yes	Yes	Yes
n	703	703	703	703
Adi R ²	0.17	0.17	0.19	0.18

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and types during the period from 2012 to 2016. RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients, Out-Patients= the total number of out-patients divided by the total number of all patients. *, **, *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively. Colum 1 (2 to 10) West Midlands (Wales, South West, South East, North West, North East, Greater London, East of England, East Midlands and Yorkshire and The Humber, respectively) equals 1 if the trust is located in West Midlands (other regions), 0 otherwise. t-statistics of coefficients are in parentheses.

Variable	Coefficient	t-Statistic
Foundation Trusts	-2.342***	-4.540
Size	2.528**	2.218
West Midlands	-6.792***	-3.694
Wales	-11.755***	-3.783
South West	-7.567***	-4.303
South East	-8.557***	-4.856
North West	-3.403**	-1.978
North East	-7.474***	-4.049
Greater London	-7.616***	-4.246
East of England	-8.623***	-4.786
East Midlands	-3.949**	-2.209
Acute	-2.957**	-1.992
Community	-4.444**	-2.266
Mental Health	-8.151***	-3.982
Male Patients	0.177***	3.106
In-Patients	-0.559***	-5.842
Out-Patients	-0.038	-1.457
Intercept	106.515***	23.081
Year FE	Yes	
n	703	
Adi R ²	0 19	

 Table C. Results of OLS Regression Test of RCI Relations with Foundation Status and Size

 from 2012 to 2016

Note: This table presents the results of OLS regression examining the association between RCI of all trusts and foundation status and size during the period from 2012 to 2016. RCI= reference cost index; Size= the natural log of number of beds available overnight; Male Patients= the total number of male patients divided by the total number of patients; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of all patients. *, **, *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.

Appendix 4

Table A. Results of Dynam	ic Panel Data Es	timation					
Variable	Mortality Rate	SDD	Infection Rate	Mean LOS	Emergency Waiting Time	In-Patient Waiting Time	Out-Patient Waiting Time
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Mortality Rate _{t-1}	0.413***						
	(0.072)						
SDD _{t-1}		0.325***					
		(0.083)					
Infection Rate t-1			0.075				
			(0.090)				
Mean LOS t-1				0.664***			
				(0.116)			
Emergency Waiting Time t-1					-0.005		
					(0.058)		
In-Patient Waiting Time t-1						0.402	
						(0.250)	
Out-Patient Waiting Time t-1							0.475***
							(0.124)
Cost t-1	-0.177*	0.466**	0.027	-0.357	-0.173	20.760	-12.070
	(-0.096)	(0.231)	(0.018)	(-0.481)	(-0.207)	(15.683)	(-11.733)
Foundation Trusts	0.021	-0.128	0.016	0.302	-0.224	17.872	13.774
	(0.067)	(-0.193)	(0.014)	(0.545)	(-0.148)	(20.705)	(13.695)
Total Episodes	0.005	-0.079	0.009	0.412	0.083	-7.598	0.523
L.	(0.051)	(-0.103)	(0.011)	(0.259)	(0.083)	(-13,919)	(5.649)
In-Patient	0.005	-0.014	-0.002**	-0.045	0.003	-1.381	0.499
	(0.006)	(-0.012)	(-0.001)	(-0.032)	(0.010)	(-1.049)	(0.807)
Out-Patient	0.002	-0.008	-0.000	0.025	-0.004	-0.723	-0.047
	(0.003)	(-0.007)	(-0.000)	(0.021)	(-0.004)	(-0.685)	(-0.348)
Male Patient	-0.005	0.014	0.000	0.043	-0.018	-0.140	-0.326

	(-0.006)	(0.015)	(0.001)	(0.048)	(-0.019)	(-1.225)	(-0.890)
AR (2)	-1.35	-1.25	-0.83	1.24	-1.21	-0.47	-1.41
	(-0.18)	(-0.21)	(-0.41)	(0.22)	(-0.23)	(-0.64)	(-0.16)
Hansen	32.67	32.89	32.25	30.99	33.29	35.89	32.24
	(0.53)	(0.52)	(0.55)	(0.62)	(0.50)	(0.38)	(0.55)
Ν	596	596	596	596	596	596	596

Note: This table presents the results of system GMM regression for healthcare cost. Column 1 to 7 present the dependent variables which are mortality rate, number of same-day discharge patients, infection rate, mean length of stay, percentage of emergency patient waited less than four hours, mean in-patient waiting time and mean out- patient waiting time respectively. instruments dated t–1 was used in this estimation. The sample period runs from 2010 till 2016. Time dummy variables were included in each model. AR 2 is a test for second-order serial autocorrelation in the residuals. Hansen test is for over-identifying restrictions and it is distributed as chi-square under the null hypothesis of instrument validity. *, ** and *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively. t-statistics of coefficients are in parentheses.

Variable	Mortality Rate	SDD	Infection Rate	Mean LOS	Emergency Waiting Time	In-Patients Waiting Time	Out-Patients Waiting Time
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
$Cost_dummy_{t-1}$	-0.000	-0.003	0.001	-0.019	0.011	3.234**	0.705
	(-0.009)	(-0.022)	(0.001)	(-0.052)	(0.011)	(1.408)	(0.959)
Foundation Trusts	0.041*	-0.066*	0.003*	-0.123**	0.026	-1.200	-0.035
	(0.022)	(-0.038)	(0.002)	(-0.049)	(0.023)	(-2.603)	(-3.503)
Total Episodes	-0.079***	0.163***	0.006***	-0.003	-0.010	-0.397	-0.860
	(-0.026)	(0.056)	(0.002)	(-0.093)	(-0.015)	(-2.857)	(-1.306)
In-Patients	-0.003	0.006	-0.000	-0.053***	-0.001	-0.446	-0.063
	(-0.003)	(0.006)	(-0.000)	(-0.016)	(-0.002)	(-0.409)	(-0.325)
Out-Patients	-0.001	0.003	-0.000	-0.007	-0.002**	-0.296	-0.189
	(-0.001)	(0.003)	(-0.000)	(-0.007)	(-0.001)	(-0.180)	(-0.120)
Male Patients	-0.003	0.006	0.000	0.030	0.002	0.174	0.144
	(-0.002)	(0.004)	(0.000)	(0.019)	(0.002)	(0.386)	(0.472)
Intercept	1.713***	3.041***	-0.016	4.456***	1.025***	69.870**	49.846*
	(0.281)	(0.617)	(0.026)	(1.461)	(0.169)	(31.935)	(26.236)
Trust FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	558	558	558	558	558	558	558
R-within	0.039	0.032	0.056	0.087	0.267	0.040	0.229
R-between	0.004	0.001	0.000	0.135	0.185	0.005	0.054
R-overall	0.001	0.001	0.002	0.093	0.203	0.013	0.062
Rho	0.836	0.966	0.392	0.851	0.443	0.451	0.733

Table B. Results of Panel Fixed Effect Regression of Cost dummy Relations with Healthcare Quality from 2010 to 2016

Note: Columns 1 to 7 are the dependent variables (i.e., mortality rate, number of same-day discharge patients, infection rate, mean length of stay, percentage of emergency patient waited less than four hours, mean in-patient waiting time and mean out- patient waiting time, respectively). Cost_dummy= lagged reference cost index of acute trusts with reference cost index higher than the average; Total episodes= the natural log of the number of episodes a trust carries out; In-Patient= the total number of in-patients divided by the total number of all patients; Out-Patients= the total number of out-patients divided by the total number of patients; Mortality rate= the actual number of deaths divided by the expected number; Same day discharge= The natural log of the number of patients who had surgery and returned home the same day; Infection rate= the number of patients infected divided by the total number of in-patients who were treated within 4 hours divided by the total number of emergency cases; In-patient waiting time= the number of months in-patients waited for treatment; Out-patients waiting time= the number of weeks out-patients waited for treatment. t-statistics of coefficients are in parentheses. *, ** and *** Indicate statistical significance at the p-value $\leq 0.10, \leq 0.05$ and ≤ 0.01 levels, respectively.